| UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK | | |
|---|---|------------------------|
| | X | |
| -In re | : | |
| | : | Chapter 11 Case No. |
| ALLEGIANCE TELECOM, INC., <u>et al</u> ., | : | 03-13057 (RDD) |
| | : | |
| Debtors. | : | (Jointly Administered) |
| | Х | |

DEBTORS' <u>AMENDED</u>JOINT PLAN OF REORGANIZATION <u>PURSUANT TO CHAPTER 11 OF THE BANKRUPTCY CODE</u>

KIRKLAND & ELLIS LLP Attorneys for the Debtors Citigroup Center 153 East 53rd Street New York, New York 10022 Telephone: (212) 446-4800 Facsimile: (212) 446-4800 Matthew A. Cantor Jonathan S. Henes Lisa G. Laukitis

Dated: New York, New York March 18, April 21, 2004

TABLE OF CONTENTS

Page 1

| ARTICLE I | DEFINITIONS AND CONSTRUCTION OF TERMS | 1 |
|-----------|---------------------------------------|------------|
| 1.1 | Access Provider | 1 |
| 1.2 | Acquired Assets | 1 |
| 1.3 | Administrative Expense Claim | |
| 1.4 | Administrative Expense Claim Bar Date | |
| 1.5 | Allowed | |
| 1.6 | Assumed Liabilities | |
| 1.7 | ATCW | |
| 1.8 | ATCW Debtors | |
| 1.9 | ATCW Unsecured Claim | |
| 1.10 | ATI | |
| 1.11 | ATI Note Claims | |
| 1.12 | ATI Note Indentures | |
| 1.13 | ATI Notes | |
| 1.14 | ATI Note Trustees | |
| 1.15 | ATI Unsecured Claim | |
| 1.16 | ATLT | |
| 1.17 | ATLT Agreement | |
| 1.18 | ATLT Assets | |
| 1.19 | ATLT A Certificates | |
| 1.20 | ATLT B-Certificates | |
| 1.21 | ATLT C Certificates | |
| 1.22 | ATLT Certificates | |
| 1.23 | Available Cash | |
| 1.24 | Avoidance Actions | — |
| 1.25 | Ballot | |
| 1.26 | Bankruptcy Code | |
| 1.27 | Bankruptcy Court | |
| 1.28 | Bankruptcy Rules | |
| 1.29 | Bar Date | — |
| 1.30 | Business Day | |
| 1.31 | Buyer | |
| 1.32 | Cash | |
| 1.33 | Cash Recovery | <u>34</u> |
| 1.34 | Cash Recovery Election | |
| 1.35 | Causes of Action | _ |
| 1.36 | Chapter 11 Cases | 4 <u>5</u> |
| 1.37 | Claim | 4 <u>5</u> |
| 1.38 | Claim Holder | 4 <u>5</u> |
| 1.39 | Claims Objection Deadline | 4 <u>5</u> |
| 1.40 | Class | |
| 1.41 | Closing | 4 <u>5</u> |
| 1.42 | Commencement Date | 4 <u>5</u> |
| 1.43 | Common Stock | 4 <u>5</u> |
| 1.44 | Confirmation Date | 4 <u>5</u> |
| 1.45 | Confirmation Hearing | 4 <u>5</u> |
| 1.46 | Confirmation Order | |
| 1.47 | Credit Agreement | 4 <u>6</u> |
| 1.48 | Creditors Committee | 4 <u>6</u> |
| 1.49 | Debtors | |
| 1.50 | Debtors in Possession | 5 <u>6</u> |

TABLE OF CONTENTS (continued)

Page

| 1.51 | Disclosure Schedules | 5 6 |
|-------|--|----------------|
| 1.52 | Disclosure Statement | |
| 1.53 | Disclosure Statement Hearing Date | |
| 1.54 | Disclosure Statement Order | _ |
| 1.55 | Disputed Claim | |
| 1.56 | Disputed Claim Reserve | |
| 1.57 | Disputed Claim Amount | |
| 1.58 | Distribution | |
| 1.59 | Distribution Account | |
| 1.60 | Distribution Date | |
| 1.61 | Entity | |
| 1.62 | Equity Interest | |
| 1.63 | Estates | |
| 1.64 | Excluded Assets | |
| 1.65 | Excluded Liabilities | |
| 1.66 | Federal Judgment Rate | |
| 1.67 | Final Distribution Date | _ |
| 1.68 | Final Effective Date | |
| 1.69 | Final Order | |
| 1.70 | Holder | |
| 1.71 | Impaired | |
| 1.72 | Impaired Claim | |
| 1.72 | Impaired Class | |
| 1.74 | Initial Effective Date | |
| 1.75 | Insured Claim | |
| 1.76 | Intercompany Claim | |
| 1.77 | IT Reorganized Subsidiaries | |
| 1.78 | Lien | |
| 1.79 | Litigation Cash | |
| 1.80 | New STFI Common Stock | |
| 1.81 | Non-Transferred Assets | |
| 1.82 | NTA Effective Dates | |
| 1.83 | NTA Reorganized Subsidiaries | |
| 1.84 | Operating Agreement | |
| 1.85 | Person | |
| 1.86 | Plan | |
| 1.87 | Plan Administrator | |
| 1.88 | Plan Supplement | |
| 1.89 | Post-Initial Effective Date Fees and Expenses | |
| 1.90 | Priority Non-Tax Claim | |
| 1.90 | Priority Tax Claim | |
| 1.92 | Professional | |
| 1.92 | Professional Fee Claims | |
| 1.94 | Purchase Agreement | |
| 1.95 | Record Date | |
| 1.96 | Reorganized STFI | |
| 1.90 | Reorganized Subsidiaries | |
| 1.98 | Reserve Fund | |
| 1.99 | Sale Order | |
| 1.39 | Sale Transaction | |
| 1.100 | Schedules | |
| 1.101 | Secured Claim | |
| 1.102 | Secured Claim | |
| 1.103 | Senior Discount Note Indenture | |
| 1.101 | Sentor 2 is could inter indentification in the indentin the indentification in the indentification in the indentif | |

TABLE OF CONTENTS (continued)

Page

| | 1.105 | Senior Discount Note Trustee | |
|-----|---------|---|--------------------|
| | 1.106 | Senior Lender Claims | |
| | 1.107 | Senior Lenders | |
| | 1.108 | Senior Note Indenture | |
| | 1.109 | Senior Note Trustee | |
| | 1.110 | Senior Notes | |
| | 1.111 | STFI | |
| | 1.112 | STFI Assets | |
| | 1.113 | STFI Cash Amount | |
| | 1.114 | Subordinated Claim | |
| | 1.115 | Subsidiaries | |
| | 1.116 | Tariff | |
| | 1.117 | Tariff Services | 9 <u>13</u> |
| | 1.118 | Tax Rate | |
| | 1.119 | Transition Services Agreement | 9 <u>13</u> |
| | 1.120 | Unencumbered Cash | 9 <u>13</u> |
| | 1.121 | Unimpaired | 9 <u>13</u> |
| | 1.122 | Unimpaired Claim | 9 <u>13</u> |
| | 1.123 | Unimpaired Class | |
| | 1.124 | Unsecured Claims | 9 <u>13</u> |
| | 1.125 | Utility Company | |
| | 1.126 | Utilities Order | <u>10</u> 14 |
| | 1.127 | Utility Services | <u>1014</u> |
| | 1.128 | Utility Service Order | <u>1014</u> |
| | 1.129 | Voting Deadline | <u>10</u> 14 |
| | 1.130 | Voting Record Date | |
| | 1.131 | Wind-Down Amount | |
| | 1.132 | X0 | <u>10</u> 14 |
| | 1.133 | XO Common Stock | |
| | | | |
| ART | ICLE II | TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS AND PRIORITY | |
| | | LAIMS | 1115 |
| | 2.1 | Administrative Expense Claims | |
| | 2.1 | Professional Fee Claims | |
| | 2.2 | Priority Tax Claims | |
| | 2.5 | Photicy Tax Claims | <u>++<u>10</u></u> |
| лрт | ютет | I SUMMARY CLASSIFICATION AND TREATMENT OF CLAIMS AND | |
| AKI | | | |
| | | Y INTERESTS | |
| | 3.1 | Priority Non-Tax Claims (Class 1) | |
| | 3.2 | Secured Claims (Class 2) | |
| | 3.3 | Senior Lender Claims (Class 3) | |
| | 3.4 | ATCW Unsecured Claims (Class 4) | |
| | 3.5 | ATI Unsecured Claims (Class 5) | <u>1318</u> |
| | 3.6 | Subordinated Claims (Class 6) | <u>1419</u> |
| | 3.7 | Equity Interests (Class 7) | <u>1419</u> |
| | | | |
| ART | ICLE I | V ACCEPTANCE OR REJECTION OF THE PLAN | <u>141</u> 9 |
| | 4.1 | Special Provisions Governing Unimpaired Claims. | |
| | 4.2 | Classes Entitled to Vote | |
| | 4.3 | Acceptance by Impaired Classes | |
| | 4.4 | Cramdown | |
| | | | |

TABLE OF CONTENTS (continued)

Page

| 5.1 | Substantive Consolidation of ATCW Debtors for Plan Purposes Only | |
|-----------------------|--|------------------------------|
| <u>5.15.2</u> | Deemed Consolidation of ATCW Debtors for Plan Purposes Only | |
| <u>5.2</u> 5.3 | Settlement of Certain Claims. | |
| <u>5.3</u> 5.4 | Restructuring Transactions | |
| 5.4 | -ATLT. | |
| 5.5 | ATLT. | <u></u> |
| <u>5.6</u> | Cancellation of Existing Securities and Agreements | |
| <u>5.6</u> 5.7 | Regulatory Approvals | |
| <u>5.7</u> <u>5.8</u> | Insurance Preservation | |
| Α ΡΤΙΛΙ Ε Χ | TEXECUTORY CONTRACTS AND UNEXRIPED FEASES | 2020 |
| | /I EXECUTORY CONTRACTS AND UNEXPIRED LEASES | |
| 6.1 6.2 | Assumption or Rejection of Executory Contracts and Unexpired Leases Utility Services. | |
| 6.2 6.3 | Tariff Services | |
| 0.3 | Tariff Services | <u>++</u> <u>FLLOL; POOR</u> |
| ARTICLE V | /II PROVISIONS GOVERNING DISTRIBUTIONS | <u>2230</u> |
| 7.1 | Distributions for Claims and Equity Interests Allowed as of the Initial Effective Date | |
| 7.2 | Delivery of Distributions by ATLT | |
| 7.3 | Delivery and Distributions and Undeliverable or Unclaimed Distributions | |
| 7.4 | Compliance with Tax Requirements/Allocations | |
| 7.5 | Record Date for Distribution | |
| 7.6 | Fractional ATLT Certificates and De Minimis Distributions | |
| 7.7 | Set-offs and Recoupments | |
| 7.8 | Surrender of Canceled Instruments or Securities | |
| 7.9 | ATI Notes | |
| ARTICLE V | /III PROCEDURES FOR TREATING DISPUTED CLAIMS | <u>2332</u> |
| 8.1 | Objections to Claims | |
| 8.2 | No Distributions Pending Allowance | |
| 8.3 | Personal Injury Claims | |
| 8.4 | Estimation of Claims | |
| 8.5 | Distributions Relating to Allowed Insured Claims | |
| 8.6 | Disputed Claims Reserve | |
| 8.7 | Distributions After Allowance | |
| 8.8 | Distributions After Disallowance | |
| 8.9 | Controversy Concerning Impairment | |
| ARTICLE I | X EFFECTIVENESS OF THE PLAN | <u>25<u>34</u></u> |
| 9.1 | Conditions Precedent to Initial Effective Date | |
| 9.2 | Conditions Precedent to NTA Effective Date | |
| 9.3 | Effect of Failure of Conditions | |
| 9.4 | Waiver of Conditions | |
| ARTICLE X | XEFFECT OF CONFIRMATION OF PLAN | 26 35 |
| 10.1 | Binding Effect | |
| 10.1 | Term of Injunction or Stays | |
| 10.2 | Preservation of Avoidance Actions | |
| 10.4 | Retention of Causes of Action/Reservation of Rights. | |
| 10.5 | Releases by the Debtors | |
| 10.6 | Releases by Holders of Claims and Equity Interests | |
| 10.7 | Release of Buyer. | |
| 10.8 | Exculpation and Limitation of Liability | |
| 10.9 | Discharge of Claims and Termination of Equity Interests | |

TABLE OF CONTENTS (continued)

Page

| | 10.10 | Injunction | 2838 |
|------|--------|---|----------------|
| | 10.11 | Release of Liens | 2938 |
| | 10.12 | Buyer's Lien on Certain Assets. | 2938 |
| | 10.12 | Applicability of Section 1125 of the Bankruptcy Code. | |
| | 10.15 | Applicability of Section 1125 of the Dankrupicy Code | . 27 <u>57</u> |
| ARTI | CLE XI | RETENTION OF JURISDICTION | <u> 2939</u> |
| | 11.1 | Retention of Jurisdiction | 2939 |
| | | | |
| ARTI | CLE XI | I MISCELLANEOUS PROVISIONS | <u>3040</u> |
| | 12.1 | Dissolution of the Creditors Committee. | |
| | 12.2 | Payment of Statutory Fees | |
| | 12.3 | Payment of ATI Note Trustee Fees | .3140 |
| | 12.4 | Modification of the Plan | |
| | 12.5 | Severability | .3141 |
| | 12.6 | Revocation or Withdrawal of the Plan | 3141 |
| | 12.7 | Section 1145 Exemption. | |
| | 12.8 | Section 1146 Exemption. | |
| | 12.9 | Notices | |
| | 12.10 | Governing Law | 3342 |
| | 12.10 | Withholding and Reporting Requirements | 33/13 |
| | 14.11 | winnording and reporting requirements | . |

Allegiance Telecom, Inc., Allegiance Telecom Company Worldwide, together with its direct and indirect subsidiaries, Allegiance Telecom of New York, Inc., Adgrafix Corporation, ALGX Business Internet, Inc., Allegiance Internet, Inc., Allegiance Telecom International, Inc., Allegiance Telecom of Arizona, Inc., Allegiance Telecom of Colorado, Inc., Allegiance Telecom of Florida, Inc., Allegiance Telecom of Georgia, Inc., Allegiance Telecom of Illinois, Inc., Allegiance Telecom of Indiana, Inc., Allegiance Telecom of Maryland, Inc., Allegiance Telecom of Massachusetts, Inc., Allegiance Telecom of Michigan, Inc., Allegiance Telecom of Missouri, Inc., Allegiance Telecom of New Jersey, Inc., Allegiance Telecom of North Carolina, Inc., Allegiance Telecom of Ohio, Inc., Allegiance Telecom of Visigina, Inc., Allegiance Telecom of Oregon, Inc., Allegiance Telecom of Pennsylvania, Inc., Allegiance Telecom of Texas, Inc., Allegiance Telecom of the District of Columbia, Inc., Allegiance Telecom of Virginia, Inc., Allegiance Telecom of Washington, Inc., Allegiance Telecom of Wisconsin, Inc., Allegiance Telecom of Virginia, Inc., Allegiance Telecom of Service Corporation, Coast to Coast Telecommunications, Inc., Hosting.com, Inc., InterAccess Telecoms and debtors in possession, jointly propose the following amended chapter 11 plan pursuant to section 1121(a) of title 11 of the United States Code:

ARTICLE I

DEFINITIONS AND CONSTRUCTION OF TERMS

Definitions. The following terms used in the Plan shall have the respective meanings defined below:

1.1 <u>Access Provider</u> means an Entity providing telecommunications services to the Debtors pursuant to an executory contract or a Tariff filed by such Entity with the Federal Communications Commission or a relevant state commission.

1.2 <u>Acquired Assets</u> means the "Acquired Assets" as defined under the Purchase Agreement.

1.3 Administrative Expense Claim means any right to payment, which constitutes a cost or expense of the administration of any of the Chapter 11 Cases, and is allowed under sections 503(b) and 507(a)(l) of the Bankruptcy Code, including, without limitation, any actual and necessary costs and expenses of preserving the Debtors' Estates, any actual and necessary costs and expenses of operating the Debtors' businesses after the Commencement Date, any indebtedness or obligations incurred or assumed by the Debtors, as debtors in possession, during the Chapter 11 Cases, any allowances of compensation and reimbursement of expenses to the extent allowed by a Final Order under sections 328, 330 or 503 of the Bankruptcy Code, and any fees or charges assessed against the Debtors' Estates under section 1930 of chapter 123 of title 28 of the United States Code.

1.4 <u>Administrative Expense Claim Bar Date</u> means the date that is the forty-fifth (45th) day after the Initial Effective Date.

1.5 <u>Allowed</u> means, with reference to any Claim<u></u>(a) any Claim against any Debtor that is listed by such Debtor in the Schedules, as such Schedules may be amended by the Debtors from time to time in accordance with Bankruptcy Rule 1009, as liquidated in amount and not disputed or contingent; (b) any Claim as to which no objection to allowance has been interposed and that is not listed by the Debtors in the Schedules, as may be amended, as contingent, unliquidated or disputed, in accordance with Section 8.1 of this Plan by the Debtors, the Buyer, or any other party in interest or such other applicable period of limitation fixed by the Bankruptcy Code, the Bankruptcy Rules, or the Bankruptcy Court, or as to which any objection has been determined by a Final Order to the extent such objection is determined in favor of the respective Holder; (c) any Claim as to which the liability of the Debtors and the amount thereof are determined by a Final Order of a court of competent jurisdiction other than the Bankruptcy Court; (d) any Claim expressly allowed by Debtors pursuant to a Final Order of the Bankruptcy Court; or (e) any Claim expressly allowed hereunder.

- 1.6 <u>Assumed Liabilities</u> means the "Assumed Liabilities" as defined in the Purchase Agreement.
- 1.7 <u>ATCW</u> means Allegiance Telecom Company Worldwide, a Delaware corporation.

1.8 **<u>ATCW Debtors</u>** means, collectively, ATCW and the Subsidiaries.

1.9 <u>ATCW Unsecured Claim</u> means a general unsecured claim which is a Claim, including an Insured Claim, against any of the ATCW Debtors, other than Administrative Expense Claims, Priority Tax Claims, Priority Non-Tax Claims, Secured Claims or Senior Lender Claims.

1.10 <u>ATI</u> means Allegiance Telecom, Inc., a Delaware corporation.

1.11 <u>ATI Note Claims</u> means any Claim arising under or in connection with the ATI Notes, other than the fees of the ATI Note Trustees, accruing under the ATI Note Indentures. The ATI Note Claims are hereby deemed to be Allowed ATI Unsecured Claims in the aggregate amount of [\$642,843,000]. <u>\$642,843,000</u>.

1.12 **ATI Note Indentures** means, collectively, the Senior Discount Note Indenture and the Senior Note Indenture.

1.13 <u>ATI Notes</u> means, collectively, the Senior Discount Notes and the Senior Notes.

1.14 <u>ATI Note Trustees</u> means, collectively, the Senior Discount Note Trustee and the Senior Note Trustee.

1.15 <u>ATI Unsecured Claim</u> means a general unsecured claim which is a Claim, including an Insured Claim and an ATI Note Claim, against ATI, other than Administrative Expense Claims, Priority Tax Claims, Priority Non-Tax Claims, Secured Claims or Senior Lender Claims.

1.16 <u>ATLT</u> means Allegiance Telecom Liquidating Trust or any substitute entities that constitute a means for holding the ATLT Assets.

1.17 <u>ATLT Agreement</u> means the trust agreement governing ATLT, dated as of the Initial Effective Date, substantially in the form set forth in the Plan Supplement.

1.18 <u>ATLT Assets</u> means (a) the XO Common Stock, (b) the New STFI Common Stock, (c) the Litigation Cash, (d) the Available Cash, and (e) all other assets of the Estates that are not Acquired Assets.

1.19 <u>ATLT A Certificates "A" Trust Interests</u> means the [_____] ownership interests in ATLT, which shall be issued and distributed in accordance with the Plan and which shall represent ATLT's ownership of the XO Common Stock.

1.20 <u>ATLT B-Certificates "B" Trust Interests</u> means the [_____] ownership interests in ATLT, which shall be issued and distributed in accordance with the Plan and which shall represent ATLT's ownership of the New STFI Common Stock.

1.21 <u>ATLT C Certificates "C" Trust Interests</u> means the [____] ownership interests in ATLT, which shall be issued and distributed in accordance with the Plan and which shall represent ATLT's ownership of the ATLT Assets, other than the XO Common Stock and the New STFI Common Stock.

1.22 <u>ATLT Certificates Trust Interests</u> means, collectively, the ATLT <u>A Certificates "A" Trust</u> Interests, the ATLT <u>"B-Certificates</u>" Trust Interests and the ATLT <u>"C-Certificates</u>" Trust Interests.

1.23 <u>Available Cash</u> means, on the Initial Effective Date, all Cash of the Debtors (other than Litigation Cash, Unencumbered Cash, or Cash funded by the Buyer or generated by the Debtors pursuant to the Operating Agreement) <u>less</u> the (a) Cash necessary to pay, in full, all Holders of Allowed (i) Administrative Expense Claims, (ii) Priority Tax Claims, (iii) Professional Fee Claims, (iv) Priority Non-Tax Claims, (v) Secured Claims, and (vi) Senior Lender Claims; (b) the STFI Cash Amount; (c) Cash necessary to fund the Reserve Fund; and (d) the Wind-Down Amount.

1.24 <u>Avoidance Actions</u> means any and all actual or potential avoidance claims pursuant to any applicable section of the Bankruptcy Code <u>to the extent and only to the extent such claims constitute Excluded</u> <u>Assets</u>, including, without limitation, sections 544, 545, 547, 548, 549, 550, 551, 553(b) and 724(a) of the Bankruptcy Code, arising from any transaction involving or concerning any of the Debtors that may be prosecuted in compliance with the terms and conditions of the Purchase Agreement.

1.25 **Ballot** means the ballot form distributed to all Holders of Impaired Claims on which such Holders shall indicate their acceptance or rejection of the Plan and, with respect solely to Holders of (a) Allowed ATCW Unsecured Claims and (b) Allowed ATI Unsecured Claims, if applicable, whether they shall exercise the Cash Recovery Election.

1.26 **Bankruptcy Code** means title 11 of the United States Code, as amended from time to time, as applicable to the Chapter 11 Cases.

1.27 **Bankruptcy Court** means the United States Bankruptcy Court for the Southern District of New York having jurisdiction over the Chapter 11 Cases and, to the extent of the withdrawal of any reference under section 157 of title 28 of the United States Code, the United States District Court for the Southern District of New York.

1.28 <u>Bankruptcy Rules</u> means the Federal Rules of Bankruptcy Procedure as promulgated by the United States Supreme Court under section 2075 of title 28 of the United States Code, and any Local Rules of the Bankruptcy Court.

1.29 **Bar Date** means, with respect to nongovernmental and governmental Claims against the Debtors, November 26, 2003, or any other bar date for the filing of claims established by a separate order of the Bankruptcy Court in connection with the Chapter 11 Cases.

1.30 **Business Day** means any day other than a Saturday, Sunday or any other day on which commercial banks in New York, New York are required or authorized to close by law or executive order.

1.31 **Buyer** means XO, as Buyer under the Purchase Agreement, its designee and/or any assignee of Buyer permitted by Section 9.2 of the Purchase Agreement.

1.32 **<u>Cash</u>** means legal tender of the United States of America.

1.33 <u>Cash Recovery</u> means the ratable Distribution of <u>Available</u> Cash, to the extent available, to each Holder of an (a) Allowed ATCW Unsecured Claim or (b) Allowed ATI Unsecured Claim, as and if applicable, that exercises the Cash Recovery Election in lieu of receiving its pro rata share of the ATLT <u>A</u>-<u>Certificates, ATLT B</u> <u>Certificates</u> or ATLT <u>C</u>-<u>Certificates</u> <u>B</u>" <u>Trust Interests</u>, as applicable, that would otherwise be distributed to each such Holder, in an amount equal to <u>Per50% of the aggregate value of the applicable</u> ATLT <u>A</u>-<u>Certificate</u>, <u>Per ATLT B</u>-<u>Certificate</u><u>"</u><u>Trust Interest</u>, or <u>Per50% of the aggregate value</u> <u>of the applicable</u> ATLT <u>C</u>-<u>Certificate</u><u>"B</u>" <u>Trust Interest</u>, as applicable.

1.34 <u>Cash Recovery Election</u> means the election, by the Holder of an (a) Allowed ATCW Unsecured Claim or (b) Allowed ATI Unsecured Claim, as and if applicable, to receive the Cash Recovery in lieu of receiving its pro rata share of the ATLT <u>A Certificates, ATLT B Certificates, "A" Trust Interests</u> or ATLT <u>C Certificates "B"</u> <u>Trust Interests</u>, as applicable, that would otherwise be distributed to such Holder in accordance with Sections 3.4(a) and 3.5(a) of the Plan, as applicable.

1.35 <u>Causes of Action</u> means, without limitation, any and all actions, causes of action, liabilities, obligations, rights, suits, damages, judgments, claims and demands whatsoever, whether known or unknown, existing or hereafter arising, in law, equity or otherwise, based in whole or in part upon any act or omission or other event occurring prior to the Commencement Date or during the course of the Chapter 11 Cases, including through the Initial Effective Date and the NTA Effective Dates, to the extent and only to the extent they constitute Excluded Assets or Excluded Liabilities.

1.36 <u>**Chapter 11 Cases**</u> means the cases under chapter 11 of the Bankruptcy Code commenced by the Debtors, styled *In re Allegiance Telecom, Inc., et al.*, Chapter 11 Case No. 03-13057 (RDD), which are currently pending before the Bankruptcy Court.

1.37 **Claim** means a claim, as defined in section 101(5) of the Bankruptcy Code, against a Debtor, including, but limited to (a) any right to payment from a Debtor whether or not such right is reduced to judgment, liquidated, unliquidated, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured or unsecured or (b) any right to an equitable remedy for breach of performance if such performance gives rise to a right of payment from a Debtor, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, secured or unsecured.

1.38 <u>**Claim Holder**</u> or <u>**Claimant**</u> means the Holder of a Claim.

1.39 <u>Claims Objection Deadline</u> means, for each Claim, the later of (a) 180 days after the Initial Effective Date and (b) such other period of limitation as may be specifically fixed by an order of the Bankruptcy Court for objecting to such Claim.

1.40 **<u>Class</u>** means a category of Holders of Claims or Equity Interests classified in Article III of the Plan in accordance with section 1122(a)(1) of the Bankruptcy Code.

1.41 <u>**Closing**</u> means the "Closing" as defined in the Purchase Agreement.

1.42 <u>**Commencement Date**</u> means May 14, 2003, the date on which the Debtors commenced the Chapter 11 Cases.

1.43 <u>Common Stock</u> means all issued and outstanding common stock of the Debtors as of the Commencement Date.

1.44 <u>Confirmation Date</u> means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the Bankruptcy Court's docket.

1.45 <u>Confirmation Hearing</u> means the hearing held by the Bankruptcy Court to determine the confirmation of the Plan pursuant to section 1129 of the Bankruptcy Code, as such hearing may be adjourned or continued from time to time.

1.46 <u>Confirmation Order</u> means the order of the Bankruptcy Court confirming the Plan pursuant to section 1129 of the Bankruptcy Code.

1.47 <u>Credit Agreement</u> means that certain Credit and Guaranty Agreement, dated as of February 15, 2000, as amended as of November 27, 2002, among ATCW, as borrower; all of the other Debtors, as guarantors; Goldman Sachs Credit Partners L.P., as syndication agent and sole lead arranger; General Electric Capital Corporation (as successor to Toronto Dominion (Texas), Inc.), as administrative agent; and Bank Boston, N.A. and Morgan Stanley Senior Funding, as co-documentation agents.

1.48 <u>Creditors Committee</u> means the statutory committee of unsecured creditors appointed in the Chapter 11 Cases pursuant to section 1103 of the Bankruptcy Code, as reconstituted from time to time, and its current and former members.

1.49 **Debtors** means Allegiance Telecom, Inc., Allegiance Telecom of New York, Inc., Adgrafix Corporation, ALGX Business Internet, Inc., Allegiance Internet, Inc., Allegiance TelecomCompany Worldwide, Allegiance Telecom International, Inc., Allegiance Telecom of Arizona, Inc., Allegiance Telecom of California, Inc., Allegiance Telecom of Colorado, Inc., Allegiance Telecom of Florida, Inc., Allegiance Telecom of Georgia, Inc., Allegiance Telecom of Illinois, Inc., Allegiance Telecom of Indiana, Inc., Allegiance Telecom of Maryland, Inc., Allegiance Telecom of Massachusetts, Inc., Allegiance Telecom of Michigan, Inc., Allegiance Telecom of Minnesota, Inc., Allegiance Telecom of Missouri, Inc., Allegiance Telecom of Nevada, Inc., Allegiance Telecom of New Jersey, Inc., Allegiance Telecom of North Carolina, Inc., Allegiance Telecom of Ohio, Inc., Allegiance Telecom of Oklahoma, Inc., Allegiance Telecom of Oregon, Inc., Allegiance Telecom of Pennsylvania, Inc., Allegiance Telecom of Texas, Inc., Allegiance Telecom of the District of Columbia, Inc., Allegiance Telecom of Virginia, Inc., Allegiance Telecom of Washington, Inc., Allegiance Telecom of Wisconsin, Inc., Allegiance Telecom of Virginia, Inc., Allegiance Telecom Service Corporation, Coast to Coast Telecommunications, Inc., Hosting.com, Inc., InterAccess Telecommunications Company, Jump.Net, Inc., Shared Technologies Allegiance, Inc., and Virtualis Systems, Inc.

1.50 **Debtors in Possession** means the Debtors in their capacity as debtors in possession in the Chapter 11 Cases pursuant to sections 1101, 1107(a) and 1108 of the Bankruptcy Code.

1.51 **Disclosure Schedules** shall have the meaning ascribed to such term in the Purchase Agreement.

1.52 **Disclosure Statement** means the disclosure statement relating to the Plan, as amended, supplemented, or modified from time to time, describing the Plan, that is prepared and distributed in accordance with sections 1125, 1126(b) and/or 1145 of the Bankruptcy Code and Bankruptcy Rule 3018 and/or other applicable law.

1.53 **Disclosure Statement Hearing Date** means the final date on which the Bankruptcy Court holds a hearing to consider the approval of the Disclosure Statement.

 1.54
 Disclosure Statement Order
 means that certain order of the Bankruptcy Court, dated

 [______], April 21, 2004, which is annexed to the Disclosure Statement as Exhibit D.

1.55 **Disputed Claim** means any Claim: (a) listed on the Schedules as, or proof of which is filed as, unliquidated, disputed or contingent; (b) as to which a proof of Claim designating such Claim as liquidated in amount and not contingent was not timely and properly filed; (c) as to which a Debtor, the Buyer or any other party in interest has interposed a timely objection or request for estimation in accordance with the Bankruptcy Code and the Bankruptcy Rules; or (d) is otherwise disputed by a Debtor, the Buyer or any other party in interest in accordance with applicable law, which objection, request for estimation or dispute has not been withdrawn or determined by a Final Order.

1.56 **Disputed Claim Reserve** means the reserve established and maintained by ATLT on account of Disputed Claims.

1.57 **Disputed Claim Amount** means the amount set forth on the proof of Claim relating to a Disputed Claim or, if an amount is estimated in respect of a Disputed Claim in accordance with section 502(c) of the Bankruptcy Code, Bankruptcy Rule 3018 or other applicable law, the amount so estimated pursuant to a Final Order of the Bankruptcy Court.

1.58 **Distribution** means any distribution made under the Plan to the Holders of Allowed Claims.

1.59 **Distribution Account** means an account to be established and maintained by ATLT for the purpose of making Distributions to Holders of Allowed Claims.

1.60 **Distribution Date** means any date on which a Distribution is made by ATLT on behalf of the Debtors.

1.61 **Entity** shall have the meaning set forth in section 101(15) of the Bankruptcy Code.

1.62 **Equity Interest** means any share of common stock, preferred stock, or other instrument evidencing an ownership interest in any of the Debtors, whether or not transferable, and any option, warrant or right, contractual or otherwise, to acquire any such interest.

1.63 <u>Estates</u> means the estates created in the Chapter 11 Cases by operation of section 541 of the Bankruptcy Code.

1.64 **Excluded Assets** means the "Excluded Assets" as defined in the Purchase Agreement.

1.65 **Excluded Liabilities** means the "Excluded Liabilities" as defined in the Purchase Agreement.

1.66 **Federal Judgment Rate** means the rate equal to the weekly average one-year constant maturity Treasury yield, as published by the Board of Governors of the Federal Reserve System and identified at <u>http://www.federal reserve.gov/ releases/h15/current/</u>.

1.67 **Final Distribution Date** means the date of the last Distribution by ATLT on behalf of the Debtors.

1.68 **Final Effective Date** means the date on which the Initial Effective Date and all of the NTA Effective Dates have occurred.

1.69 **Final Order** means an order of the Bankruptcy Court or any other court of competent jurisdiction as to which the time to appeal, petition for *certiorari*, or move for reargument or rehearing has expired and as to which no appeal, petition for *certiorari*, or other proceedings for reargument or rehearing shall then be pending or as to which any right to appeal, petition for *certiorari*, reargument, or rehearing shall have been waived in writing in form and substance satisfactory to the Debtors or, on and after the Initial Effective Date, ATLT, in the event that an appeal, writ of *certiorari*, or reargument or rehearing thereof has been filed or sought, such order of the Bankruptcy Court or other court of competent jurisdiction shall have been determined by the highest court to which such order was appealed, or *certiorari*, reargument or rehearing shall have been denied and the time to take any further appeal, petition for *certiorari* or move for reargument or rehearing shall have expired; provided, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure or any analogous rule under the Bankruptcy Rules or applicable state court rules of civil procedure, may be filed with respect to such order shall not cause such order not to be a Final Order.

1.70 <u>Holder</u> and, collectively, <u>Holders</u> means a Person or Entity legally holding a Claim or Equity Interest.

1.71 **Impaired** means, with respect to a Claim or Equity Interest (or Class of Claims or Equity Interests), a Claim or Equity Interest (or Class of Claims or Equity Interests) that is impaired within the meaning of section 1124 of the Bankruptcy Code.

1.72 **Impaired Claim** means a Claim classified in an Impaired Class.

1.73 **Impaired Class** means each of Classes 4, 5, 6 and 7 as set forth in Article III of the Plan.

1.74 **Initial Effective Date** means, with respect to ATI, ATCW and any Subsidiaries that do not hold or constitute Non-Transferred Assets, the first Business Day on which (a) the conditions specified in Section 9.1 of the Plan have been satisfied or waived; and (b) no stay of the Confirmation Order is in effect.

1.75 **Insured Claim** means any Claim arising from an incident or occurrence that is covered under the Debtors' insurance policies.

1.76 **Intercompany Claim** means a Claim of one Debtor against another Debtor.

1.77 **IT Reorganized Subsidiaries** means the Subsidiaries, other than STFI, that do not hold or constitute Non-Transferred Assets, on and after the Initial Effective Date.

1.78 <u>Lien</u> means any lien, lease, right of first refusal, servitude, claim, pledge, option, charge, hypothecation, easement, security interest, right-of-way, encroachment, mortgage, deed of trust and/or any other encumbrance, restriction or limitation whatsoever.

1.79 **Litigation Cash** means all proceeds, if any, actually received by ATLT arising from the prosecution or settlement of Avoidance Actions or Causes of Action commenced against any Entity or Person at the discretion of ATLT; <u>provided</u>, <u>however</u>, that the prosecution of any such Avoidance Action or Causes of Action shall be subject to the terms and conditions of the Purchase Agreement.

1.80 <u>New STFI Common Stock</u> means the shares of common stock, par value \$0.01 per share, to be authorized pursuant to the New Certificate of Incorporation of Reorganized STFI, of which up to [10,000,000] shares shall be issued pursuant to the Plan to ATLT.

1.81 **<u>Non-Transferred Assets</u>** means "Non-Transferred Assets" as defined in the Purchase Agreement.

1.82 <u>NTA Effective Dates</u> means, with respect to each Subsidiary that holds or constitutes Non-Transferred Assets, the first Business Day, or as soon as practicable thereafter, on which (a) the conditions specified in Section 9.2 of the Plan have been satisfied or waived with respect to such Subsidiary; and (b) no stay of the Confirmation Order is in effect.

1.83 **<u>NTA Reorganized Subsidiaries</u>** means the Subsidiaries, other than STFI, that hold or constitute Non-Transferred Assets, on and after their respective NTA Effective Dates.

1.84 **Operating Agreement** means the "Operating Agreement" as defined in the Purchase Agreement.

1.85 **Person** shall have the meaning set forth in section 101(41) of the Bankruptcy Code.

1.86 **Plan** means this Joint Plan of Reorganization of the Debtors pursuant to chapter 11 of the Bankruptcy Code, including without limitation, all exhibits, supplements, amendments, appendices and schedules hereto, either in their present form or as the same may be altered, amended or modified from time to time, and those documents identified in the Plan Supplement.

1.87 **<u>Plan Administrator</u>** means the person appointed by the Creditors Committee to administer the ATLT pursuant to the terms of the ATLT Agreement.

1.88 **Plan Supplement** means the supplement to the Plan that will consist of the following form of documents: (a) the Operating Agreement; (b) the Transition Services Agreement; (c) the ATLT Agreement; (d) the ATLT Certificates; (e) the Reorganized STFI Certificate of Incorporation: (e) the Reorganized STFI Bylaws; (f) the Reorganized Subsidiaries' Certificates of Incorporation; and (hg) the Reorganized Subsidiaries' Bylaws.

1.89 **Post-Initial Effective Date Fees and Expenses** means the fees and expenses in an amount equal to the aggregate of the unpaid fees and expenses requested by Professionals for services rendered to the Debtors and the Creditors Committee through the Initial Effective Date that shall be maintained in the Reserve Fund. It also includes fees and expenses in an amount equal to the anticipated additional fees and expenses for services to be rendered after the Initial Effective Date to ATLT by Professionals, or any other professionals ATLT may retain, estimated through the comp letion date of such services, together with all anticipated additional expenses for consummating the liquidation and Distribution of the ATLT Assets (including but not limited to the costs and expenses associated with operating ATLT, the cost of any insurance, United States Trustee's fees, tax obligations and other post-confirmation expenses) and other transactions or obligations contemplated by the Plan.

1.90 **Priority Non-Tax Claim** means any Claim, other than an Administrative Expense Claim or a Priority Tax Claim, entitled to priority in right of payment under section 507(a) of the Bankruptcy Code.

1.91 **Priority Tax Claim** means any Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.

1.92 **Professional** means (a) any professional employed in the Chapter 11 Cases pursuant to sections 327 or 1103 of the Bankruptcy Code and (b) any professional or other Entity seeking compensation and reimbursement in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code.

1.93 **Professional Fee Claims** means all Administrative Expense Claims for the compensation of Professionals and reimbursement of expenses incurred by such Professionals (to the extent allowed under sections 328, 330, 363 or 503 of the Bankruptcy Code) through the Initial Effective Date.

1.94 **Purchase Agreement** means that certain Asset Purchase Agreement, dated as of February 18, 2004, by and among ATI, ATCW and Buyer and approved by the Bankruptcy Court pursuant to the Sale Order, a copy of which Purchase Agreement is annexed to the Disclosure Statement as Exhibit B.

1.95 **Record Date** means, for purposes of voting on the Plan, the date on which the order of the Bankruptcy Court approving the Disclosure Statement is entered on the docket of the Bankruptcy Court and, for purposes of receiving a Distribution under the Plan, the Confirmation Date.

1.96 **Reorganized STFI** means STFI on and after the Initial Effective Date.

1.97 **<u>Reorganized Subsidiaries</u>** means, collectively, the IT Reorganized Subsidiaries and the NTA Reorganized Subsidiaries.

1.98 **Reserve Fund** means the account established by the Debtors or ATLT into which funds shall be deposited on, and possibly after, the Initial Effective Date and from which Post-Initial Effective Date Fees and Expenses shall be paid in an amount agreed to by the Debtors and the Creditors Committee.

1.99 <u>Sale Order</u> means that certain order entered by the Bankruptcy Court on February 20, 2004 in connection with the Chapter 11 Cases, which, among other things, approved the Purchase Agreement, and which is annexed to the Disclosure Statement as Exhibit C.

1.100 <u>Sale Transaction</u> means the sale of substantially all of the assets of ATI and ATCW and<u>, as</u> <u>applicable</u>, the common stock of the Reorganized Subsidiaries to Buyer, and the assumption by Buyer of the Assumed Liabilities of ATI and ATCW, as contemplated in accordance with the Purchase Agreement.

1.101 <u>Schedules</u> means the schedule of assets and liabilities, the list of Holders of Equity Interests and the statement of financial affairs filed by the Debtors under section 521 of the Bankruptcy Code and Bankruptcy Rule 1007, and all amendments and modifications thereto.

1.102 <u>Secured Claim</u> means any Claim, other than Senior Lender Claims, to the extent reflected in the Schedules or on a timely filed proof of Claim as a Secured Claim, which is secured by a Lien on Collateral to the extent of the value of such Collateral, as determined in accordance with section 506(a) of the Bankruptcy Code, or, in the event that such Claim is subject to a permissible set-off under section 553 of the Bankruptcy Code, to the extent of such permissible set-off.

1.103 <u>Senior Discount Notes</u> means those certain 11 3/4% Senior Discount Notes, due on February 15, 2008, issued pursuant to the Senior Discount Note Indenture.

1.104 <u>Senior Discount Note Indenture</u> means that certain Indenture, dated as of February 3, 1998, by and between ATI and the Senior Discount Note Trustee.

1.105 <u>Senior Discount Note Trustee</u> means The Bank of New York, as Indenture Trustee for the Senior Discount Notes.

1.106 <u>Senior Lender Claims</u> means claims in the aggregate amount of \$477,827,051, together with any post-Commencement Date interest and costs outstanding as of the date of payment, arising under or in connection with the Credit Agreement.

1.107 <u>Senior Lenders</u> means the lenders under the Credit Agreement.

1.108 <u>Senior Note Indenture</u> means that certain Indenture, dated as of July 7, 1998, by and between ATI and the Senior Note Trustee.

1.109 <u>Senior Note Trustee</u> means The Bank of New York, as Indenture Trustee for the Senior Notes.

1.110 <u>Senior Notes</u> means those 12 7/8% Senior Notes, due on May 15, 2008, issued pursuant to the Senior Note Indenture.

1.111 **STFI** means Shared Technologies Allegiance, Inc. before the Initial Effective Date.

1.112 **STFI Assets** means the capital stock of STFI and the assets of STFI and certain other Debtors that are (a) utilized exclusively in the business of STFI, (b) are necessary to operate Reorganized STFI, and (c) and as more specifically set forth in the Plan Supplement; provided, however, that the STFI Assets shall not include any Acquired Assets.

1.113 **STFI Cash Amount** means the amount of Cash necessary to fund the operation of the STFI Assets on and after the Initial Effective Date, which amount shall be not less than \$5 million, unless otherwise determined by the Debtors and the Creditors Committee.

1.114 **Subordinated Claim** means any Claim arising from rescission of a purchase or sale of a security of any of the Debtors, for damages arising from the purchase or sale of such a security, or for reimbursement or contribution allowed under section 502 of the Bankruptcy Code on account of such a Claim that is determined to be subordinated to other Claims pursuant to section 510(b) of the Bankruptcy Code.

1.115 **<u>Subsidiaries</u>** means the direct and indirect subsidiaries of ATCW.

1.116 **<u>Tariff</u>** means the schedule of terms, conditions, and prices (a) which are filed with an appropriate regulatory commission or (b) which are made generally available to the public (or such classes of customers as to be effectively available directly to the public) for the provision of products and services.

1.117 **Tariff Services** means telecommunications services required to be provided by an Access Provider pursuant to a Tariff filed by such Access Provider with the Federal Communications Commission or relevant state commission. For purposes of the Plan, the obligation of an Access Provider to provide Tariff Services does not arise under an executory contract.

1.118 <u>**Tax Rate**</u> means the rate equal to the underpayment rate specified in 26 U.S.C. § 6621 (determined without regard to 26 U.S.C. §6621(c)) as of the Initial Effective Date.

1.119 <u>**Transition Services Agreement**</u> means the "Transition Services Agreement" as defined in the Purchase Agreement.

1.120 <u>Unencumbered Cash</u> means Cash in the amount of \$26,185,395 that is being held in JP Morgan Chase account number 343605.

1.121 **<u>Unimpaired</u>** means, with respect to a Claim (or Class of Claims), a Claim (or Class of Claims) that is Unimpaired within the meaning of section 1124 of the Bankruptcy Code.

1.122 <u>Unimpaired Claim</u> means a Claim classified in an Unimpaired Class.

1.123 <u>Unimpaired Class</u> means each of Classes 1, 2 and 3 as set forth in Article III of the Plan.

1.124 <u>Unsecured Claims</u> means, collectively, all ATCW Unsecured Claims and all ATI Unsecured Claims.

1.125 <u>Utility Company</u> means a Person who provides Utility Services to Debtors, including, but not limited to, those Persons defined as Utility Companies in the Utilities Order.

1.126 <u>Utilities Order</u> means the Order Deeming Utilities Adequately Assured of Future Performance and Establishing Procedures for Determining Requests for Additional Adequate Assurance that was entered by the Bankruptcy Court on May 15, 2003.

1.127 <u>Utility Services</u> means those services generally provided by utility providers and telecommunications vendors pursuant to a Tariff requested by the Debtor via a Utility Service Order, including, but not limited to, electricity, gas, water, telephone, telecommunications, Utility Services as defined in the Utilities Order, and other utility services.

1.128 <u>Utility Service Order</u> means those orders, purchase orders, and other requests for Utility Services made by Debtors.

1.129 **<u>Voting Deadline</u>** means the date set by the Bankruptcy Court by which all Ballots for acceptance or rejection of the Plan must be received by the Debtors.

1.130 <u>Voting Record Date</u> means the record date set by the Bankruptcy Court, pursuant to Bankruptcy Rule 3017(d), for determining which creditors and equity security holders are entitled to receive solicitation materials and, when applicable, to vote on the Plan.

1.131 **Wind-Down Amount** shall mean Cash in the amount necessary to fund the wind-down and closing of the Debtors' Estates by ATLT in accordance with the Plan, which amount shall be estimated by the Debtors, in consultation with the Creditors Committee, no later than ten (10) days prior to the Confirmation Date and shall be deposited in a separate interest-bearing account.

1.132 <u>XO</u> means XO Communications, Inc., a Delaware corporation.

1.133 **XO Common Stock** means 45,380,000 shares of common stock of XO, par value \$.01 per share, to be issued to the Debtors pursuant to the Purchase Agreement, subject to adjustment as provided therein.

Interpretation, Application of Definitions and Rules of Construction. Wherever from the context it appears appropriate, each term stated in either the singular or the plural shall include both the singular and the plural and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, feminine and neuter. Unless otherwise specified, all section, schedule, or exhibit references in the Plan are to the respective section in, article of, or schedule or exhibit to the Plan, as the same may be amended, waived, or modified from time to time. The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Plan as a whole and not to any particular section, subsection, or clause contained therein. A term used in the Plan that is not defined in the Plan shall have the meaning assigned to that term in the Bankruptcy Code. The rules of construction contained in section 102 of the Bankruptcy Code shall apply to the Plan. The headings in the Plan are for convenience of reference only and shall not limit or otherwise affect the provisions hereof. In the event of any inconsistency between the terms of the Purchase Agreement and this Plan, the terms of the Purchase Agreement shall control.

In computing any period of time prescribed or allowed by the Plan, the provisions of Bankruptcy Rule 9006(a) will apply. Unless otherwise provided in this Plan, any reference in this Plan to a contract, instrument, release or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions. Any reference in this Plan to an existing document or schedule filed or to be filed means such document or schedule, as it may have

been or may be amended, modified or supplemented pursuant to this Plan. Any reference to an Entity as a Holder of a Claim or Equity Interest includes that Entity's legal successors and assigns.

This Plan is the product of extensive discussions and arm's-length negotiations between and among the Debtors, the Senior Lenders, the Creditors Committee and certain other creditors and constituencies. Each of the foregoing was represented by counsel who either (a) participated in the formulation and documentation of or (b) was afforded the opportunity to review and provide comments on, the Plan, the Disclosure Statement, and the other relevant and necessary documents ancillary **thereto**, as applicable. Additionally, the Buyer and its counsel were afforded the opportunity to review and provide comments on, the Plan and Disclosure Statement.

ARTICLE II

TREATMENT OF ADMINISTRATIVE EXPENSE CLAIMS AND PRIO RITY TAX CLAIMS

21 <u>Administrative Expense Claims</u>. Except to the extent that any Entity entitled to the payment of any Allowed Administrative Expense Claim agrees to a less favorable treatment, each Holder of an Allowed Administrative Expense Claim shall receive Cash in an amount equal to such Allowed Administrative Expense Claim on the later of (a) the Initial Effective Date, (b) the date such Administrative Expense Claim becomes an Allowed Administrative Expense Claim, or as soon as practicable thereafter, or (c) on such other date as may be ordered by the Bankruptcy Court; provided, however, that Allowed Administrative Expense Claims representing liabilities incurred in the ordinary course of business by the Debtors (including, but not limited to, real and personal property taxes and franchise fees) or liabilities arising under loans or advances to or other obligations incurred by the Debtors shall be paid in full and performed by the responsible Debtor, Reorganized Subsidiary or Reorganized STFI, as the case may be, in the ordinary course of business in accordance with the terms and subject to the conditions of any agreements governing, instruments evidencing or other documents relating to such transactions. Notwithstanding the foregoing, the Buyer and the Reorganized Subsidiaries shall not be liable (x) for any of the foregoing liabilities to the extent such liabilities do not constitute Assumed Liabilities, (y) for more than 50% of all Transfer Taxes (as defined in the Purchase Agreement) in accordance with Section 6.10 of the Purchase Agreement and (z) any real or personal property Taxes (as defined in the Purchase Agreement) or similar ad valorem obligations other than those to be borne by Buyer in accordance with the proration provided under Section 6.12 of the Purchase Agreement. Except as provided under applicable non-bankruptcy law, post-petition interest will not be paid on Allowed Administrative Expense Claims.

2.2 **Professional Fee Claims**. The Holders of Professional Fee Claims shall file their respective final fee applications for the allowance of compensation for services rendered and reimbursement of expenses incurred through the Initial Effective Date by no later than the date that is forty-five (45) days after the Initial Effective Date, or such other date that may be fixed by the Bankruptcy Court. If granted by the Bankruptcy Court, such award shall be paid in full in such amounts as are Allowed by the Bankruptcy Court either (a) on the date such Professional Fee Claim becomes an Allowed Professional Fee Claim, or as soon as practicable thereafter, or (b) on such date and upon such other terms as may be mutually agreed upon between such Holder of an Allowed Professional Fee Claim and the Debtors. Notwithstanding anything to the contrary contained herein, none of neither the Buyer nor any of the Reorganized Subsidiaries shall be liable for any Professional Fee Claims <u>. Requests for compensation under section 503(b) of the Bankruptcy Code must be filed with the Bankruptcy Court and served on the Debtors, the Creditors Committee, the Buyer and other parties in interest by the Administrative Expense Claims Bar Date.</u>

2.3 **Priority Tax Claims**. Except to the extent that a Holder of an Allowed Priority Tax Claim has been paid by the Debtors prior to the Initial Effective Date or agrees to a different treatment, each Holder of an Allowed Priority Tax Claim shall receive, in full and complete settlement, satisfaction and discharge of its Allowed Priority Tax Claim, including post-petition interest, (a) Cash in an amount equal to such Allowed Priority Tax Claim on the later of (i) the Initial Effective Date, (ii) the date such Priority Tax Claim becomes an Allowed Priority Tax Claim, or as soon as practicable thereafter, and (iii) on such other date as may be ordered by the Bankruptcy Court, or (b) equal annual Cash payments in an aggregate amount equal to such Allowed Priority Tax Claim, together with interest at the Tax Rate, over a period through the sixth (6th) anniversary of the date of assessment of such Allowed

Priority Tax Claim. Notwithstanding anything to the contrary contained herein, none of <u>neither</u> the Buyer nor any of the Reorganized Subsidiaries shall be liable for any Priority Tax Claims.

ARTICLE III

SUMMARY CLASSIFICATION AND TREATMENT OF CLAIMS AND EQUITY INTERESTS

Claims, other than Administrative Expense Claims, Professional Fee Claims and Priority Tax Claims, are classified for all purposes, including voting, confirmation and Distribution pursuant to the Plan as follows:

| Class | Designation | Impairment | Entitled to Vote |
|---------|-------------------------|------------|-----------------------|
| Class 1 | Priority Non-Tax Claims | Unimpaired | No (deemed to accept) |
| Class 2 | Secured Claims | Unimpaired | No (deemed to accept) |
| Class 3 | Senior Lender Claims | Unimpaired | No (deemed to accept) |
| Class 4 | ATCW Unsecured Claims | Impaired | Yes |
| Class 5 | ATI Unsecured Claims | Impaired | Yes |
| Class 6 | Subordinated Claims | Impaired | No (deemed to reject) |
| Class 7 | Equity Interests | Impaired | No (deemed to reject) |

A Claim or Interest is classified in a particular Class for voting and distribution purposes only to the extent the Claim or Interest qualifies within the description of that Class, and is classified in another Class or Classes to the extent that the Claim or Interest qualifies within the description of such other Class or Classes. A Claim or Interest is classified in a particular Class regardless of whether the Claim or Interest is an Allowed Claim or Interest in that Class, or only asserted as such, but only to the extent that it has not been paid, released, disallowed or otherwise satisfied before the Initia l Effective Date.

The Allowed Claims against and Interests in the Debtors shall be classified and receive the treatment specified below.

The Buyer and the Reorganized Subsidiaries shall not be liable for any Claims against or Equity Interests in the Debtors, except for Assumed Liabilities.

3.1 Priority Non-Tax Claims (Class 1).

- (a) Treatment: On the later of (i) the Initial Effective Date, (ii) the date such Priority Non-Tax Claim becomes an Allowed Priority Non-Tax Claim, or as soon thereafter as is practicable, or (iii) such other date as may be ordered by the Bankruptcy Court, each Holder of an Allowed Priority Non-Tax Claim shall be paid in Cash, in full satisfaction, settlement, release and discharge of, and in exchange for such Allowed Priority Non-Tax Claim, and thereby rendered Unimpaired, except to the extent that the Debtors, with the consent of the Creditors Committee which shall not be unreasonably withheld, delayed or conditioned, and any Holder of such Allowed Priority Non-Tax Claim agree to a different treatment.
- (b) **Voting:** Class 1 is Unimpaired by the Plan. Pursuant to section 1126(f) of the Bankruptcy Code, each Holder of an Allowed Priority Non-Tax Claim in Class 1 is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan.

3.2 Secured Claims (Class 2).

- (a) Treatment: On the later of (i) the Initial Effective Date, (ii) the date on which a Secured Claim becomes an Allowed Secured Claim, or as soon thereafter as practicable, or (iii) such other date as may be ordered by the Bankruptcy Court, each Allowed Secured Claim shall be, at the election of the Debtors (i) reinstated (only to the extent the collateral securing such Secured Claim does not constitute an Acquired Asset), (ii) paid in Cash, in full satisfaction, settlement, release and discharge of and in exchange for such Allowed Secured Claim to the collateral securing such Allowed Secured Date interest, (iii) satisfied by the Debtors' surrender of the collateral securing such Allowed Secured Claim unless such collateral is an Acquired Asset under the Purchase Agreement, in which case the Lien of such Holder shall attach to the proceeds of such Acquired Assets, (iv) offset against, and to the extent of, the Debtors' claims against the Holder of such Allowed Secured Claim, or (v) otherwise rendered Unimpaired, except to the extent that the Debtors, with the consent of the Creditors Committee which shall not be unreasonably withheld, delayed or conditioned, and a Holder of an Allowed Secured Claim agree to a different treatment.
- (b) **Voting:** Class 2 is Unimpaired by the Plan. Pursuant to section 1126(f) of the Bankruptcy Code, each Holder of an Allowed Secured Claim in Class 2 is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan.

3.3 Senior Lender Claims (Class 3).

- (a) <u>**Treatment:**</u> On the Initial Effective Date, or as soon as practicable thereafter, each Holder of an Allowed Senior Lender Claim shall receive Cash in an amount equal to one hundred percent (100%) of such Holder's Allowed Senior Lender Claim.
- (b) Voting: Class 3 is Unimpaired by the Plan. Pursuant to section 1126(f) of the Bankruptcy Code, each Holder of an Allowed Secured Claim in Class 3 is conclusively presumed to have accepted the Plan and is not entitled to vote to accept or reject the Plan. In the event that there is not sufficient Cash to pay Holders of Allowed Senior Lender Claims in full on the Initial Effective Date, pursuant to the ATLT Agreement, ATLT shall sell such shares of XO Common Stock as are necessary to pay the Holders of Senior Lender Claims in full. Such sales of XO Common Stock shall be conducted in a commercially reasonable manner as determined by the Plan Administrator.

3.4 ATCW Unsecured Claims (Class 4).

Treatment. On the later of (i) the Initial Effective Date, (ii) the date such ATCW Unsecured (a) Claim becomes an Allowed ATCW Unsecured Claim, or as soon as practicable thereafter, or (iii) such other date the Bankruptcy Court may order, each Holder of an Allowed ATCW Unsecured Claim shall receive its pro rata share of (x) either the ATLT A Certificates"A" Trust Interests or the applicable Cash Recovery, (y) either the ATLT <u>"B-Certificates</u>" Trust Interests or the applicable Cash Recovery, and (z) either the ATLT "C Certificates or the applicable Cash Recovery. Holders of Claims in Class 4 that do not make an election with respect to the Cash Recovery will be deemed to have made the Cash Recovery Election. For purposes of this Section 3.4. the term pro rata shall mean the ratio of such Holder's Allowed ATCW Unsecured Claim over the sum of all (1) Allowed ATCW Unsecured Claims and (2) Allowed ATI Unsecured Claims. Notwithstanding the foregoing, to the extent a Holder of an Allowed ATCW Unsecured Claim is entitled to less than one (1) ATLT A Certificate. ATLT B Certificate or ATLT C Certificate, then such Holder shall be deemed to have made the Cash Recovery Election, as applicable. The shares of ATLT A Certificates, ATLT B Certificates and ATLT C Certificates otherwise distributable to Holders of Class 4 Claims that elect and receive a Cash Recovery will be extinguished" Trust Interests. In the event that the Available Cash is insufficient to satisfy in full the Claims of all Holders in Class 4 that elect a Cash Recovery (with such insufficiency being determined as of the Initial Effective Date), to satisfy in full the Claims of all Holders in Class 4 that opted to

receive a Cash Recovery (either through an affirmative determination not to elect to receive ATLT Trust Interests or through the default mechanism of the Cash Recovery Election) each such Holder shall receive its proportionate share of the that elected, or was deemed to have elected, a Cash Recovery, and shall be satisfied with ATLT A Certificates receive its Claims portion of the remainderCash Recovery, ATLT B Certificates and ATLT C Certificates, as if applicable-, as follows: first to those Holders that elected to receive a Cash Recovery in lieu of the receipt of ATLT "B" Trust Interests until all such Holders' Cash Recovery has been satisfied in full, provided that if the Available Cash is insufficient to satisfy the Cash Recovery for all such Holders, the Available Cash shall be divided pro rata among such Holders, and provided further that the remainder of such Holders' Claims shall be satisfied with ATLT "B" Trust Interests; and second to those Holders that elected to receive a Cash Recovery in lieu of the receipt of ATLT "A" Trust Interests until all such Holders' Cash Recovery has been satisfied in full, provided that if the Available Cash is insufficient to satisfy the Cash Recovery for all such Holders, the Available Cash shall be divided pro rata among such Holders, and provided further that the remainder of such Holders' Claims shall be satisfied with ATLT "A" Trust Interests

<u>Holders of Claims in Class 4 that do not make an election with respect to the Cash Recovery will</u> <u>be deemed to have made the Cash Recovery Election. For purposes of this Section 3.4, the term</u> <u>pro rata shall mean the ratio of such Holder's Allowed ATCW Unsecured Claim over the sum of</u> <u>all (1) Allowed ATCW Unsecured Claims and (2) Allowed ATI Unsecured Claims.</u>

(b) **Voting.** Class 4 is Impaired by the Plan. Each Holder of an Allowed ATCW Unsecured Claim in Class 4 is entitled to vote to accept or reject the Plan.

3.5 ATI Unsecured Claims (Class 5).

(a) Treatment. On the later of (i) the Initial Effective Date, (ii) the date such ATI Unsecured Claim becomes an Allowed ATI Unsecured Claim, or as soon as practicable thereafter, or (iii) such other date the Bankruptcy Court may order, each Holder of an Allowed ATI Unsecured Claim shall receive its pro rata share of (x) either the ATLT "A-Certificates" Trust Interests or the applicable Cash Recovery, (y) either the ATLT "B-Certificates" Trust Interests or the applicable Cash Recovery, and (z) the ATLT "C" Trust Interests. In the event that the Available Cash is insufficient to satisfy in full the Claims of all Holders in Class 4 that elect. or were deemed to have elected, a Cash Recovery (with such insufficiency being determined as of the Initial Effective Date), the Holders of Claims in Class 5 will not be entitled to any Cash Recovery. In the event that the Available Cash is insufficient (with such insufficiency being determined as of the Initial Effective Date) to satisfy in full the Claims of all Holders in Class 4 that elect, or are deemed to have elected, a Cash Recovery (either through an affirmative determination not to elect to receive ATLT Trust Interests or through the ATLT C Certificates or default mechanisms of the applicable Cash Recovery Election;). the Holders of Claims in Class 5 will not be entitled to any Cash Recovery. If the Available Cash is sufficient to satisfy in full the Claims of all Holders in Class 4 that a elect, or are deemed to have elected, a Cash Recovery (with such sufficiency being determined as of the Initial Effective Date) then Holders of Claims in Class 5 that elected, or were deemed to have elected, a Cash Recovery will be entitled to such Cash Recovery, provided, however, that in the event that the Available Cash is insufficient to satisfy in full the Claims of all Holders in Class 45 that elect a Cash Recovery (with such insufficiency being determined as of the Initial Effective Date), the Holders of Allowed Claims in Class 5 will not be entitled to any Cash Recovery. Holders of Claims in Class 5 that do not make an election with respect to the Cash Recovery will be deemed to have made the Cash Recovery Election. For purposes of this Section 3.5, the term pro rata shall mean the ratio of such Holder's Allowed ATI Unsecured Claim over the sum of all (1) Allowed ATI Unsecured Claims and (2) Allowed ATI Unsecured Claims. Notwithstanding the foregoing, to the extent a Holder of an Allowed ATI Unsecured Claim is entitled to less than one (1) ATLT A Certificate, ATLT B Certificate or ATLT C Certificate, then such Holder shall be

deemed to have made the Cash Recovery Election, as applicable. In the event that there is sufficient Available Cash to permit Holders of Allowed Claims in Class 5 to make a Cash Recovery Election, the shares of ATLT A Certificates, ATLT B Certificates, and ATLT C Certificates otherwise distributable to Holders of Class 5 Claims that elect and receive a Cash Recovery will be extinguished. In the event that the Available Cash is insufficient to satisfy in full the Claims of all Holders in Class 5 that elect a Cash Recovery (with such insufficiency being determined as of the Initial Effective Date), each such Holder shall receive its proportionate share of the that elected a Cash Recovery, and the remainder of its Claims shall be satisfied with ATLT A Certificates receive its portion of the Cash Recovery, if applicable, as follows: first to those Holders that elected to receive a Cash Recovery in lieu of the receipt of ATLT "B-Certificates and ATLT C Certificates" Trust Interests until all such Holders' Cash Recovery has been satisfied in full, as applicable provided that if the Available Cash is insufficient to satisfy the Cash Recovery for all such Holders, the Available Cash shall be divided pro rata among such Holders, and provided further that the remainder of such Holders' Claims shall be satisfied with ATLT "B" Trust Interests; and second to those Holders that elected to receive a Cash Recovery in lieu of the receipt of ATLT "A" Trust Interests until all such Holders' Cash Recovery has been satisfied in full, provided that if the Available Cash is insufficient to satisfy the Cash Recovery for all such Holders, the Available Cash shall be divided pro rata among such Holders, and provided further that the remainder of such Holders' Claims shall be satisfied with ATLT "A" Trust Interests.

<u>Holders of Claims in Class 5 that do not make an election with respect to the Cash Recovery will</u> <u>be deemed to have made the Cash Recovery Election. For purposes of this Section 3.5, the term</u> <u>pro rata shall mean the ratio of such Holder's Allowed ATI Unsecured Claim over the sum of all</u> (1) Allowed ATI Unsecured Claims and (2) Allowed **ATCW Unsecured Claims.**

- (b) **Voting.** Class 5 is Impaired by the Plan. Each Holder of an Allowed ATI Unsecured Claim in Class 5 is entitled to vote to accept or reject the Plan.
- 3.6 Subordinated Claims (Class 6).
- (a) **Treatment.** Holders of Subordinated Claims, if any, shall not be entitled to, and shall not retain, any property or interest in property on account of such Subordinated Claims.
- (b) <u>Voting</u>. Holders of Subordinated Claims shall receive no Distribution under the Plan. Therefore, each Holder of a Subordinated Claim in Class 6 is conclusively presumed to have rejected the Plan. Pursuant to section 1126(g) of the Bankruptcy Code, Holders of Subordinated Claims are not entitled to vote to accept or reject the Plan. <u>Interests (Class 7) Interests (Class 7)Equity</u>.
- (a) **<u>Treatment</u>**. Holders of Equity Interests shall not be entitled to, and shall not retain, any property or interest in property on account of such Equity Interests.
- (b) <u>Voting</u>. Holders of Equity Interests shall receive no Distribution under the Plan. Therefore, each Holder of an Equity Interest in Class 7 is conclusively presumed to have rejected the Plan. Pursuant to section 1126(g) of the Bankruptcy Code, Holders of Equity Interests are not entitled to vote to accept or reject the Plan.

ARTICLE IV

ACCEPTANCE OR REJECTION OF THE PLAN

4.1 <u>Special Provisions Governing Unimpaired Claims.</u> Except as otherwise provided in the Plan or in the Purchase Agreement, nothing shall affect the Debtors' or ATLT's rights and defenses with respect to any Unimpaired Claims, including, but not limited to, all rights with respect to legal and equitable defenses to set-offs

and recoupments (solely in respect of assets that do not constitute Acquired Assets) against such Unimpaired Claims.

4.2 <u>Classes Entitled to Vote</u>. Classes 4 and 5 are Impaired Classes and are entitled to vote to accept or reject the Plan. By operation of law, each Unimpaired Class of Claims is deemed to have accepted the Plan and is not entitled to vote to accept or reject the Plan. Classes 1, 2 and 3 are Unimpaired Classes and are deemed to have accepted the Plan. By operation of law, any Class of Claims or Equity Interests that is not entitled to receive or retain any property of the Debtors under the Plan is deemed to have rejected the Plan. Classes 6 and 7 are not entitled to receive or retain any property of the Debtors, are not entitled to vote and are deemed to have rejected the Plan.

4.3 <u>Acceptance by Impaired Classes</u>. An Impaired Class of Claims shall have accepted the Plan if (a) the Holders (other than any Holder designated under section 1126(e) of the Bankruptcy Code) of at least twothirds (2/3) in amount of the Allowed Claims actually voting in such Class have voted to accept the Plan and (b) the Holders (other than any Holder designated under section 1126(e) of the Bankruptcy Code) of more than one-half (1/2) in number of the Allowed Claims actually voting in such Class have voted to accept the Plan.

4.4 <u>**Cramdown**</u>. The Debtors will request Confirmation of the Plan, as it may be modified from time to time, under section 1129(b) of the Bankruptcy Code, with respect to any Impaired Class that rejects, or is deemed to have rejected the Plan.

ARTICLE V

MEANS FOR IMPLEMENTATION

5.1 <u>Deemed Consolidation of ATCW Debtors for Plan Purposes Only.</u>Substantive Consolidation of ATCW Debtors for Plan Purposes Only. On the Initial Effective Date, the ATCW Debtors shall be deemed merged into ATCW, and (i) all assets and all liabilities of the ATCW Debtors shall be deemed merged into ATCW, (ii) all guaranties of any ATCW Debtor of the payment, performance, or collection of obligations of another ATCW Debtor shall be eliminated and canceled, (iii) any obligation of any ATCW Debtor and all guaranties thereof executed by one or more of the other ATCW Debtors shall be treated as a single obligation, and such guaranties shall be deemed a single Claim against the consolidated ATCW Debtors, (iv) all joint obligations of two or more ATCW Debtors, and all multiple Claims against such entities on account of such joint obligations shall be treated and allowed only as a single Claim against the consolidated ATCW Debtors, (v) all Claims between or among the ATCW Debtors shall be canceled, and (vi) each Claim filed in the Chapter 11 Cases of any ATCW Debtor shall be deemed filed against the consolidated ATCW Debtors and a single obligation of the consolidated ATCW Debtors on and after the Initial Effective Date,

Entry of the Confirmation Order will constitute the approval, pursuant to section 105(a) of the Bankruptcy Code, effective as of the Initial Effective Date, of the deemed substantive consolidation of the Chapter 11 Cases for purposes of voting on, confirmation of and distribution under the Plan. On and after the Initial Effective Date, (i) no distributions will be made under the Plan on account of Intercompany Claims among the ATCW Debtors, (ii) all guarantees of the ATCW Debtors of the obligations of any other ATCW Debtor will be eliminated so that any claim against any ATCW Debtor and any guarantee thereof executed by any other ATCW Debtor and any joint or several liability of any of the ATCW Debtors will be deemed to be one obligation of the consolidated ATCW Debtors and (iii) each and every Claim filed or to be filed in the Chapter 11 Cases of any of the ATCW Debtors will be deemed filed against the consolidated ATCW Debtors, and will be deemed one Claim against and obligation of the consolidated ATCW Debtors.

In accordance with the settlement of Claims and controversies under this<u>e</u> Plan and for purposes of voting and Distributions under the Plan only: (a) all assets and all liabilities of the ATCW Debtors will be treated as though the ATCW Debtors were merged; (b) any pre-Initial Effective Date obligation of any ATCW Debtor and all guarantees thereof executed by one or more of the ATCW Debtors will be deemed to be one obligation of the consolidated ATCW Debtors; (c) any Claims filed or to be filed in connection with any such obligation and such guarantees will be deemed one Claim against the consolidated ATCW Debtors; and (d) each and every Claim filed

in the individual Chapter 11 Case of any of the ATCW Debtors will be deemed filed against the consolidated ATCW Debtors in the consolidated case.

In addition, on the Initial Effective Date each Debtor that holds an Intercompany Claim against a direct subsidiary shall contribute such Intercompany Claim to the capital of such direct subsidiary. In the event the obligor under the Intercompany Claim is an indirect subsidiary, the holder shall contribute such Intercompany Claim to the capital of the direct subsidiary through which such holder indirectly controls such obligor. Such direct subsidiary (and, to the extent applicable, each of its direct or indirect subsidiaries to which the Intercompany Claim is then contributed) shall, in turn, be treated as the holder of such Intercompany Claim and shall contribute such Intercompany Claim to the capital of its direct subsidiaries in accordance with the immediately preceding two sentences. An Intercompany Claim shall continue to be contributed to the capital of subsidiaries until it is contributed to the capital of the subsidiary that is the obligor under such Claim, thereby eliminating such Intercompany Claim. On the Initial Effective Date, all other Intercompany Claims (including without limitation Intercompany Claims that a Debtor holds against a direct or indirect parent corporation) shall be eliminated and discharged. Notwithstanding the foregoing, at the election of XO and in its sole discretion, any Intercompany Claims between two ATCW Debtors may be eliminated other than through capital contributions as described above or may not be eliminated and discharged hereunder. In no event shall Distributions be made hereunder on account of any Intercompany Claims.

Notwithstanding the foregoing, the deemed consolidation <u>and substantive consolidation (each</u> <u>for Plan purposes only</u>) shall not (other than for purposes related to funding Distributions under the Plan) affect (w) the legal and organizational structure of the ATCW Debtors or the Reorganized Subsidiaries, (x) pre- and post-Commencement Date guaranties, liens and security interests that were required to be maintained (i) in connection with executory contracts or unexpired leases that were entered into during the Chapter 11 Cases or that have been or will be assumed by the ATCW Debtors or (ii) pursuant to the Plan or the Purchase Agreement, (y) distributions out of any insurance policies or proceeds of such policies, and (z) the tax treatment of the ATCW Debtors. *In addition, on the Initial Effective Date*, for all purposes, all other Intercompany Claims between and among the ATCW Debtors shall be eliminated and discharged. In no event shall Distributions be made hereunder on account of Intercompany Claims between and among the ATCW Debtors.

5.2 <u>Settlement of Certain Claims</u>. As discussed in detail in the Disclosure Statement, pursuant to section 1125 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration for the classification, $d\underline{D}$ istribution, releases, and other benefits provided under the Plan, upon the Initial Effective Date, the provisions of the Plan shall constitute a good faith compromise and settlement of all Claims and controversies resolved pursuant to the Plan. All Plan $d\underline{D}$ istributions made to Holders of Allowed Claims in any Class are intended to be and shall be final, and no Plan $d\underline{D}$ istribution to the Holder of a Claim in one Class shall be subject to being shared with or reallocated to the Holders of any Claim in another Class by virtue of any prepetition collateral trust agreement, shared collateral agreement, subordination agreement, or other similar inter-creditor arrangement.

5.3 Restructuring Transactions.

(a) **Transfer of Acquired Assets and Common Stock of IT Reorganized Subsidiaries**. On the Initial Effective Date, and in accordance with the terms of the Purchase Agreement and Sale Order, for good and valuable consideration, all of the Acquired Assets shall be sold, conveyed <u>(including, without limitation, assigned, transferred and delivered to either the Buyer or *the IT Reorganized Subsidiaries*, as applicable, and the common stock of the IT Reorganized Subsidiaries) shall be sold, conveyed, assigned, transferred and delivered to the Buyer, free and clear of all Liens, Claims, Equity Interests, and Interests (as defined in the Sale Order) of Entities; provided, however, to the extent that an Early Closing Election (as defined in the Purchase Agreement) has been delivered and the Sale Transaction has closed, then Section 5.3(<u>a) (other than the last sentence thereof), (c), (</u>d) and (e) of the Plan shall be inapplicable. Notwithstanding anything to the contrary contained herein, (i) prior to the Initial Effective Date, the parties to the Purchase Agreement retain their rights to make such an Early Closing Election (as provided in the Purchase Agreement) and to consummate the transactions contemplated under the Purchase Agreement if such Early Closing Election is made and (ii) the Buyer and the Reorganized Subsidiaries if the Early Closing Election is not made (x) shall not be liable for any</u>

Claims against or Equity Interests in the Debtors, other than the Assumed Liabilities, and (y) shall not be deemed to have assumed or be liable for any liabilities of the Reorganized Subsidiaries of the types and in the nature of the Excluded Liabilities (as defined in Section 2.4 of the Purchase Agreement). After the Initial Effective Date, each IT Reorganized Subsidiary shall have no further obligations under the Plan other than the Assumed Liabilities that relate to such IT Reorganized Subsidiary.

(b) <u>**Transfer of Excluded Assets**</u>. On the Initial Effective Date, all of the Excluded Assets (together with the other ATLT Assets but not including any Acquired Assets) shall be transferred to ATLT free and clear of all Liens, Claims and Equity Interests.

(c) <u>Non-Transferred Assets</u>. On each of the NTA Effective Dates, and in accordance with the terms of the Purchase Agreement and Sale Order, for good and valuable consideration, the common stock of the relevant NTA Reorganized Subsidiary (and any accompanying Non-Transferred Asset(s)) shall be conveyed, assigned, transferred and delivered to the Buyer, free and clear of all Liens, Claims, Equity Interests, and Interests (as defined in the Sale Order) of Entities. After each NTA Effective Date, the relevant NTA Reorganized Subsidiary shall have no further obligations under the Plan other than the Assumed Liabilities that relate to such NTA Reorganized Subsidiary.

(d) **Continued Corporate Existence and Remaining of Assets in the Reorganized Subsidiaries**. Each of the IT Reorganized Subsidiaries shall continue to exist after the Initial Effective Date, and each NTA Reorganized Subsidiary shall continue to exist after its NTA Effective Date, as applicable, as a separate legal Entity, with all the powers of a corporation, limited liability company, joint venture, or partnership, as applicable, under the laws of their respective states of incorporation, formation, or organization, and without prejudice to any right that Buyer has to alter or terminate such existence (whether by merger, acquisition, or otherwise) under such applicable state law. Except as otherwise provided in the Plan or the Purchase Agreement, on and after the Initial Effective Date and each of the NTA Effective Dates, as applicable, all property of the Subsidiaries, other than STFI, the STFI Assets and the Excluded Assets, shall remain in the Reorganized Subsidiaries, free and clear of all Claims, Liens, charges, Equity Interests, Interests (as defined in the Sale Order) or and other encumbrances. On and after the Initial Effective Date or each of the NTA Effective Dates, as applicable, the Reorganized Subsidiaries may operate their businesses and may use, acquire or dispose of their property, without the supervision or approval of the Bankruptcy Court and free of any restrictions of the Bankruptcy Code.

(e) <u>Reorganized Subsidiaries</u>. On the Initial Effective Date, the stock of all of the IT Reorganized Subsidiaries shall be transferred to Buyer, in such order and manner as Buyer may choose in its sole discretion in accordance with the Purchase Agreement, in exchange for consideration being given to the Debtors under the Purchase Agreement, which such consideration shall be distributed to Holders of Allowed Claims in accordance with the Plan. On the NTA Effective Date for each Subsidiary that holds or constitutes Non-Transferred Assets, the stock of such NTA Reorganized Subsidiary shall be transferred to Buyer in the manner chosen by Buyer in its sole discretion in accordance with the Purchase Agreement. All required corporate actions with respect to the Reorganized Subsidiaries will be determined by Buyer in its sole discretion.

(i) <u>New Certificates of Incorporation and New By-laws</u>. On or immediately prior to the Initial Effective Date or each of the NTA Effective Dates, as applicable, the Reorganized Subsidiaries will file their respective New Certificates of Incorporation with the applicable Secretaries of State in their respective states of incorporation. After the Initial Effective Date or each of the NTA Effective Dates, as applicable, the Reorganized Subsidiaries may amend and restate their respective New Certificate of Incorporation and other constituent documents as permitted by the laws of their respective states of incorporation.

(ii) <u>Directors and Officers of the Reorganized Subsidiaries</u>. Subject to section 1129(a)(5) of the Bankruptcy Code, the directors and officers of the Subsidiaries, other than STFI, shall resign as of the <u>Final Effective Date</u> <u>applicable effective date of each respective Subsidiary</u>. As of the Initial Effective Date or each of the NTA Effective Dates, as applicable, the initial board of directors of each of the Reorganized Subsidiaries shall be appointed by the Buyer in accordance with the respective Reorganized Subsidiaries' New Certificate of Incorporation. Pursuant to section 1129(a)(5), the Buyer, will disclose <u>by making a filing with</u> <u>the Bankruptcy Court</u>, on or prior to the Confirmation Date, the identity and affiliations of any Person proposed to serve on the initial board of directors of each of the Reorganized Subsidiaries. To the extent any such Person is an "insider" under the Bankruptcy Code, the nature of any compensation for such Person will also be disclosed. Each such director and officer shall serve from and after the Initial Effective Date or each of the NTA Effective Dates, as applicable, pursuant to the terms of the New Certificates of Incorporation and other constituent documents of the Reorganized Subsidiaries.

(iii) <u>Corporate Action</u>. As of the Initial Effective Date or each of the NTA Effective Dates, as applicable, the adoption and filing of the New Certificates of Incorporation, the approval of the New Bylaws, the appointment of directors and officers for each of the Reorganized Subsidiaries, and all actions contemplated hereby shall be deemed to be authorized and approved in all respects (subject to the provisions hereof). All matters provided for herein involving the corporate structure of the Reorganized Subsidiaries, and any corporate action required by the Reorganized Subsidiaries in connection with the Plan, shall be deemed to have occurred and shall be in effect, pursuant to applicable law, without any requirement of further action by the security holders or directors of the Reorganized Subsidiaries. On the Initial Effective Date or each of the NTA Effective Dates, as applicable, the appropriate officers of the Reorganized Subsidiaries and members of the board of directors of the Reorganized Subsidiaries are authorized and directed to issue, execute and deliver the agreements, documents, securities and instruments contemplated by the Plan in the name of and on behalf of the Reorganized Subsidiaries.

(f) <u>Reorganized STFI</u>. Reorganized STFI shall continue to exist after the Initial Effective Date as a separate legal Entity, with all powers of a corporation under the laws of the State of Delaware, and without prejudice to any right to alter or terminate such existence (whether by merger, acquisition, or otherwise) under the laws of the State of Delaware. Except as otherwise provided in the Plan or any related document, on and after the Initial Effective Date, the STFI Assets shall vest in Reorganized STFI free and clear of all Claims and Liens. On and after the Initial Effective Date, Reorganized STFI may operate the STFI Assets and may use, acquire or dispose of the STFI Assets without the supervision or approval of the Bankruptcy Court and free of any restrictions of the Bankruptcy Code or Bankruptcy Rules, other than those restrictions expressly imposed by the Plan or the Confirmation Order.

(i) <u>Issuance of New Securities</u>. On the Initial Effective Date, Reorganized STFI shall issue all securities, notes, instruments, certificates, and other documents of Reorganized STFI required to be issued pursuant to the Plan, including, without limitation, the New STFI Common Stock, which shall be distributed to ATLT for the benefit of Holders of ATLT <u>B-Certificates</u><u>"B" Trust Interests</u> in accordance with Sections 3.4(a) and 3.5(a) of the Plan.

(ii) <u>New Certificate of Incorporation and New By-laws.</u> On or immediately prior to the Initial Effective Date, STFI will file its New Certificate of Incorporation with the Secretary of State for the State of Delaware in accordance with the relevant sections of the corporate laws of the State of Delaware. After the Initial Effective Date, Reorganized STFI may amend and restate its New Certificate of Incorporation and other constituent documents as permitted by the laws of the State of Delaware. The New Certificate of Incorporation will prohibit the issuance of nonvoting equity securities.

(iii) Directors and Officers of the Debtors and Reorganized STFI. Subject to section 1129(a)(5) of the Bankruptcy Code, the directors and officers of Reorganized STFI shall resign as of the Initial Effective Date. As of the Initial Effective Date, the initial board of directors of Reorganized STFI shall be appointed by the Creditors Committee and in accordance with Reorganized STFI's New Certificate of Incorporation. Pursuant to section 1129(a)(5), the Creditors Committee will disclose, on or prior to the Confirmation Date, the identity and affiliations of any Person proposed to serve on the initial board of directors of Reorganized STFI. To the extent any such Person is an "insider" under the Bankruptcy Code, the nature of any compensation for such Person will also be disclosed. Each such director and officer shall serve from and after the Initial Effective Date pursuant to the terms of the New Certificate of Incorporation and other constituent documents of Reorganized STFI.

(iv) <u>Corporate Action</u>. As of the Initial Effective Date, the adoption and filing of the New Certificate of Incorporation, the approval of the New By-laws, the appointment of directors and officers for Reorganized STFI, and all actions contemplated hereby shall be deemed to be authorized and approved in all respects (subject to the provisions hereof). All matters provided for herein involving the corporate structure of Reorganized STFI, and any corporate action required by the Debtors or Reorganized STFI in connection with the Plan, shall be deemed to have occurred and shall be in effect, pursuant to applicable law, without any requirement of further action by the security holders or directors of the Debtors or Reorganized STFI. On the Initial Effective Date, the appropriate officers of Reorganized STFI and members of the board of directors of Reorganized STFI are authorized and directed to issue, execute and deliver the agreements, documents, securities and instruments contemplated by the Plan in the name of and on behalf of Reorganized STFI.

(v) <u>Working Capital.</u> On the Initial Effective Date, to provide Reorganized STFI with the working capital to operate the STFI Assets on a going-forward basis and, to the extent applicable, to fund any obligations under Section 4980B of the Internal Revenue Code to provide continuation of group medical coverage with respect to any employee or former employee employed by or in connection with any of the Debtors or other qualified beneficiary, the Debtors shall deposit Cash in an amount equal to the STFI Cash Amount into a bank account of Reorganized STFI or allow Reorganized STFI to retain Cash in its bank accounts equal to the STFI Cash Amount. Reorganized STFI shall not be funded with Cash provided under the Operating-Agreement to fund the Non-Transferred Assets set forth in the Purchase Agreement and such Cash shall at all times be held in separate bank accounts from Cash used to fund Reorganized STFI and ATLT.

5.4 <u>ATLT.</u>

(a) **Execution of the ATLT Agreement**. On the Initial Effective Date, the ATLT Agreement shall be executed, and all other necessary steps shall be taken to establish the ATLT and the beneficial interests therein which shall be for the benefit of the Holders of Allowed Claims in Classes 4 and 5 as provided in Sections 3.4 and 3.5 hereof. In the event of any conflict between the terms of this Section 5.4<u>5.5</u> and the terms of the ATLT Agreement, the terms of the ATLT Agreement shall govern.

(b) **<u>Purpose of ATLT</u>**. From and after the Initial Effective Date, ATLT shall, in accordance with the ATLT Agreement, (i) wind-down the Debtors' affairs, including making Distributions as contemplated hereinin the Plan, (ii) (a) liquidate, by conversion to Cash or other methods or (b) distribute directly to Holders of <u>ATLT Trust Interests</u>, the ATLT Assets, to the extent applicable, as expeditiously as reasonably possible in <u>order to maximize the recovery of Holders of ATLT Trust Interests in its sole discretion</u>, (iii) investigate, enforce and prosecute the Avoidance Actions and the Causes of Action, <u>provided, however, that any analysis regarding whether a Cause of Action should be commenced against any member of the Creditors Committee shall be conducted by the Debtors' attorneys or another party selected by the Debtors prior to the Confirmation Date, (iv) resolve<u>object to, settle, compromise, dispute and prosecute</u> Disputed Claims, (v) administer the Plan and take such actions as are necessary to effectuate the Plan, and (vi) file appropriate tax returns, each in the exercise of its fiduciary obligations and ATLT shall retain such professionals as are necessary and appropriate in furtherance of such fiduciary obligations.</u>

<u>In the event that there is not sufficient Cash to pay Holders of Allowed Senior Lender</u> <u>Claims in full on the Initial Effective Date, pursuant to the ATLT Agreement, ATLT shall sell such shares of</u> <u>XO Common Stock as are necessary to pay the Holders of Senior Lender Claims in full. Such sales of XO</u> <u>Common Stock shall be conducted in a commercially reasonable manner as determined by the Plan</u> <u>Administrator.</u>

<u>To maximize the value of the recovery for Holders of Allowed Claims in Classes 4 and 5 that</u> <u>receive ATLT Trust Interests pursuant to the Plan, during the life of the ATLT, the Plan Administrator, in</u> <u>consultation with the directors of ATLT, and in the exercise of their fiduciary duties to the Holders of such</u> <u>ATLT Trust Interests, will (a) sell the XO Common Stock and/or the New STFI Common Stock to third-</u> <u>parties and distribute the proceeds of such sale or sales to Holders of ATLT Trust Interests; (b) directly</u> <u>distribute the XO Common Stock and/or the New STFI Common Stock underlying the ATLT Trust Interests</u> <u>to the Holders of such ATLT Trust Interests; or (c) engage in combinations of distributions and sales of XO</u> <u>Common Stock and New STFI Common Stock, which in the business judgment of the directors of ATLT,</u> <u>would be most likely to maximize the value of the ATLT Trust Interests. The Plan Administrator and the</u> <u>directors of ATLT will owe fiduciary duty to the Holders of Allowed Claims that receive ATLT Trust</u> <u>Interests.</u>

(c) <u>ATLT Assets</u>. ATLT shall consist of the ATLT Assets, which, for the sake of clarity, shall include <u>(a)</u> the XO Common Stock, <u>(b)</u> the New STFI Common Stock, <u>all(c) the Litigation Cash, (d) the</u> Available Cash<u>(e) all Avoidance Actions</u> and <u>Causes of Action, whether or not such actions are commenced before</u> the <u>Excluded Initial Effective Date, and (f) all other assets of the Estates that are not the Acquired</u> Assets. <u>On</u> the Initial Effective Date, the Debtors shall transfer all of the ATLT Assets to ATLT free and clear of all Claims and Liens.

(d) <u>The ATLT Certificates</u>. On the Initial Effective-<u>Trust Interests</u>. *On the Initial Effective*_Date, ATLT shall issue [___] ATLT A Certificates, [___] ATLT B Certificates, and [___] ATLT C Certificates. The ATLT Certificates, in accordance with Sections 3.4 and 3.5 of the Plan, shall be distributed to the-Holders of Allowed (a) ATCW Unsecured Claims and (b) ATI Unsecured Claims <u>(i) that elect to receive ATLT Trust</u> <u>Interests or (ii) as a result of an oversubscription to the Cash Recovery Election will be required to</u> <u>receive ATLT Trust Interests, will receive notification from the Plan Administrator as to the number of</u> <u>ATLT Trust Interests to be allocated to such Holder</u>. The ATLT <u>Certificates shall represent</u><u>Trust Interests</u> <u>will not be certificated but, rather, will be reflected through book entries made by</u> the <u>value of ATLT Plan</u> <u>Administrator</u>.

(e) <u>Appointment of Plan Administrator</u>. The Creditors Committee shall appoint the Plan Administrator who shall be retained effective as of the Initial Effective Date. <u>The Creditors Committee shall disclose, by</u> <u>filing with the Bankruptcy Court, the name of the Plan Administrator at least ten (10) days prior to the</u> <u>Confirmation Hearing</u>. The Plan Administrator shall, in accordance with the ATLT Agreement, serve in such capacity through the earlier of the date ATLT is dissolved and the date such Plan Administrator resigns, is terminated or otherwise unable to serve; provided, however, that, in the event that the Plan Administrator resigns, is terminated or unable to serve, then the directors of ATLT shall have the right to select a successor who shall be appointed as the Plan Administrator and shall serve in such capacity until ATLT is dissolved or until such Plan Administrator resigns, is replaced or is terminated.

(f) **<u>Responsibilities of Plan Administrator</u>**. As set forth in more detail in the ATLT Agreement, the responsibilities of the Plan Administrator shall include, but shall not be limited to: (i) the wind-down of the Debtors' affairs, including making Distributions as contemplated herein, (ii) the liquidation, by conversion to Cash or other methods in a commercially reasonable manner, of the ATLT Assets such shares of XO Common Stock as necessary to pay Holders of Allowed Senior Lender Claims in full pursuant to Section 3.3 of the extent applicable Plan, (iii) maximize the value of the recovery for Holders of Allowed Claims in Classes 4 and 5 that receive ATLT Trust Interests pursuant to the Plan, by (a) selling the XO Common Stock and/or the New STFI Common Stock to third-parties and distribute the proceeds of such sale or sales to Holders of ATLT Trust Interests; (b) directly distributing the XO Common Stock and/or the New STFI Common Stock underlying the ATLT Trust Interests to the Holders of such ATLT Trust Interests: or (c) engaging in combinations of distributions and sales of XO Common Stock and New STFI Common Stock, each at the direction of the directors of ATLT, (iv) the investigation, enforcement and prosecution of the Avoidance Actions and the Causes of Action, provided, however, that any(v) conduct the analysis regarding whether a Cause of Action should be commenced against any member of the Creditors Committee shall be conducted by the Debtors' attorneys or another party selected by the Debtors prior Claims and object to the Confirmation Date, settle, compromise, dispute and prosecute Disputed Claims, including assertion of equitable subordination claims, as applicable, (iv vi) resolution of Disputed Claims, (v vii) administration of the Plan, and (viviii) filing of appropriate tax returns, each in the exercise of its fiduciary obligations and ATLT shall retain such professionals as are necessary and appropriate in furtherance of such fiduciary obligations. The Plan Administrator may invest (including any earnings thereon or proceeds therefrom) as permitted by section 345 of the Bankruptcy Code or as otherwise approved by the Bankruptcy Court.

(g) Bonding of Plan Administrator. The Creditors Committee is analyzing whether to obtain a bond to protect the Holders of Allowed Claims receive ATLT Trust Interests in the event the Plan Administrator defalcates the Cash and securities held by ATCW. The United States Trustee's Office believes the Plan Administrator should be bonded. In that regard, if the Creditors Committee determines not to obtain such a bond, the United States Trustee's Office may ask the Bankruptcy Court to direct that such a bond be obtained.

(h) (g) Directors of ATLT. On the Initial Effective Date, the directors of ATLT shall be appointed by the Creditors Committee. There shall be three directors of ATLT. The Creditors Committee will identify the directors of ATLT at least ten (10) days prior to the Confirmation Hearing.

(i) (h) Post-Initial Effective Date Professional Fees and Expenses. From and after the Initial Effective Date, the attorneys for the Debtors, the attorneys for the Creditors Committee and other Professionals retained in the Chapter 11 Cases may, from time to time, provide legal or other professional services in connection with the Plan as requested by the Plan Administrator. Such services may be paid without further application to the Bankruptcy Court within ten (10) days after submission of a bill to the Plan Administrator. If an objection is asserted by the Plan Administrator with respect to a bill and remains unresolved, the Plan Administrator may not pay the amounts requested in such bill; provided, however, that the affected Professional may file an application for allowance with the Bankruptcy Court on ten (10) days' notice to the Plan Administrator. After a hearing before the Bankruptcy Court, the fees and expenses will be paid in the amounts fixed by the Bankruptcy Court. Notwithstanding anything to the contrary contained herein, the Buyer and the Reorganized Subsidiaries shall not be liable for any of such fees and expenses.

(i) (i)-Post-Initial Effective Date Fees and Expenses. From and after the Initial Effective Date, ATLT shall, in the ordinary course of business and without the necessity of Bankruptcy Court approval, pay the reasonable fees and expenses <u>including the fees and charges assessed against the Debtors' Estates under section 1930 of title 28 of the United States Code</u>, related to the implementation and consummation of the Plan, other than the professional fees described in Section <u>5.45.5</u>(h) of the Plan. Any dispute with respect to such fees and expenses will be resolved by the Bankruptcy Court. Notwithstanding anything to the contrary contained herein, the Buyer and the Reorganized Subsidiaries shall not be liable for any of such fees and expenses.

(k) (j)-<u>Fiduciary Duties of Plan Administrator and Fiduciary Duties Directors of ATLT</u>. The Plan Administrator <u>and each of the directors of ATLT</u> shall act in a fiduciary capacity for the interests of all Holders of ATLT <u>Certificates Trust Interests</u>.

(h) (k) Dissolution of ATI and ATCW. On the date that ATI or ATCW no longer holds a Non-Transferred Asset, whether directly or indirectly, the Plan Administrator shall file on behalf of ATI and ATCW. as the case may be, a certification of dissolution with the Bankruptcy Court and, as a result thereof, such Entity will be dissolved without the necessity for any other or further actions to be taken by or on behalf of ATI and ATCW such Entity or payments to be made in connection therewith; provided, however, that ATI and ATCW shall file with the official public office for keeping corporate records in its state of incorporation or organization a certification of dissolution or equivalent document, provided, further, that ATLT shall be dissolved no later than three years from the Effective Date unless the Bankruptcy Court, upon motion within the six month period prior to the third anniversary (or the end of any extension period approved by the Bankruptcy Court), determines that a fixed period extension (not to exceed three years, together with any prior extensions, without a favorable letter ruling from the IRS that any further extension would not adversely affect the status of ATLT as a liquidating trust for federal income tax purposes) is necessary to facilitate or complete the recovery and liquidation of the ATLT Assets. After (a) the Distribution of all Cash, including the Litigation Cash, if any, Excluded Assets (or on-the proceeds thereof) and the ATLT Certificates Trust Interests pursuant to the Plan, (b) the filing by or on behalf of ATLT of a certification of dissolution with the Bankruptcy Court in accordance with this Section, and (c) the taking of all necessary actions to effectuate the closing of the Chapter 11 Cases, ATI and ATCW shall be deemed dissolved for all purposes without the necessity for any other or further actions; provided, however, that ATLT shall file with the

official public office for keeping corporate records in the state of incorporation or organization of each of the Debtors a certificate of dissolution or equivalent document.

(<u>m</u>) ()-<u>Closing of the Chapter 11 Cases</u>. ATLT shall take all necessary actions to effectuate the closing of the Chapter 11 Cases of the Debtors. Upon the Distribution of all ATLT Assets, the Plan Administrator shall take all necessary actions to effectuate the dissolution of ATLT.

(n) <u>Distributions. The Plan Administrator shall distribute at least annually and in accordance with</u> the ATLT Agreement, beginning on the Effective Date or as soon thereafter as is practicable, all Cash on hand and permitted investments, except such amounts (i) as would be distributable to holders of Disputed Claims if such Disputed Claims had been Allowed on the Effective Date (but only until such Claims are resolved), (ii) as are reasonably necessary to meet contingent liabilities and to maintain the value of the ATLT Assets during liquidation, (iii) to pay reasonable expenses (including, but not limited to, any taxes imposed on ATLT or in respect of the ATLT Assets), and (iv) to satisfy other liabilities incurred by ATLT in accordance with the Plan or the ATLT Agreement.

- (o) Federal Income Tax Treatment.
 - (ii) (m) <u>Tax TreatmentGeneral</u>. If possible, ATLT will be structured to be treated for federal income tax purposes as a "liquidating trust," as defined in Treasury Regulation Section 301.7701-4(d), according to the guidelines established by the <u>IRS Internal</u> <u>Revenue Service</u> in Rev. Proc. 94-45, 1994-2 C.B. 684, for the formation of liquidating trusts. If ATLT cannot be structured to comply with Rev. Proc. 94-45, then ATLT will be structured as another entity (or entities) intended not to be subject to federal income tax, <u>i.e.</u>, a "flow through" entity.
 - (ii)ATLT Assets Treated as Owned by Creditors. All parties (including, without
limitation, the Debtors, the Plan Administrator, and the Holders of Claims in
Classes 4 and 5) shall, for all federal income tax purposes, treat the transfer of the
ATLT Assets to ATLT as (a) a transfer of the ATLT Assets directly to the Holders
of Allowed Claims in Classes 4 and 5 followed by (b) such Holders' transfer of the
ATLT Assets to ATLT in exchange for Trust Interests in ATLT. Accordingly, the
Holders of such Claims shall be treated for federal income tax purposes as the
grantors and owners of their respective shares of the ATLT Assets.
 - (iii) <u>Tax Reporting.</u>
 - A.
 The Plan Administrator shall file the tax returns for ATLT that are required by law

 to be filed.
 The Plan Administrator shall also annually send to each Holder of an interest in ATLT a separate statement setting forth the Holder's share of items of income, gain, loss, deduction, or credit, and will instruct all such Holders to report such items on their federal income tax returns.

 Attribute
 Attribute

 Attrit
 Attribute
 - B.
 As soon as possible after the Effective Date, the Plan Administrator shall make a

 good faith valuation of the ATLT Assets, and such valuation shall be used

 consistently by all parties (including, without limitation, the Debtors, the Plan

 Administrator, and the Holders of Allowed Claims in Classes 4 and 5) for all federal

 income tax purposes. The Plan Administrator also shall file (or cause to be filed)

 any other statements, returns, or disclosures relating to ATLT that are required by

 any governmental unit.

- Subject to definitive guidance from the IRS or a court of competent jurisdiction to <u>C.</u> the contrary (including the receipt by the Plan Administrator of a private letter ruling if requested, or the receipt of an adverse determination by the IRS upon audit if not contested), the Plan Administrator shall (i) treat any assets allocable to, or retained on account of, Disputed Claims as held by one or more discrete trusts for federal income tax purposes (the "ATLT Disputed Claims Reserve"), consisting of separate and independent shares to be established in respect of each Disputed Claim, in accordance with the trust provisions of the Tax Code (sections 641 et. seq.). (ii) treat as taxable income or loss of the ATLT Dispute Claims Reserve, with respect to any given taxable year, the portion of the taxable income or loss of ATLT that would have been allocated to the Holders of Disputed Claims had such Claims been Allowed on the Effective Date (but only for the portion of the taxable year with respect to which such Claims are unresolved), (iii) treat as a distribution from ATLT for Disputed Claims any increased amounts distributed by ATLT as a result of any Disputed Claims resolved earlier in the taxable year, to the extent such distributions related to taxable income or loss of ATLT for Disputed Claims determined in accordance with the provisions hereof, and (iv) to the extent permitted by applicable law, report consistent with the foregoing for state and local income tax purposes. All Holders of Claims in Classes 4 and 5 shall report, for tax purposes, consistent with the foregoing.
- <u>D.</u> The Plan Administrator shall be responsible for payments, out of the ATLT Assets
 of any taxes imposed on ATLT or its assets, including the ATLT Disputed Claims
 <u>Reserve.</u> In the event, and to the extent, any Cash retained on account of Disputed
 <u>Claims in the ATLT Disputed Claims Reserve is insufficient to pay the portion of
 any such taxes attributable to the taxable income arising from the assets allocable
 to, or retained on account of, Disputed Claims, such taxes shall be (i) reimbursed
 <u>from any subsequent Cash amounts retained on account of Disputed Claims, or (ii)
 to the extent such Disputed Claims subsequently have been resolved, deducted from
 <u>any amounts distributable by the Plan Administrator as a result of the resolutions of
 such Disputed Claims.

 </u></u></u>
- E.
 The Plan Administrator may request an expedited determination of the taxes of ATLT, including the ATLT Disputed Claims Reserve, under section 505(b) of the Bankruptcy Code for all returns for, or on behalf of, ATLT for all taxable periods through the dissolution of ATLT.

5.5 **Cancellation of Existing Securities and Agreements**. On the Initial Effective Date or each of the NTA Effective Dates, as applicable, except to the extent otherwise provided herein, all notes, instruments, certificates, and other documents evidencing (a) the Senior Lender Claims, (b) the Senior Notes, (c) the Senior Discount Notes, (d) the Common Stock, (e) Equity Interests, and (f) any stock options, warrants or other rights to purchase Common Stock other than a Claim that is reinstated and rendered unimpaired under the Plan or Equity Interest held by a Debtor in any Subsidiary shall be canceled and the obligations of the Debtors or Reorganized STFI thereunder or in any way related thereto shall be discharged. On the Initial Effective Date, except to the extent otherwise provided herein, the ATI Note Indentures shall be deemed to be canceled, as permitted by section 1123(a)(5)(F) of the Bankruptcy Code, and the obligations of the Debtors of allowing the ATI Note Trustees, or their agent or servicer, to make the Distributions to be made on account of the Allowed ATI Unsecured Claims that relate to the ATI Notes in accordance with the Plan.

5.6 **Regulatory Approvals.** The transactions contemplated in this Plan, including, but not limited to, mergers, investments, transfers of assets, dissolutions, consolidations, and other transactions, shall be approved and effective as of the Initial Effective Date without the need for any further state or local regulatory approvals, and without any requirement or further action by the Debtors, ATLT, the Reorganized Subsidiaries or Reorganized STFI. All current customers will continue to receive service under the same rates, terms, and conditions as they

currently enjoy and the state or local regulatory commissions and agencies may not impose any requirements on the Debtors, the Reorganized Subsidiaries or Reorganized STFI to allow the current customers to terminate currently effective agreements with Debtors without liability. Impacted customers will receive notice of all of these organizational changes. Where required, tariffs of subsidiaries that will be created, merged or dissolved pursuant to this Plan will be appropriately incorporated into existing or new tariffs held by such Entities. State and local regulatory commissions will be able to review these tariff changes to assure that consumers' rights are fully protected. This Plan is not dependent on any rate changes.

5.7 **Insurance Preservation**. Nothing in this Plan, including any releases, shall diminish or impair the enforceability of any policies of insurance that may cover any claims against the Debtors or any other Person.

ARTICLE VI

EXECUTORY CONTRACTS AND UNEXPIRED LEASES

6.1 Assumption or Rejection of Executory Contracts and Unexpired Leases.

(a) Executory Contracts and Unexpired Leases. Schedules 1, 2, 3, and 4 will be filed with the Bankruptcy Court and served on all parties listed on such Schedules at least 20 days prior to the Confirmation Hearing in accordance with the requirements of the Bid Procedures Order, as modified by the Sale Order. Pursuant to sections 365(a) and 1123(b)(2) of the Bankruptcy Code, and subject to the terms and conditions of the Purchase Agreement, all executory contracts and unexpired leases between the Debtors and any Person or Entity that are not listed on Schedules 2, 3, 43 and 5 annexed hereto4 shall be deemed rejected by the Debtors on the Initial Effective Date. Contracts and leases listed on Schedule 1 shall be deemed rejected as of the Initial Effective Date. Contracts and leases listed on Schedule 2 shall be deemed rejected assumed sixty and (60) days after Buyer obtains the regulatory consents set forthwhere necessary in Section 2.5 of accordance with the Purchase Agreement relating and Sale Order) assigned to such contract or leaseBuver on the Initial Effective Date. Contracts and leases listed on Schedule 3 shall be deemed assumed and (where necessary in accordance with the Purchase Agreement and Sale Order) assigned to Buyer on the Initial Effective Date. Contracts listed on Schedule 4 shall be deemed assumed and (where necessary in accordance with the Purchase Agreement and Sale Order) assigned to Buyer on the date that Buyer obtains the regulatory consents set forth in Section 2.5 of the Purchase Agreement relating to such contract or lease. Contracts listed and leases listed on Schedule 4 shall be deemed rejected on Schedule the 5 shall be deemed rejected on the earliedater of (i) 20 days after notice of such rejection has been given by the Debtors to the counterparty to such contract, (ii)-180 days after the Initial Effective Date or and (iiii) another the date-later than the Initial Effective Date as reflected on such Schedule 5.4: provided, however, that the Buver must perform under such rejected contract or lease, as the case may be, until the applicable effective date of rejection; provided, further, that in the event that notice of a rejection of any contract or lease on Schedule 4 is given by the Debtors to the counterparty to such contract or lease, then such contract or lease shall be deemed rejected on the date that is twenty (20) days after the counterparty receives such notice. Schedules 2, 3-Pursuant to Section 6.3(e) of the Purchase Agreement, Schedule 4 and 5 annexed heretoto the Plan may be modified at any time by the Buyer prior to the Initial Effective Date in accordance with the terms and conditions of the Purchase Agreement, however, such modification may only be made with respect to the applicable effective date of the rejection of the contracts and leases listed on Schedule 4 as of the twentieth (20th) day prior to the Confirmation Hearing. The effective dates of the assumptions and/or rejections of various contracts and leases, as set forth above, may be the subject of disputes raised by certain of the Debtors' counter-parties to those contracts and leases. In accordance with terms of the Purchase Agreement and the Sale Order, the Debtors and the Buver shall cooperate with each other in the preparation of such Schedules 1, 2, 3 and 4.

(b) <u>Claims Based on Rejection of Executory Contracts or Unexpired Leases</u>. All proofs of Claim with respect to Claims arising from the rejection of executory contracts or unexpired leases, if any, must be filed with the Bankruptcy Court within thirty (30) days after the date of <u>notice of the</u> entry of an order of the Bankruptcy Court (including the Confirmation Order) <u>or other order</u> approving such rejection. Any Claims arising from the rejection of an executory contract or unexpired lease not filed within such time will be forever barred from

assertion against the Debtors, their Estates and property, or ATLT, unless otherwis e ordered by the Bankruptcy Court or provided herein. <u>Solely for purposes of voting on the Plan, any contract that is not a Tariff shall</u> <u>be deemed an executory contract.</u>

(c) Cure of Defaults for Executory Contracts and Unexpired Leases Assumed Any monetary amounts by which each executory contract and unexpired lease to be assumed pursuant to the Plan is in default shall be satisfied, pursuant to section 365(b)(1) of the Bankruptcy Code, by payment of the default amount in Cash on the later of (i) the Initial Effective Date, (ii) the date the Bankruptcy Court determined by Final Order the default amount, or (iii) on such other terms as the parties to such executory contracts or unexpired leases and the Buver may otherwise agree. In the event of a dispute regarding: (a) the amount of any cure payments, (b) the ability of Reorganized STFI or any assignee, including Buyer, to provide "adequate assurance of future performance" (within the meaning of section 365 of the Bankruptcy Code) under the contract or lease to be assumed, or (c) any other matter pertaining to assumption, the cure payments required by section 365(b)(1) of the Bankruptcy Code shall be made following the entry of a Final Order resolving the dispute and approving the assumption. The Bidding Procedures Order (as defined in and modified by the Sale Order) provides, and the Confirmation Order shall provide, as applicable, for notices of proposed assumption and proposed cure amounts to be sent to applicable third parties and for procedures for objecting thereto and resolution of disputes by the Bankruptcy Court. Disputes with respect to cure between the Debtors and the Buyer shall be governed by the terms and conditions of Section 3.5 of the Purchase Agreement. The Buyer shall have standing in the Chapter 11 Cases with regard to cure amounts and all other matters relating to the assumption or assignment of executory contracts or unexpired leases, including, without limitation, adequate assurance matters.

(d) <u>Insurance Policies</u>. All of the Debtors' insurance policies and any agreements, documents or instruments relating thereto, are treated as executory contracts under the Plan. On the Initial Effective Date, the Debtors shall be deemed to have assumed all insurance policies and any agreements, documents and instruments relating to coverage of all Insured Claims and shall be deemed to have assigned all of the Debtors' rights thereunder to ATLT. Notwithstanding the foregoing, Distributions under the Plan to any Holder of an Insured Claim shall be in accordance with the treatment provided under Article III of the Plan; <u>provided</u>, <u>however</u>, that any Distribution with respect to any and all Insured Claims shall be limited to the amount of proceeds payable under the insurance policies and any agreements, documents and instruments relating thereto.

Utility Services. Utility Companies shall not be entitled to request any additional deposits, 6.2 payments or other financial security from the Debtors, Reorganized STFI, the Reorganized Subsidiaries or Buyer as a result of, arising out of, or in connection with, the Chapter 11 Cases. On and after the Initial Effective Date, all Utility Companies that provided Utility Services pursuant to Tariff prior to the Initial Effective Date shall continue to provide such services to the Debtors. Reorganized STFI, the Reorganized Subsidiaries or Buver after the Initial Effective Date without interruption in the same manner as they did prior to the Initial Effective Date as the Tariffs for such Utility Services may be amended from time to time in accordance with applicable law. On and after the Initial Effective Date, all Utility Companies that provided Utility Services pursuant to an unexpired contract, which has been assumed or assumed and assigned by the Debtors to the Buyer, shall continue to provide to the Debtors, Reorganized STFI, the Reorganized Subsidiaries or Buyer without interruption all Utility Services provided pursuant to such contract in accordance with the terms of such contract and applicable law; provided, however, that to the extent such contracts are deemed to be executory contracts, the Debtors have cured all defaults and otherwise satisfied the requirements for the assumption or assumption and assignment of such contracts under section 365 of the Bankruptcy Code. To the extent that a contract for Utility Service has expired, has been rejected, or has been terminated by Debtors prior to the Initial Effective Date, no action will be taken by Debtors, Reorganized STFI, the Reorganized Subsidiaries or Buver with respect to such expired, rejected, or terminated Utility Services contract. Any Claim against a Debtor by a Utility Company (or a Holder of a Claim of a Utility Company) for the provision of Utility Services prior to the Commencement Date to such Debtor pursuant to Tariff or expired, terminated, or rejected contract shall be deemed to be an ATCW Unsecured Claim and shall be treated in accordance with Section 3.4 of the Plan. The Buyer shall have standing with respect to Claims arising out of Utility Services.

6.3 Tariff Services. After the Initial Effective Date, all Utility<u>Access Companies Providers</u> shall continue to provide to the Debtors, Reorganized STFI, the <u>Buyer or the</u> Reorganized Subsidiaries-or-Buyer, as the <u>case may be</u>, without interruption all <u>UtilityTariff</u> Services. <u>specifically including usage-sensitive access services</u>, provided to the Debtors prior to the Initial Effective Date whether such Utility Services were provided pursuant to a contract or Tariff. Utility<u>Access</u> Companies<u>Providers</u> shall not be entitled to request any additional deposits or other financial security from the Debtors, Reorganized STFI, the Reorganized Subsidiaries or Buyer as a result of, arising out of, or in connection with, the Chapter 11 Cases. Any Claim against a Debtor by <u>a Utility Companyan</u> <u>Access Provider</u> (or a Holder of a Claim of <u>a Utility Companyan Access Provider</u>) for the provision of Utility<u>Tariff</u> Services to such Debtor prior to the Commencement Date shall be deemed to be an ATCW Unsecured Claim and shall be treated in accordance with Section 3.4 hereof<u>of the Plan</u>. The Buyer shall have standing with respect to Claims arising out of Utility<u>Tariff</u> Services.

6.3-Risks of Litigation Relating to Utility Services and Tariff Services - After the Initial <u>6.4</u> Effective Date, all Access Providers shall continue. The principal providers of telecommunications services to the Debtors have asserted that the Plan violates the Bankruptcy Code, is contrary to provide to the Debtors customary practice in other telecommunications bankruptcy cases, Reorganized STFI, the Buyer or the Reorganized Subsidiaries, as the case may be, without interruption all Tariff Services, specifically including usagesensitive access services, provided to the Debtors prior to the Initial Effective Date. Access Providers shall not be entitled to request any additional deposits or other financial security from the Debtors, Reorganized STFI, the Reorganized Subsidiaries or Buyer as a result of, arising out of, or in connection with, the Chapter 11 Cases. Any Claim against a Debtor by an Access Provider (or a Holder of a Claim of an Access Provider) for the provision of Tariff Services to such Debtor prior to the Commencement Date shall be deemed to be an ATCW Unsecured Claim and shall be treated in accordance with Section 3.4 hereof. The Buyer shall have standing with respect to Claims arising out of Tariff Services cannot be confirmed because it purports to require them, as well as other Utility Companies and Access Providers, to continue providing services to the Debtors, Reorganized STFL the Reorganized Subsidiaries and the Buver, without the assumption (and/or assumption and assignment) and cure of substantial alleged monetary defaults under various tariffs and agreements. These Utility Companies and Access Providers contend, among other things, that they provide Utility Services and Tariff Services pursuant to executory contracts within the meaning of the Bankruptcy Code and that, therefore, the Debtors, Reorganized STFI, the Reorganized Subsidiaries and/or the Buyer may not continue to demand the provision of such services post-confirmation without assuming the executory contracts and paying all amounts owed to these providers and otherwise curing all defaults relating thereto. The Debtors and the Credi tors Committee dispute these contentions and believe that notwithstanding these arguments, the Plan is confirmable because it complies with the requirements of the Bankruptcy Code. The Debtors acknowledge, however, that the services generally provided by these telecommunications providers are critical to the Debtors' ongoing operations and to those of the Reorganized Subsidiaries.

6.5 Tariffs Services Determined to be Executory Contracts. To the extent the Court determines a Utility Service provided pursuant to a Tariff is provided pursuant to an executory contract within the meaning of section 365 of the Bankruptcy Code, such Utility Services *shall be treated in accordance with* the provisions of section 6.1 of this Plan and the Debtors shall provide the Utility Company with notice of their intent to reject, assume or assume and assign such Tariffed Utility Services within fifteen (15) days after the determination by the Court.

ARTICLE VII

PROVISIONS GOVERNING DISTRIBUTIONS

7.1 **Distributions for Claims and Equity Interests Allowed as of the Initial Effective Date**. Except as otherwise provided herein or as may be ordered by the Bankruptcy Court, Distributions to be made on account of Claims that are Allowed as of the Initial Effective Date shall be made on the Initial Effective Date, or as soon as practicable thereafter. Unless otherwise specifically provided for or contemplated in the Plan or Confirmation Order, or required by applicable bankruptcy law, post-petition interest shall not accrue or be paid on any Claims and no Holder of a Claim shall be entitled to interest accruing on or after the Commencement Date. For tax purposes,

Distributions received in respect of Allowed Claims shall be allocated first to the principal amount of the Allowed Claims with any excess allocated to unpaid interest that accrued on such Claims.

7.2 **Delivery of Distributions by ATLT**. ATLT shall make all Distributions required to be distributed under the Plan, except that the ATI Note Trustees shall deliver the Distributions, if any, to the Holders of Allowed ATI Note Claims in accordance with the ATI <u>Note</u> Indentures and this Plan. Any Distribution required to be made pursuant to this Plan on a day other than a Business Day shall be made on the next succeeding Business Day. ATLT may employ or contract with other Entities to assist in or make the Distributions required by the Plan without further order of the Court.

7.3 **Delivery and Distributions and Undeliverable or Unclaimed Distributions**.

(a) **Delivery of Distributions in General**. Distributions to Holders of Allowed Claims shall be made at the address of the Holder of such Claim as indicated on the records of the Debtors or, if such Holder holds such Claims through a Nominee, Distributions with respect to such Claims will be made to such Nominee and such Nominee shall in turn, make appropriate book entries to reflect such Distributions to such Holders.

(b) <u>Undeliverable Distributions</u>.

- (i) Holding and Investment of Undeliverable Distributions. If a Distribution of Cash is returned to ATLT as undeliverable or is otherwise unclaimed for one (1) year after the Initial Effective Date, such Cash shall be distributed on a pro rata basis to Holders of Allowed Unsecured Claims in accordance with the Sections 3.4(a) and 3.5(a) of the Plan. Undeliverable Distributions shall remain in the possession of ATLT until the earlier of such time as a Distribution becomes deliverable or until the first anniversary of the Initial Effective Date. Undeliverable Cash shall not be entitled to any interest, dividends or other accruals of any kind. As soon as reasonably practicable, ATLT shall make all Distributions that become deliverable.
- (ii) Failure to Claim Undeliverable Distributions. Any Holder of an Allowed Claim that does not assert a Claim pursuant to the Plan for an undeliverable or unclaimed Distribution within one (1) year after the Initial Effective Date shall be deemed to have forfeited its Claim for such undeliverable or unclaimed Distribution and shall be forever barred from asserting any such Claim against any of the Debtors or their Estates, Reorganized STFI or its property. Nothing contained herein shall require ATLT, or any other party, to attempt to locate any Holder of an Allowed Claim.

7.4 <u>Compliance with Tax Requirements/Allocations</u>. In connection with the Plan, to the extent applicable, ATLT shall comply with all tax withholding and reporting requirements imposed on it by any governmental unit, and all Distributions pursuant hereto shall be subject to such withholding and reporting requirements. In addition, ATLT will pay taxes on the taxable net income or gain allocable to Holders of Disputed Claims on behalf of such Holders and, when such Disputed Claims are ultimately resolved, Holders whose Disputed Claims are determined to be Allowed Claims will receive Distributions from ATLT net of the taxes that ATLT had paid previously on their behalf.

7.5 **Record Date for Distribution**. At the close of business on the Initial Effective Date, the transfer register for the ATI Notes shall be closed and there shall be no further changes in the record Holders of any ATI Notes. Moreover, ATLT shall have no obligation to recognize the transfer of any ATI Notes occurring after the Initial Effective Date, and shall be entitled for all purposes herein to recognize and deal only with those Holders of record as of the close of business on the Initial Effective Date.

7.6 **Fractional ATLT Certificates and De Minimis Distributions**. ATLT Certificates shall be issued in whole numbers only. There shall be no fractional shares of ATLT Certificates. No Cash payment of less than \$250.00500.00 shall be required to be made to the Holder of any Allowed Claim until the final date on which Cash is distributed to Holders of Allowed Claims pursuant to this Plan.

7.7 Set-offs and Recoupments. The Debtors or ATLT may, but shall not be required to, set-off against or recoup from any Allowed Claim on which payments are to be made pursuant to the Plan, any claims of any nature whatsoever (except for those claims and rights (including, without limitation, set off rights) constituting Acquired Assets), the Debtors or ATLT may have against the Holders of such Claim that is not released under Article X of the Plan and the Distributions to be made pursuant hereto on account of such Claim, but neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Debtors of any such Claim the Debtors may have against the Holder of such Claim. The deemed substantive consolidation of the ATCW Debtors shall not adversely affect the right of set-off of any Holder of an Allowed Claim. Pursuant to Section 3.5(a) of the Purchase Agreement, all pre-Commencement Date accounts receivable of the Debtors owed by ILECs shall be set off against the ILEC Cure Amounts (as defined in the Purchase Agreement) and thereby used as currency to pay the ILEC Cure Amounts.

7.8 **Surrender of Canceled Instruments or Securities** As a condition precedent to receiving any Distribution pursuant to the Plan on account of an Allowed Claim, the Holder of such Claim shall tender the applicable instruments, securities or other documentation evidencing such Claim to ATLT, and in the case of the Credit Agreement, the agent under the Credit Agreement shall have either (a) delivered such release documents as requested by the Debtors (at the Debtors expense) or (b) provided a letter of authorization to the Debtors to execute and release any instruments, securities or other documentation securing the Senior Lenders Claim. Any Distribution to be distributed pursuant to the Plan on account of any such Claim shall, pending such surrender, be treated as an undeliverable Distribution pursuant to Section 7.3 hereof. Any Holder of such Claim that fails to (i) surrender such instrument, security, note, or other documentation evidencing such Claim or (ii) execute and deliver an affidavit of loss and/or indemnity reasonably satisfactory to the Plan Administrator, before the first (1st) anniversary of the Initial Effective Date shall be deemed to have forfeited all rights and Claims and may not participate in any distribution under the Plan. Any distribution so forfeited shall become property of ATLT.

7.9 **ATI Notes.** The Distributions to Holders of Allowed ATI Note Claims shall be made by the ATI Note Trustees as agents and shall be made in accordance with the ATI Note Indentures and this Plan.

ARTICLE VIII

PROCEDURES FOR TREATING DISPUTED CLAIMS

8.1 **Objections to Claims**. After the Initial Effective Date and on or before the Claims Objection Deadline, ATLT shall have the authority to file objections to Claims or Equity Interests. ATLT may settle, compromise, withdraw or litigate to judgment objections to Claims without further Bankruptcy Court approval.

8.2 **No Distributions Pending Allowance**. Notwithstanding any other provision hereof, if any portion of a Claim is a Disputed Claim, no payment or Distribution provided hereunder shall be made on account of such Claim unless and until such Disputed Claim becomes an Allowed Claim, in whole or in part; <u>provided</u>, <u>however</u>, that the resolution of a Disputed Claim shall not preclude ATLT from commencing an Avoidance Action against the Holder of such Disputed Claim. No interest shall accrue on such Disputed Claim until the date that such Disputed Claim becomes an Allowed Claim.

8.3 **Personal Injury Claims**. Notwithstanding anything to the contrary contained herein, all Personal Injury Claims are Disputed Claims. No Distributions shall be made on account of any Personal Injury Claim unless and until such Claim is liquidated and becomes an Allowed Claim. Any Personal Injury Claim which has not been liquidated prior to the Initial Effective Date and as to which a proof of Claim was timely filed in the Chapter 11 Cases, shall be determined and liquidated either (i) in the Bankruptcy Court or **United States District Court for the Southern District of New York, as appropriate, or** (ii) in the administrative or judicial tribunal in which it is pending on the Initial Effective Date or, if no action was pending on the Initial Effective Date, in any administrative or judicial tribunal of appropriate jurisdiction. Any Personal Injury Claim determined and liquidated (a) pursuant to a judgment obtained in accordance with this Section 8.3 of the Plan and applicable nonbankruptcy law which is no longer applicable or subject to review, or (b) in any alternative dispute resolution or similar proceeding as same may be approved by order of a court of competent jurisdiction, shall be deemed, to the extent applicable, an Allowed Claim in Class 4, in such liquidated amount and treated in accordance with Section 3.4(a) of the Plan. Nothing

contained in this-Section 8.3<u>of the Plan</u> shall constitute or be deemed a waiver of any Claim, right, or Cause of Action that the Debtors may have against any person in connection with or arising out of any Personal Injury Claim, including, without limitation, any rights under section 157(b) of title 28 of the United States Code.

Estimation of Claims. The Debtors or the Creditors Committee (prior to the Initial Effective 8.4 Date) or ATLT (after the Initial Effective Date) may, at any time, and from time to time, request that the Bankruptcy Court estimate any Disputed Claim pursuant to section 502(c) of the Bankruptcy Code regardless of whether an objection was previously filed with respect to such Claim, or whether the Bankruptcy Court has ruled on any such objection, and the Bankruptcy Court will retain jurisdiction to estimate any Claim at any time during litigation concerning any objection to any Claim, including during the pendency of any appeal relating to any such objection. In the event that the Bankruptcy Court estimates any Disputed Claim, that estimated amount will constitute either the Allowed amount of such Claim or a maximum limitation on such Claim against any party or Entity, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on such Claim, the Debtors, the Creditors Committee, or ATLT, with the consent of the Creditors Committee or Debtors, as applicable, which shall not be unreasonably withheld, may elect to pursue any supplemental proceedings to object to any ultimate payment on such Claim. All of the objection, estimation, settlement, and resolution procedures set forth in the Plan are cumulative and not necessarily exclusive of one another. Claims may be estimated and subsequently compromised, settled, withdrawn, or resolved by any mechanism approved by the Bankruptcy Court. To the extent that any Disputed Claim constitutes an Assumed Liability under the Purchase Agreement, the Debtors, the Creditors Committee or ATLT, as applicable, will consult with the Buyer in connection with any objection to or any estimation of such Disputed Claim.

8.5 **Distributions Relating to Allowed Insured Claims**. Distributions under the Plan to each Holder of an Allowed Insured Claim shall be in accordance with the treatment provided under Section 3.4(a) of the Plan for Class 4; <u>provided, however</u>, that in no event shall the Allowed amount of an Insured Claim exceed the maximum amount that the Debtors are required to pay in respect of such Insured Claim pursuant to any pertinent insurance policies and applicable law.<u>4</u>. Nothing contained herein shall constitute or be deemed a waiver of any Cause of Action that the Debtors or any Entity may hold against any other Entity, including, without limitation, insurers under any policies of insurance.

Disputed Claims Reserve. On the Initial Effective Date (or as soon thereafter as is practicable), 8.6 ATLT shall establish the Disputed Claims Reserve, which may be established through one or more accounts, and shall reserve in respect of each Disputed Claim, Cash, ATLT Certificates Trust Interests, as applicable, that would have been distributed to the Holder of such Disputed Claim if such Disputed Claim had been an Allowed Claim on the Initial Effective Date in an amount equal to the least of (i) the amount of the Claim filed with the Bankruptcy Court, or, if no amount was specified, an amount determined by the Debtors and the Creditors Committee, (ii) if no Claim was filed, the amount listed by the Debtors in the Schedules as not disputed, contingent or unliquidated, or (iii) the amount, if any, estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code. Any Cash reserved by ATLT on account of Disputed Claims shall be set aside, segregated and held in interest-bearing accounts or certificates of deposit. Notwithstanding anything to the contrary contained herein, the amount of Cash (including interest actually earned thereon) and the ATLT Certificates Trust Interests reserved in respect of any Disputed Claim shall constitute the maximum amount of Cash and ATLT Certificates Trust Interests to be distributed to the Holder of such Disputed Claim. The Plan Administrator shall be responsible for payments. out of the ATLT Assets of any taxes imposed on ATLT or its assets, including the ATLT Disputed Claims Reserve. In the event, and to the extent, any Cash retained on account of Disputed Claims in the ATLT Disputed Claims Reserve is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of Disputed Claims, or (ii) to the extent such Disputed Claims subsequently have been resolved, deducted from any amounts distributable by the Plan Administrator as a result of the resolutions of such Disputed Claims.

8.7 **Distributions After Allowance**. ATLT shall distribute from the Disputed Claims Reserve to the Holder of any Disputed Claim that has become an Allowed Claim, no later than the fifth business day after the end of the calendar month in which such Disputed Claim becomes an Allowed Claim, Cash plus any interest actually earned on such Cash or ATLT Certificates <u>Trust Interests</u>, as applicable in amounts equal to the Cash or ATLT
Certificates <u>Trust Interests</u> that such Holder would have received on account of such Claim if such Claim had been an Allowed Claim on the Initial Effective Date.

8.8 **Distributions After Disallowance**. If a Disputed Claim is disallowed, in whole or in part, ATLT shall, on a quarterly basis if aggregate distribution in the first quarter are over \$5 million (otherwise, the Plan Administrator may determine in its discretion), redistribute to the Holders of Allowed Unsecured Claims in accordance with the terms of this Plan each such Holder's pro rata share of the Cash (including interest actually earned thereon) and ATLT Certificates Trust Interests reserved in respect of such disallowed Disputed Claim.

8.9 <u>Controversy Concerning Impairment</u>. If a controversy arises as to whether any Claims are Impaired under the Plan, the Bankruptcy Court shall, after notice and a hearing, determine such controversy before the Confirmation Date.

ARTICLE IX

EFFECTIVENESS OF THE PLAN

9.1 <u>Conditions Precedent to Initial Effective Date</u>. The Plan shall become effective with respect to (i) ATI, (ii) ATCW, and (iii) any Subsidiary that does not hold or constitute a Non-Transferred Asset, after the following conditions have been satisfied or waived pursuant to Section 9.4 of the Plan:

(a) The Confirmation Order, in form and substance reasonably acceptable to the Debtors, the Senior Lenders, the Creditors Committee and, if applicable, the Buyer, shall have been entered by the Bankruptcy Court and shall have become a Final Order. Without limiting the generality of the foregoing, the Confirmation Order shall be in accordance with the terms and conditions of the Purchase Agreement, including but not limited to Section 6.3(a) and 6.3(b) of the Purchase Agreement.

(b) All actions, documents, instruments, and agreements, including the Certificates of Incorporation and By-laws of Reorganized STFI<u>and each of *the IT Reorganized Subsidiaries*</u>, each of the other Plan Documents and the Transaction Documents (as defined in the Purchase Agreement) necessary to implement and effectuate the Plan and the Purchase Agreement shall have been taken or executed and delivered, as the case may be.

(c) The Debtors shall have received all authorizations, consents, regulatory approvals, rulings, letters, noaction letters, opinions or documents that are necessary to implement and effectuate the Plan and the Purchase Agreement.

(d) If applicable, each of the conditions to Closing under the Purchase Agreement shall have been satisfied or waived in accordance with the provisions thereof.

(e) The Closing (including the Early Funding Date) under the Purchase Agreement shall occur prior to or simultaneously with the effectiveness of the Plan.

9.2 <u>Conditions Precedent to NTA Effective Date</u>. With respect to each Subsidiary that holds or constitutes a Non-Transferred Asset, the Plan shall become effective after the following conditions have been satisfied or waived pursuant to Section 9.4 of the Plan:

(a) The Initial Effective Date shall have occurred.

(b) All actions, documents, instruments, and agreements, including the Certificates of Incorporation and By-laws of the Reorganized Subsidiaries that hold or constitute Non-Transferred Assets, each of the other Plan Documents and the Transaction Documents (as defined in the Purchase Agreement) necessary to implement and effectuate the Plan and the Purchase Agreement with respect to the Non-Transferred Assets shall have been taken or executed and delivered, as the case may be.

(c) The Debtors shall have received all authorizations, consents, regulatory approvals, rulings, letters, noaction letters, opinions or documents that are necessary to implement and effectuate the Plan and the Purchase Agreement with respect to the Non-Transferred Assets <u>relating to such NTA Reorganized Subsidiary</u>.

9.3 Effect of Failure of Conditions. In the event that one or more of the conditions specified in Section 9.1 of the Plan have not occurred or been waived on or before the date that is one (1) year after the Confirmation Date but subject to the terms and conditions of the Purchase Agreement (including, without limitation, the Early Closing Election rights of the Buyer thereunder): (a) the Confirmation Order shall be vacated; (b) no Distributions under the Plan shall be made; (c) the Debtors (with respect to their rights in connection with Claims and Equity Interests) and all Holders of Claims and Equity Interests shall be restored to the status quo ante as of the day immediately preceding the Confirmation Date as though the Confirmation Date never occurred; provided, however, that such restoration shall not restore liensaffect in any way whatsoever the Sale Transaction, shall not affect the assumption or assignment of any Assumed Contracts or the cure amounts associated with such assumption or assignment; and (d) the Debtors' obligations with respect to Claims and Equity Interests shall remain unchanged and nothing contained in the Plan shall constitute or be deemed a waiver or release of any Claims or Equity Interests by or against the Debtors or any other Person or Entity or to prejudice in any manner the rights of the Debtors or any Person or Entity in any further proceedings involving the Debtors.

9.4 <u>Waiver of Conditions</u>. The Debtors, with the written consent of the Creditors Committee, if applicable, which consent shall not be unreasonably withheld, delayed or conditioned, and the Buyer may waive one or more of the conditions precedent to effectiveness set forth in Section 9.1 of the Plan.

ARTICLE X

EFFECT OF CONFIRMATION OF PLAN

10.1 **Binding Effect** The Plan shall be binding upon and inure to the benefit of the Debtors, the Reorganized Subsidiaries, Reorganized STFI, the Buyer, ATLT, all Holders of Allowed Claims and their prospective successor and assignees, including, but not limited to, all parties in interest in the Chapter 11 Cases, including Professionals.

10.2 <u>Term of Injunction or Stays</u>. Unless otherwise provided, all injunctions or stays provided for in the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Initial Effective Date.

10.3 **Preservation of Avoidance Actions.** On and after the Initial Effective Date, any and all Avoidance Actions shall be preserved and retained by ATLT, which shall have the exclusive right to enforce, settle and prosecute any such Avoidance Actions subject to the terms and conditions of the Purchase Agreement for the benefit of the Holders of Allowed Unsecured Claims. ATLT may pursue, abandon, settle or release any or all Avoidance Actions, as it deems appropriate, without the need to obtain approval or any other or further relief from the Bankruptcy Court. ATLT may offset any claim supporting an Avoidance Action against any payment due to an Entity under the Plan.

10.4 **<u>Retention of Causes of Action/Reservation of Rights.</u>**

(a) Nothing contained in the Plan or the Confirmation Order shall be deemed to be a waiver or the relinquishment of any rights or Causes of Action that the Debtors, ATLT, or Reorganized STFI may have or which ATLT may choose to assert on behalf of the Debtors' Estates under any provision of the Bankruptcy Code or any applicable non-bankruptcy law, including, without limitation, (i) any and all Claims against any Person or Entity, to the extent such Person or Entity asserts a crossclaim, counterclaim, and/or Claim for set-off which seeks affirmative relief against the Debtors, ATLT, the Buyer, Reorganized Subsidiaries, Reorganized STFI, their officers, directors, or representatives, or (ii) the turnover of any property of the Debtors' Estates. To the extent this provision conflicts with the Purchase Agreement, the terms and conditions of the Purchase Agreement shall control. Without limiting the generality of the immediately preceding sentence, ATLT may

not assert any Cause of Action, Claim, right of set off or other legal or equitable defense that constitutes an Acquired Asset.

(b) Nothing contained in the Plan or the Confirmation Order shall be deemed to be a waiver or relinquishment of any Claim, Cause of Action, right of set-off, or other legal or equitable defense which the Debtors had immediately prior to the Commencement Date, against or with respect to any Claim left Unimpaired by the Plan. ATLT shall have, retain, reserve, and be entitled to assert all such Claims, Causes of Action, rights of set-off, and other legal or equitable defenses which the Debtors had immediately prior to the Commencement Date fully as if the Chapter 11 Cases had not been commenced, and all of the Debtors' legal and equitable rights respecting any Claim left Unimpaired by the Plan may be asserted after the Confirmation Date to the same extent as if the Chapter 11 Cases had not been commenced. To the extent this provision conflicts with the Purchase Agreement, the terms and conditions of the Purchase Agreement shall control. Without limiting the generality of the immediately preceding sentence, ATLT may not assert any Cause of Action, Claim, right of set off or other legal or equitable defense that constitutes an Acquired Asset.

(c) <u>The Debtors are in the process of analyzing potential Causes of Action and, as of the date hereof,</u> <u>the Debtors have not completed this analysis.</u> <u>Accordingly, the Debtors cannot ascribe a value to the</u> <u>Causes of Action and it is possible that the Estates do not hold any valuable or valid Causes of Action.</u>

10.5 Releases by the Debtors RELEASES BY THE DEBTORS. As of the Initial Effective Date, for good and valuable consideration, the adequacy of which is hereby confirmed, the Debtors as debtors in possession, will be deemed to forever release, waive and discharge all claims AS OF THE INITIAL EFFECTIVE DATE, FOR GOOD AND VALUABLE CONSIDERATION, THE ADEQUACY OF WHICH IS HEREBY **CONFIRMED. THE DEBTORS AS DEBTORS IN POSSESSION. WILL BE DEEMED TO FOREVER** RELEASE, WAIVE AND DISCHARGE ALL CLAIMS, obligations OBLIGATIONS, suits SUITS, judgments,JUDGMENTS, damagesDAMAGES, demandsDEMANDS, debtsDEBTS, rightsRIGHTS, causes of action and liabilities CAUSES OF ACTION AND LIABILITIES (other than the rights of the Debtors or OTHER THAN THE RIGHTS OF THE DEBTORS OR ATLT to enforce the Plan and the contracts TO ENFORCE THE PLAN AND THE CONTRACTS, instruments INSTRUMENTS, releases RELEASES, indentures and other agreements or documents delivered hereunderINDENTURES AND OTHER AGREEMENTS OR DOCUMENTS DELIVERED HEREUNDER) whether direct or derivative WHETHER DIRECT OR DERIVATIVE, liquidated or unliquidated LIQUIDATED OR UNLIQUIDATED, fixed or contingent FIXED OR CONTINGENT, matured or unmatured MATURED OR UNMATURED, disputed or undisputed DISPUTED OR UNDISPUTED, known or unknown KNOWN OR UNKNOWN, foreseen or unforeseen FORESEEN OR UNFORESEEN, then existing or thereafter arising THEN EXISTING OR THEREAFTER ARISING, in IN lawLAW, equity or otherwise that are based in whole or in part on any actEOUITY OR OTHERWISE THAT ARE BASED IN WHOLE OR IN PART ON ANY ACT, omissionOMISSION, transactionTRANSACTION, event or other occurrence taking place on or prior to the Initial Effective Date in any way relating to the Debtors EVENT OR OTHER OCCURRENCE TAKING PLACE ON OR PRIOR TO THE INITIAL EFFECTIVE DATE IN ANY WAY RELATING TO THE DEBTORS, the THE Chapter CHAPTER 11 Cases CASES, the THE PlanPLAN, or the Disclosure Statement OR THE DISCLOSURE STATEMENT, and that could have been asserted by or on behalf of the Debtors AND THAT COULD HAVE BEEN ASSERTED BY OR ON BEHALF OF THE DEB TORS, or their Estates against OR THEIR ESTATES AGAINST (aA) the current or former representatives THE CURRENT OR FORMER REPRESENTATIVES, directors DIRECTORS, officers and employees of the Debtors and the Debtors OFFICERS AND EMPLOYEES OF THE DEBTORS AND THE DEBTORS' agents AGENTS, advisors and Professionals ADVISORS AND PROFESSIONALS, in each case in their capacity as such IN EACH CASE IN THEIR CAPACITY AS SUCH, (bB) the Holders of THE HOLDERS OF ATI Note Claims and the NOTE CLAIMS AND THE ATI Note NOTE Trustees TRUSTEES, in each case in their capacity as such IN EACH CASE IN THEIR CAPACITY AS SUCH, (cC) the Holders of the Senior Lender Claims-THE HOLDERS OF THE SENIOR LENDER CLAIMS, (dD) the current or former members of the Creditors Committee and the advisors and attorneys for the Creditors Committee THE CURRENT OR FORMER MEMBERS OF THE CREDITORS COMMITTEE AND THE ADVISORS AND ATTORNEYS FOR THE CREDITORS COMMITTEE, in each case in their capacity as such IN EACH CASE IN THEIR CAPACITY AS SUCH, (eE) the THE Buyer BUYER, and AND (fE) the respective affiliates and current or former representatives THE RESPECTIVE AFFILIATES AND CURRENT OR FORMER REPRESENTATIVES. officers OFFICERS, directors DIRECTORS, employees EMPLOYEES, agents AGENTS, members MEMBERS,

direct and indirect shareholders.<u>DIRECT AND INDIRECT SHAREHOLDERS</u>, advisors.<u>ADVISORS</u>, attorneys and professionals of the foregoing<u>ATTORNEYS AND PROFESSIONALS OF THE FOREGOING</u>, in each case in their capacity as such<u>IN EACH CASE IN THEIR CAPACITY AS SUCH</u>; <u>providedPROVIDED</u>, <u>howeverHOWEVER</u>, nothing in this Section<u>NOTHING IN SECTION</u> 10.5 of the Plan shall effect a release in favor of any person other than the Debtors with respect to Causes of Action based on willful misconduct<u>OF THE</u> <u>PLAN SHALL EFFECT A RELEASE IN FAVOR OF ANY PERSON OTHER THAN THE DEBTORS</u> <u>WITH RESPECT TO CAUS ES OF ACTION BASED ON WILLFUL MISCONDUCT</u>, <u>criminal</u><u>CRIMINAL</u> conduct<u>CONDUCT</u>, misuse of confidential information that causes damage<u>MISUSE OF CONFIDENTIAL</u> <u>INFORMATION THAT CAUSES DAMAGE</u>, fraud<u>FRAUD</u>, ultra vires acts or gross negligence<u>ULTRA</u> <u>VIRES ACTS OR GROSS NEGLIGENCE</u>.

Releases by Holders of Claims and Equity Interests RELEASES BY HOLDERS OF 10.6 CLAIMS RECEIVING DISTRIBUTIONS. On the Initial Effective Date, each Holder of a Claim or Equity Interest shall be deemed to forever release, waive and discharge all Claims, Equity Interests, demands ON THE **INITIAL EFFECTIVE DATE. FOR GOOD AND VALUABLE CONSIDERATION. THE ADEOUACY OF** WHICH IS HEREBY CONFIRMED, IN CONSIDERATION FOR THE OBLIGATIONS OF THE DEBTORS UNDER THE PLAN, AND OTHER CONTRACTS, debts INSTRUMENTS, rights RELEASES, causes of action or liabilities AGREEMENTS OR DOCUMENTS TO BE DELIVERED IN CONNECTION WITH THE PLAN, whether direct or derivative EACH PERSON WHO, liquidated or unliquidated DIRECTLY **OR INDIRECTLY, fixed or contingentIS ENTITLED TO RECEIVE A DISTRIBUTION UNDER THE** PLAN, matured or unmatured INCLUDING PERSONS ENTITLED TO RECEIVE A DISTRIBUTION VIA AN ATTORNEY, disputed or undisputed AGENT, known or unknown INDENTURE TRUSTEE, foreseen or unforeseen OR SECURITIES INTERMEDIARY, then existing or thereafter arisingSHALL BE DEEMED TO FOREVER RELEASE, in lawWAIVE AND DISCHARGE ALL CLAIMS, equity or otherwise that are based in whole or in part on any act or omission DEMANDS, transaction DEBTS, event or other occurrence taking place on or prior to the Initial Effective Date in any way relating to the Debtors RIGHTS, the Chapter 11 Cases CAUSES OF ACTION OR LIABILITIES, the Plan or the Disclosure Statement against WHETHER DIRECT OR DERIVATIVE. LIOUIDATED OR UNLIOUIDATED, FIXED OR CONTINGENT, MATURED OR UNMATURED, DISPUTED OR UNDISPUTED, KNOWN OR UNKNOWN, FORESEEN OR UNFORESEEN, THEN EXISTING OR THEREAFTER ARISING, IN LAW, EQUITY OR OTHERWISE THAT ARE BASED IN WHOLE OR IN PART ON ANY ACT OR OMISSION. TRANSACTION. EVENT OR OTHER OCCURRENCE TAKING PLACE ON OR PRIOR TO THE INITIAL EFFECTIVE DATE IN ANY WAY RELATING TO THE DEBTORS, THE CHAPTER 11 CASES, THE PLAN OR THE DISCLOSURE STATEMENT AGAINST (aA) the current or former representatives THE CURRENT OR FORMER REPRESENTATIVES, directors DIRECTORS, officers and employees of the Debtors and the Debtors OFFICERS AND EMPLOYEES OF THE DEBTORS AND THE DEBTORS' agents AGENTS, advisors and Professionals ADVISORS AND PROFESSIONALS, in each case in their capacity as suchIN EACH CASE IN THEIR CAPACITY AS SUCH; (bB) the Holders of THE HOLDERS OF ATI Note Claims and the NOTE CLAIMS AND THE ATI Note NOTE Trustees TRUSTEES, in each case in their capacity as such IN EACH CASE IN THEIR CAPACITY AS SUCH, (cC) the Holders of the Senior Lender Claims THE HOLDERS OF THE SENIOR LENDER CLAIMS, (dD) the current or forme r members of the Creditors Committee and the advisors and attorneys for the Creditors Committee THE CURRENT OR FORMER MEMBERS OF THE **CREDITORS COMMITTEE AND THE ADVISORS AND ATTORNEYS FOR THE CREDITORS** COMMITTEE, in each case in their capacity as such IN EACH CASE IN THEIR CAPACITY AS SUCH, (+E) the THE BuyerBUYER, IN THEIR CAPACITY AS SUCH. (FF) the THE ATI NoteNOTE Trustees TRUSTEES, and (g) the respective affiliates and current or former representatives IN THEIR CAPACITY AS SUCH, officers AND (G) THE RESPECTIVE AFFILIATES AND CURRENT OR FORMER REPRESENTATIVES, directors OFFICERS, employees DIRECTORS, agents EMPLOYEES, members AGENTS, direct and indirect shareholders MEMBERS, advisors DIRECT AND INDIRECT SHAREHOLDERS, attorneys and professionals of the foregoingADVISORS, in each case in their capacity as such; providedATTORNEYS AND PROFESSIONALS OF THE FOREGOING, however IN EACH CASE IN THEIR CAPACITY AS SUCH: **PROVIDED**, nothing in this Section 10.6 of the Plan shall effect a release in favor of any person other than the Debtors with respect to Causes of Action based on willful misconduct HOWEVER, criminal conduct NOTHING IN SECTION 10.6 OF THE PLAN SHALL EFFECT A RELEASE IN FAVOR OF ANY PERSON OTHER THAN THE DEBTORS WITH RESPECT TO CAUSES OF ACTION BASED ON WILLFUL MISCONDUCT, misuse of confidential information that causes damageCRIMINAL CONDUCT, fraudMISUSE

OF CONFIDENTIAL INFORMATION THAT CAUSES DAMAGE, ultra vires acts or gross negligence FRAUD, ULTRA VIRES ACTS OR GROSS NEGLIGENCE.

Release of BuyerRELEASE OF BUYER. In consideration for the Cash and XO Common 10.7 Stock to be provided under the Purchase Agreement, (a) Buyer, its Affiliates (as defined in the Purchase Agreement) and all of Buver's direct and indirect subsidiaries. (b) Buver's and all of Buver's Affiliates (as defined in the Purchase Agreement) and all of Buyer's direct and indirect subsidiaries' members, officers, directors, employees, shareholders, partners, Representatives (as defined in the Purchase Agreement), consultants, attorneys, accountants, experts, financial advisors, investment bankers, dealer-managers, placement agents, and agents (each in their respective capacities as such), and (c) such other parties related to Buver and Buver's Affiliates (as defined in the Purchase Agreement) and Buyer's direct and indirect subsidiaries to which Section 10.8 of this Plan applies (each in their respective capacities as such), are hereby released from all claims, causes of action, and other assertions of liability of any Person or Entity whether directly, indirectly, derivatively, contractually, statutorily, and/or otherwise based on any actions taken or not taken, or on any other matter or circumstance whatsoever occurring, on or prior to the Initial Effective Date, relating to the Debtors' business, the Chapter 11 Cases, the Plan or Disclosure Statement; provided, however, that nothing herein shall relieve Buyer of its obligations under the Purchase Agreement and nothing in this Section 10.7 shall be construed as a release of any entity's fraud, malpractice, criminal conduct or willful misconduct. In no event shall Buyer have any liability or obligation for any Claim or Equity Interest in the Debtors arising prior to the Initial Effective Date, other than the Assumed Liabilities. ON THE INITIAL EFFECTIVE DATE. IN CONSIDERATION FOR THE CASH AND XO COMMON STOCK TO BE PROVIDED UNDER THE PURCHASE AGREEMENT, (A) BUYER, ITS AFFILIATES (AS DEFINED IN THE PURCHASE AGREEMENT) AND ALL OF BUYER'S DIRECT AND INDIRECT SUBSIDIARIES, (B) BUYER'S AND ALL OF BUYER'S AFFILIATES' (AS DEFINED IN THE PURCHASE AGREEMENT) AND ALL OF BUYER'S DIRECT AND INDIRECT SUBSIDIARIES' MEMBERS, OFFICERS, DIRECTORS. EMPLOYEES. SHAREHOLDERS. PARTNERS. REPRESENTATIVES (AS DEFINED IN THE PURCHASE AGREEMENT), CONSULTANTS, ATTORNEYS, ACCOUNTANTS, EXPERTS, FINANCIAL ADVISORS, INVESTMENT BANKERS, DEALER-MANAGERS, PLACEMENT AGENTS, AND AGENTS (EACH IN THEIR RESPECTIVE CAPACITIES AS SUCH), AND (C) SUCH OTHER PARTIES RELATED TO BUYER AND BUYER'S AFFILIATES (AS DEFINED IN THE PURCHASE AGREEMENT) AND BUYER'S DIRECT AND INDIRECT SUBSIDIARIES TO WHICH SECTION 10.8 OF THE PLAN APPLIES (EACH IN THEIR RESPECTIVE CAPACITIES AS SUCH), ARE RELEASED FROM ALL CLAIMS, CAUSES OF ACTION, AND OTHER ASSERTIONS OF LIABILITY OF ANY PERSON OR ENTITY WHETHER DIRECTLY, INDIRECTLY, DERIVATIVELY, CONTRACTUALLY, STATUTORILY, AND/OR OTHERWISE BASED ON ANY ACTIONS TAKEN OR NOT TAKEN. OR ON ANY OTHER MATTER OR CIRCUMSTANCE WHATSOEVER OCCURRING, ON OR PRIOR TO THE INITIAL EFFECTIVE DATE, RELATING TO THE DEBTORS, THE DEBTORS' BUSINESS, THE **CHAPTER 11 CASES. THE PLAN OR DISCLOSURE STATEMENT: PROVIDED. HOWEVER. THAT NOTHING SHALL RELIEVE BUYER OF ITS OBLIGATIONS UNDER THE PURCHASE AGREEMENT** AND NOTHING IN SECTION 10.7 OF THE PLAN SHALL BE CONSTRUED AS A RELEASE OF ANY ENTITY'S FRAUD. MALP RACTICE. CRIMINAL CONDUCT OR WILLFUL MISCONDUCT. IN NO EVENT SHALL BUYER HAVE ANY LIABILITY OR OBLIGATION FOR ANY CLAIM OR EQUITY INTEREST IN THE DEBTORS ARISING PRIOR TO THE INITIAL EFFECTIVE DATE, OTHER THAN THE ASSUMED LIABILITIES.

10.8 **Exculpation and Limitation of Liability**. Except as provided in the Plan or the Confirmation Order, none of the Debtors, the Buyer, the Holders of ATI Note Claims, the ATI Note Trustees, the Holders of the Senior Lender Claims, the Creditors Committee nor the individual members thereof, nor any of their respective present or former members, representatives, officers, directors, shareholders, employees, advisors, attorneys or agents acting in such capacity, shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Equity Interest, or any other party in interest, or any of their respective agents, direct or indirect shareholders, employees, representatives, financial advisors, attorneys or affiliates, or any of their respective successors or assigns, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, the pursuit of confirmation of the Plan, the consummation of the Plan, or the administration of the Plan or the property to be distributed under the Plan, except for their willful misconduct, criminal conduct, misuse of confidential information that causes damages, fraud, ultra vires acts or gross negligence, and in all respects shall be entitled to rely reasonably upon the advice of counsel with respect to their duties and responsibilities under the Plan.

10.9 **Discharge of Claims and Termination of Equity Interests**. Except as provided in the Purchase Agreement, Plan or the Confirmation Order, pursuant to section 1141(d) of the Bankruptcy Code, (a) the rights afforded under the Plan and the treatment of all Claims and Equity Interests herein, shall be in exchange for and in complete satisfaction, discharge and release of Claims and Equity Interests of any nature whatsoever, including any interest accrued on Claims from and after the Commencement Date, against any Debtor or any of its assets or properties, (b) on the Initial Effective Date or each of the NTA Effective Dates, as applicable, all such Claims against, and Equity Interests in, any Debtor shall be satisfied, discharged and released in full and (c) all Persons and Entities shall be precluded from asserting against Reorganized STFI, the Buyer and the Reorganized Subsidiaries, and their respective successors or their assets or properties any other or further Claims or Equity Interests based upon any act or omission, transaction or other activity of any kind or nature that occurred prior to the Initial Effective Dates, as applicable

10.10 Injunction INJUNCTION. All injunctions or stays provided for in the Chapter 11 Cases pursuant to sectionsALL INJUNCTIONS OR STAYS PROVIDED FOR IN THE CHAPTER 11 CASES PURSUANT TO SECTIONS 105 and AND 362 of the Bankruptcy Code or otherwise and in effect on the Confirmation DateOF THE BANKRUPTCY CODE OR OTHERWISE AND IN EFFECT ON THE CONFIRMATION DATE, shall remain in full force and effect until the closing of the Chapter 11 Cases pursuant to sectionSHALL REMAIN IN FULL FORCE AND EFFECT. PURSUANT TO SECTION 350(aA) of the Bankruptey Code OF THE BANKRUPTCY CODE. UNTIL THE EARLIEST OF (A) THE TIME THE CHAPTER 11 CASES ARE CLOSED. (B) THE TIME THE CHAPTER 11 CASES ARE DISMISSED. OR (C) THE TIME A DISCHARGE IS GRANTED OR DENIED. Subject to the occurrence of the Initial Effective DateSUBJECT TO THE OCCURRENCE OF THE INITIAL EFFECTIVE DATE, the entry of the Confirmation Order shall permanently enjoin all Persons that have held THE ENTRY OF THE CONFIRMATION ORDER SHALL PERMANENTLY ENJOIN FOREVER ALL PERSONS AND ENTITIES THAT HAVE HELD, currently hold or may hold a Claim or an Equity Interest in the Debtors from taking any of the following actions based on such Claim or Equity InterestCURRENTLY HOLD OR MAY HOLD A CLAIM OR AN EQUITY INTEREST IN THE DEBTORS FROM TAKING ANY OF THE FOLLOWING ACTIONS BASED ON SUCH CLAIM **OR EOUITY INTEREST, whetherWHETHER directly DIRECTLY, indirectly INDIRECTLY,** derivatively DERIVATIVELY, contractually CONTRACTUALLY, statutorily STATUTORILY and AND/orOR otherwiseOTHERWISE: (aA) commencingCOMMENCING, conducting or continuing in any mannerCONDUCTING OR CONTINUING IN ANY MANNER, directly or indirectly DIRECTLY OR INDIRECTLY, any ANY suit SUIT, action or other proceeding of any kind against any or all of the Debtors ACTION OR OTHER PROCEEDING OF ANY KIND AGAINST ANY OR ALL OF THE **DEBTORS**, the Reorganized Subsidiaries THE REORGANIZED SUBSIDIARIES, Reorganized REORGANIZED STFI, the THE Buyer BUYER (together with its Affiliates and its Affiliates TOGETHER WITH ITS AFFILIATES AND ITS AFFILIATES' respective RESPECTIVE members MEMBERS, shareholders SHAREHOLDERS, partners and Representatives PARTNERS AND **REPRESENTATIVES** (as defined in the Purchase AgreementAS DEFINED IN THE PURCHASE AGREEMENT)), THE CREDITORS COMMITTEE OR THE INDIVIDUAL MEMBERS THEREOF. ATLT or their respective property or assets OR THEIR RESPECTIVE PROPERTY OR ASSETS; (bB) enforcingENFORCING, levyingLEVYING, attachingATTACHING, collecting or otherwise recovering in any manner or by any means COLLECTING OR OTHERWISE RECOVERING IN ANY MANNER OR BY ANY MEANS, whether directly or indirectly WHETHER DIRECTLY OR INDIRECTLY, any ANY judgment, JUDGMENT, award AWARD, decree or order against any or all of the Debtors DECREE OR ORDER AGAINST ANY OR ALL OF THE DEBTORS, the THE Buyer BUYER (together with its Affiliates and its Affiliates TOGETHER WITH ITS AFFILIATES AND ITS AFFILIATES' respective RESPECTIVE members MEMBERS, shareholders SHAREHOLDERS, partners and Representatives PARTNERS AND **REPRESENTATIVES** (as defined in the Purchase AgreementAS DEFINED IN THE PURCHASE AGREEMENT)), ATLT or their respective property or assets OR THEIR RESPECTIVE PROPERTY OR ASSETS; (eC) creatingCREATING, perfecting or enforcing in any mannerPERFECTING OR ENFORCING IN ANY MANNER, directly or indirectly DIRECTLY OR INDIRECTLY, any Lien against any or all of the Debtors ANY LIEN AGAINST ANY OR ALL OF THE DEBTORS, the Reorganized Subsidiaries THE REORGANIZED SUBSIDIARIES, Reorganized REORGANIZED STFI, the THE Buyer BUYER (together with its Affiliates and its Affiliates TOGETHER WITH ITS AFFILIATES AND ITS AFFILIATES' respectiveRESPECTIVE membersMEMBERS, shareholdersSHAREHOLDERS, partners and Representatives PARTNERS AND REPRESENTATIVES (as defined in the Purchase AgreementAS DEFINED

IN THE PURCHASE AGREEMENT)), ATLT or their respective property or assets OR THEIR RESPECTIVE PROPERTY OR ASSETS; (dD) asserting any setoffASSERTING ANY SETOFF, right of subrogation or recoupment of any kindRIGHT OF SUBROGATION OR RECOUPMENT OF ANY KIND, directly or indirectly DIRECTLY OR INDIRECTLY, against any debtAGAINST ANY DEBT, liability or obligation due to the Debtors LIABILITY OR OBLIGATION DUE TO THE DEBTORS, the Reorganized Subsidiaries THE REORGANIZED SUBSIDIARIES, Reorganized REORGANIZED STFI, the THE Buyer BUYER (together with its Affiliates and its Affiliates TOGETHER WITH ITS AFFILIATES AND ITS AFFILIATES' respectiveRESPECTIVE members.MEMBERS, shareholders.SHAREHOLDERS, partners and Representatives PARTNERS AND REPRESENTATIVES (as defined in the Purchase AgreementAS DEFINED IN THE PURCHASE AGREEMENT)), ATLT or their respective property OR THEIR RESPECTIVE PROPERTY; and AND (eE) proceeding in any manner in any place whatsoever that does not conform to or comply with or is inconsistent with the provisions of the Plan or the Purchase Agreement PROCEEDING IN ANY MANNER IN ANY PLACE WHATSOEVER THAT DOES NOT CONFORM TO OR COMPLY WITH OR IS INCONSISTENT WITH THE PROVISIONS OF THE PLAN OR THE PURCHASE AGREEMENT. NOTWITHSTANDING THIS SECTION 5(L)(5), THE SET OFF RIGHTS OF ANY HOLDERS OF ALLOWED CLAIMS ARE PRESERVED TO THE EXTENT OF APPLICABLE LAW.

10.11 **<u>Release of Liens</u>**. On the Initial Effective Date, except as otherwise provided in the Plan, all mortgages, deeds of trust, Liens or other security interests against the property of the Estates shall be released except Buyer's Lien on certain assets set forth in Section 10.12 of the Plan.

10.12 **Buyer's Lien on Certain Assets.** Beginning on the Initial Effective Date, Buyer solely shall have a valid, first-priority, perfected Lien on any and all of the Non-Transferred Assets. The Confirmation Order shall permanently enjoin all Persons that have held, currently hold or may hold a Claim against or an Equity Interest in the Debtors or any Interest (as defined in the Sale Order) in the Non-Transferred Assets from taking any of the actions listed in Section 10.10 of the Plan that relate to the Non-Transferred Assets and/or are against Buyer or any individual, partnership, joint venture, corporation, business trust, limited liability company, trust, unincorporated organization, joint stock company, labor union, estate, governmental entity, or any other entity that Buyer directly or indirectly controls or is under common control with. Buyer's Lien on each Non-Transferred Asset shall remain valid and enforceable until the earlier of (a) the transfer of such Non-Transferred Asset from any Debtor to Buyer or (b) the receipt by Buyer of all of the proceeds from the sale of such Non-Transferred Asset.

10.13 **Applicability of Section 1125 of the Bankruptcy Code.** The protection afforded by section 1125(e) of the Bankruptcy Code with regard to the solicitation of acceptances or rejections of the Plan and with regard to the offer, issuance, sale or purchase of the New STFI Common Stock, XO Common Stock, or any other security, shall apply to the full extent provided by law, and the entry of the Confirmation Order shall constitute the determination by the Bankruptcy Court that the Debtors, Buyer (together with its Affiliates and its Affiliates' respective members, shareholders, partners and Representatives (as defined in the Purchase Agreement)), the Creditors Committee (and any subcommittee thereof), the Senior Lenders, and each of their respective officers, directors, partners, employees, members, agents, attorneys, accountants, financial advisors, investment bankers, dealer-managers, placement agents, and other professionals, shall have acted in good faith and in compliance with the applicable provisions of the Bankruptcy Code pursuant to section 1125(e) of the Bankruptcy Code and that the Plan has been proposed in good faith and not by any means forbidden by law.

ARTICLE XI

RETENTION OF JURISDICTION

11.1 **Retention of Jurisdiction**. The Bankruptcy Court shall have exclusive jurisdiction of all matters arising out of, and related to, the Chapter 11 Cases, the Plan and the Purchase Agreement pursuant to, and for the purposes of, sections 105(a) and 1142 of the Bankruptcy Code and for, among other things, the following purposes:

(a) To hear and determine pending applications for the assumption or rejection of executory contracts or unexpired leases, if any are pending, and the allowance of cure amounts and Claims resulting therefrom;

<u>37</u>

(b) To hear and determine the allowance of Claims arising out of or relating to Utility Services and Tariff Services;

(c) To hear and determine any and all adversary proceedings, applications and contested matters;

(d) To hear and determine any objections to Administrative Expense Claims, or any objections to or requests to estimate, any Claims;

(e) To enter and implement such orders as may be appropriate in the event the Confirmation Order is for any reason stayed, revoked, modified or vacated;

(f) To issue such orders in aid of execution and consummation of the Plan, to the extent authorized by section 1142 of the Bankruptcy Code;

(g) To consider any amendments to, or modifications of, the Plan, to cure any defect or omission, or reconcile any inconsistency in any order of the Bankruptcy Court, including without limitation, the Confirmation Order or in the Plan;

(h) To hear and determine all applications for compensation and reimbursement of expenses of Professionals under sections 328, 330, 331 and 503(b) of the Bankruptcy Code;

(i) To hear and determine disputes arising in connection with the interpretation, implementation or enforcement of the Plan, any documents contained in the Plan Supplement, the Purchase Agreement and the Confirmation Order, including but not limited to Claims for indemnification thereunder;

(j) To hear and determine disputes involving ATLT;

(k) To recover all assets of the Debtors and property of the Debtors' Estates, wherever located;

(1) To hear and determine all Avoidance Actions and Causes of Action which may be brought by ATLT;

(m) To hear and determine all disputes relating to Section 10.12 of the Plan including, but not limited to, disputes relating to the validity and enforcement of Buyer's liens on the Non-Transferred Assets and the entering, implementing, and/or enforcing of such orders and/or injunctions as may be appropriate hereunder. Buyer shall have standing to assert claims or actions which are in any way related to or arising out of or based on or in any way connected with this provision, the Plan (including without limitation, Section 10.12 of the Plan) and the Purchase Agreement.

(n) To hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505 and 1146 of the Bankruptcy Code;

(o) To hear and determine matters concerning the Disputed Claims Reserve;

(p) To hear any other matter not inconsistent with the Bankruptcy Code; and

(q) To enter a final decree closing the Chapter 11 Cases.

ARTICLE XII

MISCELLANEOUS PROVISIONS

12.1 **Dissolution of the Creditors Committee.** On the Initial Effective Date, the Creditors Committee shall be dissolved and the members thereof shall be released and discharged of and from all further authority, duties, responsibilities and obligations relating to and arising from and in connection with the Chapter 11 Cases, and the retention or its employment of its attorneys or other Professionals shall terminate; provided, however, that the

Creditors Committee shall continue to exist after such date with respect to (a) all applications filed pursuant to sections 328, 330 and 331 of the Bankruptcy Code seeking payment of fees and expenses incurred and (b) any matters pending as of the Initial Effective Date in the Chapter 11 Cases until such matters are finally resolved, including the Distributions under the Plan and any actions taken in connection with the Avoidance Actions and/or Causes of Action.

12.2 **Payment of Statutory Fees.** With respect to each Chapter 11 Case, all fees payable pursuant to section 1930 of title 28 of the United States Code, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid for each quarter until such Chapter 11 Case is converted, dismissed or closed, whichever occurs first. The Buyer and the Reorganized Subsidiaries shall have no liability for such fees.

12.3 **Payment of ATI Note Trustee Fees.** On the Initial Effective Date, the ATI Note Trustees (together with their attorneys, advisors and professionals) shall receive payment of any outstanding fees and reimbursement of expenses incurred by such Entities in connection with their role as an ATI Note Trustee. To the extent that payment of such fees is not made in full on the Initial Effective Date, the ATI Note Trustees will retain a lien on the Available Cash to the extent of any deficiency.

12.4 **Modification of the Plan**. Subject to the limitations contained in the Plan and the Purchase Agreement (a) the Debtors reserve the right, in accordance with the Bankruptcy Code and the Bankruptcy Rules, to alter, amend or modify the Plan prior to the entry of the Confirmation Order, <u>provided</u>, <u>however</u>, that any material alteration, amendment or modification of the Plan shall be subject to (i) the consent of the Creditors Committee, which shall not be unreasonably withheld, and (ii) the <u>prior</u> written consent of the Buyer, and (b) after the entry of the Confirmation Order, the Debtors, ATLT, the Reorganized Subsidiaries or Reorganized STFI, as the case may be, may (in each case with the consent of the Creditors Committee and the Senior Lenders, which consent may not be unreasonably withheld, delayed or conditioned by either the Creditors Committee or the Senior Lenders <u>and the</u> <u>prior written consent of the Buyer</u>), upon order of the Bankruptcy Court, amend or modify the Plan, in accordance with section 1127(b) of the Bankruptcy Code, or remedy any defect or omission or reconcile any inconsistency in the Plan in such manner as may be necessary to carry out the purpose and intent of the Plan. A Holder of a Claim or Equity Interest that has accepted the Plan shall be deemed to have accepted the Plan as altered, amended or modified, if the proposed alteration, amendment or modification does not materially and adversely change the treatment of the Claim of such Holder.

12.5 <u>Severability</u>. In the event that the Bankruptcy Court determines that any provision in the Plan is invalid, void or unenforceable, such provision shall be invalid, void or unenforceable with respect to the Holder or Holders of such Claims or Equity Interests as to which the provision is determined to be invalid, void or unenforceable. Subject to the consent of Buyer which shall not be unreasonably withheld, delayed or conditioned, the invalidity, voidness or unenforceability of any such provision shall in no way limit or affect the enforceability and operative effect of any other provision of the Plan.

12.6 **Revocation or Withdrawal of the Plan**. The Debtors reserve the right to revoke or withdraw the Plan prior to the Confirmation Date with the prior written consent of the Buyer, unless the Purchase Agreement shall have been terminated. If the Debtors revoke or withdraw the Plan prior to the Confirmation Date, then the Plan shall be deemed null and void. In such event, nothing contained in the Plan shall constitute or be deemed a waiver or release of any claims by or against the Debtors or any other Person or Entity or to prejudice in any manner the rights of the Debtors or any Person or Entity in any further proceedings involving the Debtors.

12.7 <u>Section 1145 Exemption.</u> Section 1145 (a) of the Bankruptcy Code provides that certain federal, state and local requirements regarding registration of securities do not apply to securities which are offered or sold under a plan of reorganization of a security of a debtor, of an affiliate participating in a joint plan with a debtor or of a successor to a debtor under the plan of reorganization. The Debtors have not obtained, and do not intend to obtain, a "no-action" letter from the Securities and Exchange Commission to the effect that the Securities and Exchange Commission will not take enforcement action if the New STFI Common Stock, and the XO Common Stock are issued in accordance with the provisions of the Plan without registration under the Securities Act of 1933.

12.8 <u>Section 1146 Exemption.</u> Pursuant to section 1146(c) of the Bankruptcy Code, any transfers of property pursuant hereto or, if applicable, pursuant to the Purchase Agreement, shall not be subject to any document recording tax, stamp tax, conveyance fee, intangibles or similar tax, mortgage tax, stamp act, real estate transfer tax, mortgage recording tax or other similar tax or governmental assessment in the United States, and the Confirmation Order shall direct the appropriate state or local governmental officials or agents to forgo the collection of any such tax or governmental assessment and to accept for filing and recordation any of the foregoing instruments or other documents without the payment of any such tax or governmental assessment.

12.9 **Notices**. All notices, requests and demands to or upon the Debtors or, on and after the Initial Effective Date, ATLT, to be effective shall be in writing and, unless otherwise expressly provided in the Plan, shall be deemed to have been duly given or made when actually delivered by messenger or required overnight courier service or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

If to the Debtors or ATLT:

Allegiance Telecom Liquidating Trust-

Attn: _____ Facsimile:

with a copy to:

Kirkland & Ellis LLP Citigroup Center 153 East 53rd Street New York, New York 10022 Attn: Jonathan S. Henes, Esq. Facsimile: (212) 446-4900

If to the Buyer:

XO Communications Inc. 1111 Sunset Hills Road Reston, Virginia 20190 Attn: Lee Weiner, Senior Vice President and General Counsel Facsimile: (703) 547-2025

with a copy to:

Brown Rudnick Berlack Israels LLP One Financial Center Boston, Massachusetts 02111 Attn: Steven D. Pohl, Esq. Facsimile: (617) 856-8201

If to the Creditors Committee:

Akin Gump Strauss Hauer & Feld LLP 590 Madison Avenue New York, New York 10022 Attn: Ira S. Dizingoff, Esq. Facsimile: (212) 872-1002

If to the Senior Lenders:

Paul, Hastings, Janofsky & Walker LLP 600 Peachtree Street, N.E., 24th Floor Atlanta, Georgia 30308 Attn: Jesse Austin III, Esq. Facsimile: (404) 815-2424

12.10 **Governing Law**. Except to the extent the Bankruptcy Code, Bankruptcy Rules or other federal law is applicable, or to the extent an exhibit to the Plan provides otherwise, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of the State of New York, without giving effect to the principles of conflicts of law of such jurisdiction.

12.11 <u>Withholding and Reporting Requirements</u>. In connection with the consummation of the Plan, the Debtors, the Reorganized Subsidiaries, Reorganized STFI, and ATLT, as the case may be, shall comply with all withholding and reporting requirements imposed by any federal, state, local or foreign taxing authority and all Distributions hereunder shall be subject to any such withholding and reporting requirements.

Dated: New York, New York March 18, <u>April 21</u>, 2004

By: /s/ Mark B. Tresnowski

Mark B. Tresnowski Executive Vice President General Counsel and Secretary (Allegiance Telecom, Inc.) Document comparison done by DeltaView on Tuesday, April 20, 2004 18:07:27

| Document 1 | file://I:\Allegiance Telecom, Inc\Plan\XO Plan (sub con) v12 [FILED].doc |
|---------------|---|
| Document 2 | file://I:\Allegiance Telecom, Inc\Plan\Amended XO Plan v.2.doc |
| Rendering set | Basic K&E |

| Legend: | |
|-------------------|--|
| Insertion | |
| Deletion | |
| Moved from | |
| <u>Moved to</u> | |
| Format change | |
| Moved deletion | |
| Inserted cell | |
| Deleted cell | |
| Moved cell | |
| Split/Merged cell | |
| Padding cell | |

| Redline | Redline Summary: | | |
|---------|------------------|--|--|
| No. | Change | Text | |
| 1 | Deletion | 33 | |
| 2 | Insertion | 36 | |
| 3-6 | Change | "Telecom, Inc\Plan\XO Plan (sub con) v12.doc" changed to "Telecom, Inc\Plan\Amended XO Plan v.2.doc" | |
| 7 | Change | "DEBTORS' JOINT PLAN OF" changed to "DEBTORS' AMENDED JOINT PLAN OF" | |
| 8-9 | Change | "March 18, 2004" changed to "April 21, 2004" | |
| 10-13 | Change | "Telecom, Inc\Plan\XO Plan (sub con) v12.doc" changed to "Telecom, Inc\Plan\Amended XO Plan v.2.doc" | |
| 14-15 | Change | "Liabilities 1" changed to "Liabilities 2" | |
| 16-17 | Change | "1.7 ATCW1" changed to "1.7 ATCW2" | |
| 18-19 | Change | "1.17 ATLT Agreement 2" changed to "1.17 ATLT Agreement 3" | |
| 20-21 | Change | "1.18 ATLT Assets 2" changed to "1.18 ATLT | |

| | | Assets 3" |
|-------|--------|--|
| 22-23 | Change | "1.19 ATLT A Certificates 2" changed to "1.19 ATLT "A" Trust Interests 3" |
| 24-25 | Change | "1.20 ATLT B Certificates 2" changed to "1.20 ATLT "B" Trust Interests 3" |
| 26-27 | Change | "1.21 ATLT C Certificates 2" changed to "1.21 ATLT "C" Trust Interests 3" |
| 28-29 | Change | "1.22 ATLT Certificates 2" changed to "1.22 ATLT Trust Interests 3" |
| 30-31 | Change | "1.23 Available Cash 2" changed to "1.23 Available Cash 3" |
| 32-33 | Change | "1.27 Bankruptcy Court 3" changed to "1.27 Bankruptcy Court 4" |
| 34-35 | Change | "1.28 Bankruptcy Rules 3" changed to "1.28 Bankruptcy Rules 4" |
| 36-37 | Change | "1.29 Bar Date 3" changed to "1.29 Bar Date 4" |
| 38-39 | Change | "1.30 Business Day 3" changed to "1.30 Business Day 4" |
| 40-41 | Change | "1.31 Buyer 3" changed to "1.31 Buyer 4" |
| 42-43 | Change | "1.32 Cash 3" changed to "1.32 Cash 4" |
| 44-45 | Change | "1.33 Cash Recovery 3" changed to "1.33 Cash Recovery 4" |
| 46-47 | Change | "Recovery Election 3" changed to "Recovery Election 4" |
| 48-49 | Change | "1.35 Causes of Action 3" changed to "1.35 Causes of Action 4" |
| 50-51 | Change | "1.36 Chapter 11 Cases 4" changed to "1.36 Chapter 11 Cases 5" |
| 52-53 | Change | "1.37 Claim 4" changed to "1.37 Claim 5" |
| 54-55 | Change | "1.38 Claim Holder 4" changed to "1.38 Claim Holder 5" |
| 56-57 | Change | "Objection Deadline 4" changed to "Objection Deadline 5" |
| 58-59 | Change | "1.40 Class 4" changed to "1.40 Class 5" |
| 60-61 | Change | "1.41 Closing 4" changed to "1.41 |
| · | | |

| | | Closing 5" |
|-------|--------|---|
| 62-63 | Change | "1.42 Commencement Date4" changedto "1.42Commencement Date5" |
| 64-65 | Change | "1.43 Common Stock 4" changed to "1.43 Common Stock 5" |
| 66-67 | Change | "1.44 Confirmation Date 4" changed to "1.44 Confirmation Date 5" |
| 68-69 | Change | "1.45 Confirmation Hearing 4" changed to "1.45 Confirmation Hearing 5" |
| 70-71 | Change | "1.46 Confirmation Order 4" changed to "1.46 Confirmation Order 6" |
| 72-73 | Change | "1.47 Credit Agreement 4" changed to "1.47 Credit Agreement 6" |
| 74-75 | Change | "1.48 Creditors Committee4" changedto "1.48Creditors Committee6" |
| 76-77 | Change | "1.49 Debtors 4" changed to "1.49 Debtors 6" |
| 78-79 | Change | "Possession 5" changed to "Possession6" |
| 80-81 | Change | "1.51 Disclosure Schedules 5" changed to "1.51 Disclosure Schedules 6" |
| 82-83 | Change | "1.52 Disclosure Statement 5" changed to "1.52 Disclosure Statement 6" |
| 84-85 | Change | "Hearing Date 5" changed to "Hearing Date 6" |
| 86-87 | Change | "Statement Order 5" changed to "Statement Order 7" |
| 88-89 | Change | "1.55 Disputed Claim 5" changed to "1.55 Disputed Claim 7" |
| 90-91 | Change | "Claim Reserve 5" changed to "Claim Reserve 7" |
| 92-93 | Change | "Claim Amount 5" changed to "Claim Amount 7" |
| 94-95 | Change | "1.58 Distribution 5" changed to "1.58 Distribution 7" |
| 96-97 | Change | "1.59 Distribution Account 5" changed to "1.59 Distribution Account 7" |
| 98-99 | Change | "1.60 Distribution Date 5" changed to "1.60 |

| | | Distribution Date 7" |
|---------|--------|---|
| 100-101 | Change | "1.61 Entity 5" changed to "1.61 Entity 7" |
| 102-103 | Change | "1.62 Equity Interest 5" changed to "1.62 Equity Interest 7" |
| 104-105 | Change | "1.63 Estates 6" changed to "1.63 Estates 7" |
| 106-107 | Change | "1.64 Excluded Assets 6" changed to "1.64 Excluded Assets 7" |
| 108-109 | Change | "Liabilities 6" changed to "Liabilities 8" |
| 110-111 | Change | "Judgment Rate 6" changed to "Judgment Rate 8" |
| 112-113 | Change | "Distribution Date 6" changed to "Distribution Date 8" |
| 114-115 | Change | "Effective Date 6" changed to "Effective Date 8" |
| 116-117 | Change | "1.69 Final Order 6" changed to "1.69 Final Order 8" |
| 118-119 | Change | "1.70 Holder 6" changed to "1.70 Holder 8" |
| 120-121 | Change | "1.71 Impaired 6" changed to "1.71 Impaired 8" |
| 122-123 | Change | "1.72 Impaired Claim 6" changed to "1.72 Impaired Claim 8" |
| 124-125 | Change | "1.73 Impaired Class 6" changed to "1.73 Impaired Class 8" |
| 126-127 | Change | "Effective Date 6" changed to "Effective Date 8" |
| 128-129 | Change | "1.75 Insured Claim 6" changed to "1.75 Insured Claim 9" |
| 130-131 | Change | "1.76 Intercompany Claim6" changed to "1.76 Intercompany Claim9" |
| 132-133 | Change | "Subsidiaries 6" changed to "Subsidiaries 9" |
| 134-135 | Change | "1.78 Lien 7" changed to "1.78 Lien 9" |
| 136-137 | Change | "1.79 Litigation Cash 7" changed to "1.79 Litigation Cash 9" |
| 138-139 | Change | "Common Stock 7" changed to "Common Stock 9" |
| 140-141 | Change | "1.81 Non-Transferred Assets 7" changed to "1.81 Non-Transferred Assets 9" |
| 142-143 | Change | "Effective Dates 7" changed to "Effective Dates 9" |
| 144-145 | Change | "Subsidiaries 7" changed to "Subsidiaries 9" |
| 146-147 | Change | "1.84 Operating Agreement7" changedto "1.84Operating Agreement9" |

| 148-149 | Change | "1.85 Person 7" changed to "1.85 Person 10" |
|---------|--------|--|
| 150-151 | Change | "1.86 Plan 7" changed to "1.86 Plan 10" |
| 152-153 | Change | "Administrator7" changed to"Administrator10" |
| 154-155 | Change | "Supplement 7" changed to "Supplement 10" |
| 156-157 | Change | "and Expenses 7" changed to "and Expenses 10" |
| 158-159 | Change | "Non-Tax Claim 7" changed to "Non-Tax Claim 10" |
| 160-161 | Change | "1.91 Priority Tax Claim 7" changed to "1.91 Priority Tax Claim 10" |
| 162-163 | Change | "1.92 Professional 8" changed to "1.92 Professional 10" |
| 164-165 | Change | "Fee Claims 8" changed to "Fee Claims 10" |
| 166-167 | Change | "1.94 Purchase Agreement 8" changed to "1.94 Purchase Agreement 11" |
| 168-169 | Change | "1.95 Record Date 8" changed to "1.95 Record Date 11" |
| 170-171 | Change | "1.96 Reorganized STFI 8" changed to "1.96 Reorganized STFI 11" |
| 172-173 | Change | "Subsidiaries 8" changed to "Subsidiaries 11" |
| 174-175 | Change | "1.98 Reserve Fund8" changed to "1.98Reserve Fund11" |
| 176-177 | Change | "1.99 Sale Order 8" changed to "1.99 Sale Order 11" |
| 178-179 | Change | "Transaction 8" changed to "Transaction11" |
| 180-181 | Change | "1.101 Schedules 8" changed to "1.101 Schedules 11" |
| 182-183 | Change | "1.102 Secured Claim 8" changed to "1.102 Secured Claim 11" |
| 184-185 | Change | "Discount Notes 8" changed to "Discount Notes 12" |
| 186-187 | Change | "Note Indenture 8" changed to "Note Indenture 12" |
| 188-189 | Change | "Note Trustee 8" changed to "Note Trustee 12" |
| 190-191 | Change | "Lender Claims 8" changed to "Lender Claims 12" |
| 192-193 | Change | "1.107 Senior Lenders9" changed to"1.107 Senior Lenders12" |
| 194-195 | Change | "Note Indenture 9" changed to "Note Indenture 12" |

| 196-197 | Change | "Note Trustee 9" changed to "Note Trustee 12" |
|---------|--------|--|
| 198-199 | Change | "1.110 Senior Notes 9" changed to "1.110 |
| 100 100 | | Senior Notes 12" |
| 200-201 | Change | "1.111 STFI 9" changed to "1.111 STFI 12" |
| 202-203 | Change | "1.112 STFI Assets 9" changed to "1.112 STFI Assets 12" |
| 204-205 | Change | "Cash Amount 9" changed to "Cash Amount 12" |
| 206-207 | Change | "1.114 Subordinated Claim 9" changed to "1.114 Subordinated Claim 12" |
| 208-209 | Change | "1.115 Subsidiaries 9" changed to "1.115 Subsidiaries 13" |
| 210-211 | Change | "1.116 Tariff 9" changed to "1.116 Tariff 13" |
| 212-213 | Change | "1.117 Tariff Services 9" changed to "1.117 Tariff Services 13" |
| 214-215 | Change | "1.118 Tax Rate 9" changed to "1.118 Tax Rate 13" |
| 216-217 | Change | "Services Agreement 9" changed to "Services Agreement 13" |
| 218-219 | Change | "1.120 Unencumbered Cash 9" changed to "1.120 Unencumbered Cash 13" |
| 220-221 | Change | "1.121 Unimpaired 9" changed to "1.121 Unimpaired 13" |
| 222-223 | Change | "1.122 Unimpaired Claim 9" changed to "1.122 Unimpaired Claim 13" |
| 224-225 | Change | "1.123 Unimpaired Class 9" changed to "1.123 Unimpaired Class 13" |
| 226-227 | Change | "1.124 Unsecured Claims 9" changed to "1.124 Unsecured Claims 13" |
| 228-229 | Change | "1.125 Utility Company 10" changed to "1.125 Utility Company 13" |
| 230-231 | Change | "1.126 Utilities Order10" changed to"1.126 Utilities Order14" |
| 232-233 | Change | "1.127 Utility Services 10" changed to "1.127 Utility Services 14" |
| 234-235 | Change | "Service Order 10" changed to "Service Order 14" |
| 236-237 | Change | "1.129 Voting Deadline 10" changed to "1.129 Voting Deadline 14" |
| 238-239 | Change | "Record Date 10" changed to "Record Date 14" |
| 240-241 | Change | "1.131 Wind-Down Amount 10" changed to |

| | | "1.131 Wind-Down Amount 14" |
|---------|-----------|--|
| | _ | "1.132 XO 10" changed to "1.132 XO |
| 242-243 | Change | 14" |
| 244-245 | Change | "Common Stock 10" changed to "Common Stock 14" |
| 246-247 | Change | "TAX CLAIMS 11" changed to "TAX CLAIMS 15" |
| 248-249 | Change | "Expense Claims 11" changed to "Expense Claims 15" |
| 250-251 | Change | "Fee Claims 11" changed to "Fee Claims 15" |
| 252-253 | Change | "Tax Claims 11" changed to "Tax Claims 16" |
| 254-255 | Change | "EQUITY INTERESTS 12" changed to "EQUITY INTERESTS 16" |
| 256-257 | Change | "Claims (Class 1) 12" changed to "Claims (Class 1) 16" |
| 258-259 | Change | "Claims (Class 2) 12" changed to "Claims (Class 2) 17" |
| 260-261 | Change | "Claims (Class 3) 13" changed to "Claims (Class 3) 17" |
| 262-263 | Change | "Claims (Class 4) 13" changed to "Claims (Class 4) 17" |
| 264-265 | Change | "Claims (Class 5) 13" changed to "Claims (Class 5) 18" |
| 266-267 | Change | "Claims (Class 6) 14" changed to "Claims (Class 6) 19" |
| 268-269 | Change | "Interests (Class 7) 14" changed to "Interests (Class 7) 19" |
| 270-271 | Change | "OF THE PLAN 14" changed to "OF THE PLAN 19" |
| 272-273 | Change | "Unimpaired Claims.14" changed to"Unimpaired Claims.19" |
| 274-275 | Change | "Entitled to Vote 15" changed to "Entitled to Vote 19" |
| 276-277 | Change | "Impaired Classes 15" changed to "Impaired Classes 20" |
| 278-279 | Change | "4.4 Cramdown 15" changed to "4.4 Cramdown 20" |
| 280-281 | Change | "IMPLEMENTATION 15" changed to "IMPLEMENTATION 20" |
| 282 | Insertion | 5.1 SubstantivePlan Purposes Only 20 |
| 283-284 | Change | "5.1 Deemed Consolidation" changed to "5.2 Deemed Consolidation" |
| 285-286 | Change | "Purposes Only. 15" changed to "Purposes |

| | | Only. Error! Bookmark not defined." |
|---------|-----------|--|
| 287-288 | Change | "5.2 Settlement" changed to "5.3 Settlement" |
| 289-290 | Change | "Certain Claims. 15" changed to "Certain Claims. 21" |
| 291-292 | Change | "5.3 Restructuring" changed to "5.4 Restructuring" |
| 293-294 | Change | "Transactions. 15" changed to "Transactions. 21" |
| 295 | Deletion | 5.4 ATLT. 18 |
| 296 | Insertion | 5.5 ATLT. 24 |
| 297 | Change | "Cancellation" changed to "5.6 Cancellation" |
| 298-299 | Change | "Agreements 20" changed to "Agreements 27" |
| 300-301 | Change | "5.6 Regulatory" changed to "5.7 Regulatory" |
| 302-303 | Change | "Approvals. 20" changed to "Approvals. 28" |
| 304-305 | Change | "5.7 Insurance Preservation" changed to "5.8 Insurance Preservation" |
| 306-307 | Change | "Preservation20" changed to "Preservation 28" |
| 308-309 | Change | "UNEXPIRED LEASES 20" changed to "UNEXPIRED LEASES 28" |
| 310-311 | Change | "Unexpired Leases 20" changed to "Unexpired Leases 28" |
| 312-313 | Change | "6.2 Utility Services. 21" changed to "6.2 Utility ServicesBookmark not defined." |
| 314-315 | Change | "6.3 Tariff Services. 22" changed to "6.3 Tariff Services. Error! Bookmark not defined." |
| 316-317 | Change | "DISTRIBUTIONS 22" changed to "DISTRIBUTIONS 30" |
| 318-319 | Change | "Effective Date 22" changed to "Effective Date 30" |
| 320-321 | Change | "Distributions by ATLT 22" changed to "Distributions by ATLT 30" |
| 322-323 | Change | "Distributions 22" changed to "Distributions 30" |
| 324-325 | Change | "Requirements/Allocations 23" changed to "Requirements/Allocations 31" |
| 326-327 | Change | "Distribution 23" changed to "Distribution 31" |

| 328 | Change | "7.6 Fractional ATLTDe Minimis Distributions" changed to "7.6 De Minimis Distributions" |
|---------|--------|---|
| 329-330 | Change | "Distributions 23" changed to "Distributions 31" |
| 331-332 | Change | "Recoupments23" changed to"Recoupments31" |
| 333-334 | Change | "Securities 23" changed to "Securities 32" |
| 335-336 | Change | "7.9 ATI Notes 23" changed to "7.9 ATI Notes 32" |
| 337-338 | Change | "DISPUTED CLAIMS 23" changed to "DISPUTED CLAIMS 32" |
| 339-340 | Change | "8.1 Objections to Claims23" changedto "8.1 Objections to Claims32" |
| 341-342 | Change | "Pending Allowance 24" changed to "Pending Allowance 32" |
| 343-344 | Change | "Injury Claims 24" changed to "Injury Claims 32" |
| 345-346 | Change | "8.4 Estimation of Claims24" changedto "8.4 Estimation of Claims33" |
| 347-348 | Change | "Insured Claims 24" changed to "Insured Claims 33" |
| 349-350 | Change | "Claims Reserve 24" changed to "Claims Reserve 33" |
| 351-352 | Change | "After Allowance 25" changed to "After Allowance 33" |
| 353-354 | Change | "Disallowance 25" changed to "Disallowance 34" |
| 355-356 | Change | "Impairment 25" changed to "Impairment 34" |
| 357-358 | Change | "OF THE PLAN 25" changed to "OF THE PLAN 34" |
| 359-360 | Change | "Effective Date 25" changed to "Effective Date 34" |
| 361-362 | Change | "Effective Date 25" changed to "Effective Date 34" |
| 363-364 | Change | "Conditions 26" changed to "Conditions 35" |
| 365-366 | Change | "Conditions 26" changed to "Conditions 35" |
| 367-368 | Change | "CONFIRMATION OF PLAN26" changedto "CONFIRMATION OF PLAN35" |
| 369-370 | Change | "10.1 Binding Effect26" changed to"10.1 Binding Effect35" |
| 371-372 | Change | "Injunction or Stays 26" changed to "Injunction |

| | | or Stays 35" |
|---------|--------|---|
| 373-374 | Change | "Avoidance Actions. 26" changed to |
| 575-574 | Change | "Avoidance Actions. 35" |
| 375-376 | Change | "of Rights. 26" changed to "of Rights. 36" |
| 377-378 | Change | "the Debtors 27" changed to "the Debtors 36" |
| 379-380 | Change | "Holders of Claims and Equity Interests 27" changed to "Holders of Claims Receiving Distributions 36" |
| 381-382 | Change | "10.7 Release of Buyer. 28" changed to "10.7 Release of Buyer. 37" |
| 383-384 | Change | "of Liability 28" changed to "of Liability 37" |
| 385-386 | Change | "Equity Interests 28" changed to "Equity Interests 37" |
| 387-388 | Change | "10.10 Injunction 28" changed to "10.10 Injunction 38" |
| 389-390 | Change | "10.11 Release of Liens 29" changed to "10.11 Release of Liens 38" |
| 391-392 | Change | "Certain Assets. 29" changed to "Certain Assets. 38" |
| 393-394 | Change | "Bankruptcy Code. 29" changed to "Bankruptcy Code. 39" |
| 395-396 | Change | "JURISDICTION 29" changed to "JURISDICTION 39" |
| 397-398 | Change | "Jurisdiction 29" changed to "Jurisdiction 39" |
| 399-400 | Change | "PROVISIONS 30" changed to "PROVISIONS 40" |
| 401-402 | Change | "Committee. 30" changed to "Committee. 40" |
| 403-404 | Change | "Statutory Fees. 31" changed to "Statutory Fees. 40" |
| 405-406 | Change | "Trustee Fees. 31" changed to "Trustee Fees. 40" |
| 407-408 | Change | "of the Plan 31" changed to "of the Plan 40" |
| 409-410 | Change | "12.5 Severability 31" changed to "12.5 Severability 41" |
| 411-412 | Change | "of the Plan 31" changed to "of the Plan 41" |
| 413-414 | Change | "Exemption. 31" changed to "Exemption. 41" |
| 415-416 | Change | "Exemption. 31" changed to "Exemption. 41" |
| 417-418 | Change | "12.9 Notices 32" changed to "12.9 |

| | | Notices 41" |
|---------|-----------|--|
| 419-420 | Change | "12.10 Governing Law 33" changed to "12.10 Governing Law 42" |
| 421-422 | Change | "Requirements 33" changed to "Requirements 43" |
| 423 | Deletion | 33 |
| 424 | Insertion | 38 |
| 425-428 | Change | "Telecom, Inc\Plan\XO Plan (sub con) v12.doc" changed to "Telecom, Inc\Plan\Amended XO Plan v.2.doc" |
| 429 | Change | "the following chapter 11" changed to "the following amended chapter 11" |
| 430 | Change | "to any Claim (a) any Claim" changed to "to any Claim: (a) any Claim" |
| 431-432 | Change | "aggregate amount of [\$642,843,000]." changed to "aggregate amount of \$642,843,000." |
| 433-434 | Change | "1.19 ATLT A Certificates»" changed to "1.19 ATLT "A" Trust Interests»" |
| 435 | Change | "means the [] ownership interests" changed to "means the ownership interests" |
| 436-437 | Change | "1.20 ATLT B Certificates»" changed to "1.20 ATLT "B" Trust Interests»" |
| 438 | Change | "means the [] ownership interests" changed to "means the ownership interests" |
| 439-440 | Change | "1.21 ATLT C Certificates»" changed to "1.21 ATLT "C" Trust Interests»" |
| 441 | Change | "means the [] ownership interests" changed to "means the ownership interests" |
| 442-443 | Change | "1.22 ATLT Certificates»" changed to "1.22 ATLT Trust Interests»" |
| 444-445 | Change | "collectively, the ATLT A Certificates, the ATLT" changed to "collectively, the ATLTInterests, the ATLT" |
| 446-448 | Change | ", the ATLT B Certificates and the ATLT" changed to ", the ATLT "B" Trust Interests and the ATLT" |
| 449-451 | Change | "and the ATLT C Certificates." changed to "and the ATLT "C" Trust Interests." |
| 452 | Change | "Bankruptcy Code, including," changed to "Bankruptcy Code to theAssets, including," |
| 453 | Change | "Distribution of Cash, to the" changed to "Distribution of Available Cash, to the" |
| 454-458 | Change | "of the ATLT Aas applicable," changed to "of the ATLT "A" Trustas applicable," |

| | | "amount aqual to ¢ as applicable." |
|------------|------------|--|
| 459-467 | Change | "amount equal to \$as applicable." |
| | | changed to "amount equal to 50% ofInterest, |
| | - | as applicable." |
| 468-471 | Change | "of the ATLT Aas applicable," changed to "of |
| | | the ATLT "A" Trustas applicable," |
| | | "Court, dated [], which is |
| 472-473 | Change | annexed" changed to "Court, dated April 21, |
| | | 2004, which is annexed" |
| 474 | Change | "satisfied or waived; and (b) no" changed to |
| +/+ | Change | "satisfied or waived and (b) no" |
| | | "which up to [10,000,000] shares shall" |
| 475-476 | Change | changed to "which up to 10,000,000 shares |
| | | shall" |
| 477 | | "such Subsidiary; and (b) no" changed to |
| 477 | Change | "such Subsidiary and (b) no" |
| 478 | Deletion | Agreement; (d) the ATLT Certificates |
| | | "; (e) the Reorganized" changed to |
| 479 | Change | "Reorganized" |
| 480 | Moved to | Incorporation; (e) the |
| | | "; (f) the Reorganized" changed to |
| 481 | Change | "Reorganized STFI Bylaws; (f) the |
| | Change | Reorganized |
| 482 | Deletion | Reorganized STFI Bylaws; (|
| 482 483 | Moved from | a neorganizeu STET Dyldws, (|
| 400 | |) the Deersenined Cubeidieries" shares of t |
| 484 | Change | ") the Reorganized Subsidiaries" changed to |
| | - | "Subsidiaries" |
| 485 | Change | "Subsidiaries Certificates" changed to |
| | - | "Subsidiaries' Certificates" |
| 100 107 | Change | "Incorporation; and (h) the Reorganized" |
| 486-487 | | changed to "Incorporation; and (g) the |
| | - | Reorganized" |
| 488 | Change | "Subsidiaries Bylaws." changed to |
| · · | | "Subsidiaries' Bylaws." |
| 489 | Change | "and ATCW and the common" changed to |
| | | "and ATCW and, as applicable, the common" |
| 490 | Change | "of STFI, (b) are necessary to" changed to "of |
| | | STFI, (b) necessary to" |
| 491 | Change | "STFI, and (c) and as more specifically" |
| | | changed to "STFI, and (c) more specifically" |
| 492 | Change | "documents ancillary, as applicable." changed |
| | | to "documents ancillary thereto, as applicable." |
| 402 | Change | "comments on, the Plan and" changed to |
| 493 | | "comments on the Plan and" |
| 40.4.405 | | "contained herein, none of the Buyer nor" |
| 494-495 | Change | changed to "contained herein, neither the |
| | - | Tenangea to contained herein, notition the |

| | | Buyer nor" |
|---------|-----------|--|
| | | "Professional Fee Claims." changed to |
| 496 | Change | "Professional Fee ClaimsExpense Claims Bar Date." |
| 497-498 | Change | "contained herein, none of the Buyer nor" changed to "contained herein, neither the |
| | | Buyer nor" |
| 499 | Change | "(i) reinstated (only toAsset), (ii) paid" changed to "(i) reinstated, (ii) paid" |
| 500 | Change | "Secured Claim, (iii) satisfied" changed to "Secured Claim together(iii) satisfied" |
| 501 | Insertion | reject the Plan. In thethe Plan Administrator. |
| 502-503 | Change | "either the ATLT A Certificates or the applicable" changed to "either the ATLT "A" Trustor the applicable" |
| 504-506 | Change | "either the ATLT B Certificates or the applicable" changed to "either the ATLT "B" Trustor the applicable" |
| 507-512 | Change | "Recovery, and (z) either In the event" changed to "Recovery, and (z) theInterests. In the event" |
| 513 | Change | "insufficient to satisfy(with such insufficiency" changed to "insufficient (with such insufficiency" |
| 514-515 | Change | "Effective Date), each such Holder" changed to "Effective Date) to) each such Holder" |
| 516-517 | Change | "such Holder shall receiveof the Cash Recovery," changed to "such Holder that elected,a Cash Recovery," |
| 518-526 | Change | "Cash Recovery, andshallas applicable" changed to "Cash Recovery, shallRecovery, if applicable" |
| 527-528 | Change | "applicable." changed to "applicable, as follows:ATLT "A" Trust Interests" |
| 529 | Change | "" changed to "Holders of Claims inATI Unsecured Claims." |
| 530-532 | Change | "either the ATLT A Certificates or the applicable" changed to "either the ATLT "A" Trustor the applicable" |
| 533-535 | Change | "either the ATLT B Certificates or the applicable" changed to "either the ATLT "B" Trustor the applicable" |
| 536-540 | Change | "Recovery, and (z) eitherapplicable Cash Recovery" changed to "Recovery, and (z) |

| | | theof the Cash Recovery" |
|---------|------------|--|
| | | "Recovery Election; provided, however, that" |
| 541-543 | Change | changed to "Recovery Election), theprovided |
| | J - | however, that" |
| | | "Holders in Class 4 that elect" changed to |
| 544-545 | Change | "Holders in Class 5 that elect" |
| 546 | Deletion | Effective Date), the to any Cash Recovery. |
| 547 | Moved from | Holders of Claims inClaims and (2) Allowed |
| | | "ATI Unsecured ClaimsDate), each such |
| 548 | Change | Holder" changed to "each such Holder" |
| | | "such Holder shall receiveof the Cash |
| 549-550 | Change | Recovery," changed to "such Holder that |
| | Ũ | elected a Cash Recovery," |
| | | "Cash Recovery, and theas applicable" |
| 551-559 | Change | changed to "Cash Recovery, shallATLT "A" |
| | | Trust Interests" |
| 560 | Moved to | Holders of Claims inClaims and (2) Allowed |
| 561 | Insertion | ATCW Unsecured Claims. |
| 562-563 | Change | "(Class 7)Equity»" changed to "(Class |
| 302-303 | Change | 7)Equity»" |
| 564 | Deletion | 5.1 Deemed |
| 565 | Moved from | Consolidation of ATCWfor Plan Purposes |
| 303 | | Only |
| 566-568 | Change | "." changed to "Substantive Consolidationfor |
| 500 500 | | Plan Purposes Only" |
| 569 | Insertion | . On the InitialInitial Effective Date. |
| 570 | Insertion | Entry of the ConfirmationATCW Debtors. |
| 571-572 | Change | "controversies under this" changed to |
| | | "controversies under the" |
| 573 | Moved to | In addition, on the Initial Effective Date |
| 574 | Insertion | each Debtor that holds anany Intercompany |
| | | Claims. |
| 575 | Change | "consolidation shall not (other" changed to |
| | | "consolidation andonly) shall not (other" |
| 576 | Moved from | ATCW Debtors. InInitial Effective Date |
| 577 | Deletion | , for all purposes, allamong the ATCW |
| | | Debtors. |
| 578-579 | Change | "classification, distribution" changed to |
| | | "classification, Distribution" |
| 580-581 | Change | "Plan. All Plan distributions" changed to "Plan. |
| - | 5 | All Plan Distributions" |
| 582-583 | Change | "and no Plan distribution" changed to "and no |
| | | Plan Distribution" |
| 584-588 | Change | "Acquired Assets shall beand the common" |
| | | changed to "Acquired Assetslimitation, the |

| | | common" |
|---------|-----------|---|
| 589 | Change | "Subsidiaries shall be sold," changed to "Subsidiaries) shall be sold," |
| 590 | Change | "to the Buyer, free and clear" changed to "to the Buyer free and clear" |
| 591 | Change | "Section 5.3(d) and (e) of" changed to "Section 5.3(a) (other(c), (d) and (e) of" |
| 592 | Change | "Subsidiaries if the Earlynot made (x) shall not" changed to "Subsidiaries (x) shall not" |
| 593-594 | Change | "Sale Order) or other encumbrances." changed to "Sale Order) and other encumbrances." |
| 595 | Change | "sole discretion inAgreement, in exchange" changed to "sole discretion, in exchange" |
| 596 | Change | "sole discretion inAgreement. All required" changed to "sole discretion. All required" |
| 597-598 | Change | "resign as of the Final Effective Date. As of the" changed to "resign as of theSubsidiary. As of the" |
| 599 | Change | "1129(a)(5), the Buyer will disclose" changed to "1129(a)(5), the Buyer, will disclose" |
| 600 | Change | "will disclose, on or prior" changed to "will disclose by making aCourt, on or prior" |
| 601-602 | Change | "Holders of ATLT B Certificates in accordance" changed to "Holders of ATLT "B" Trust Interests in accordance" |
| 603 | Change | "the Operating AgreementPurchase Agreement and" changed to "the Operating Agreement and" |
| 604-605 | Change | "this Section 5.4 and the terms" changed to "this Section 5.5 and the terms" |
| 606-608 | Change | "contemplated herein, (ii) liquidate, by" changed to "contemplated in the Plan, (ii) (a) liquidate, by" |
| 609 | Change | "other methods, the ATLT Assets," changed to "other methods or (b)the ATLT Assets," |
| 610 | Change | "reasonably possible, (iii) investigate," changed to "reasonably possible in(iii) investigate," |
| 611-613 | Change | "of Action, provided,resolve Disputed Claims," changed to "of Action, (iv) objectDisputed Claims," |
| 614 | Insertion | In the event that therethe Plan Administrator. |
| 615 | Insertion | To maximize the value ofATLT Trust Interests. |

| | | "shall include the VO Common" shanged to |
|---------|-----------|---|
| 616 | Change | "shall include the XO Common" changed to "shall include (a) the XO Common" |
| 617 | Change | "Common Stock, the New STFI" changed to "Common Stock, (b) the New STFI" |
| 618-619 | Change | "Common Stock, all Available Cash" changed to "Common Stock, (c) the(d) the Available Cash" |
| 620-625 | Change | "Available Cash and theof all Claims and Liens." changed to "Available Cash, (e) allnot the Acquired Assets." |
| 626-629 | Change | "(d) The ATLTin accordance" changed to "(d) The ATLT TrustDate, in accordance" |
| 630 | Change | "of the Plan, shall beto the Holders of" changed to "of the Plan, Holders of" |
| 631 | Change | "Unsecured Claims. The ATLT" changed to "Unsecured Claims (i) thatsuch Holder. The ATLT" |
| 632-635 | Change | ". The ATLT Certificatesthe value of ATLT." changed to ". The ATLT Trustthe Plan Administrator." |
| 636 | Change | "Effective Date. The Plan" changed to "Effective Date. TheHearing. The Plan" |
| 637 | Change | "shall include: (i) the wind-down" changed to "shall include, but shallto: (i) the wind-down" |
| 638-645 | Change | "liquidation, by(iii) the investigation," changed to "liquidation, in a(iv) the investigation," |
| 646-656 | Change | "of Action, provided,Date, (iv) resolution" changed to "of Action, (v) conduct(vi) resolution" |
| 657-658 | Change | "Disputed Claims, (v) administration" changed to "Disputed Claims, (vii) administration" |
| 659-660 | Change | "Plan, and (vi) filing of" changed to "Plan, and (viii) filing of" |
| 661 | Insertion | obligations. The Planby the Bankruptcy Court. |
| 662-663 | Insertion | (g) Bonding of Plansuch a bond be obtained. |
| 664-665 | Change | "(g) Directors of" changed to "(h) Directors of" |
| 666 | Insertion | directors of ATLT. Theat least ten (10) days |
| 667 | Moved to | prior |
| 668 | Insertion | to |
| 669 | Moved to | the Confirmation |
| 670 | Insertion | Hearing |
| 671-672 | Change | "(h) Post-Initial" changed to "(i) Post-Initial" |

| 673-674 | Change | "(i) Post-Initial" changed to "(j) Post-Initial" |
|---------|-----------|---|
| 675 | Change | "and expenses related to" changed to "and expenses, includingStates Code, related to" |
| 676-677 | Change | "in Section 5.4(h) of the Plan." changed to "in Section 5.5(h) of the Plan." |
| 678-680 | Change | "(j) Plan Administrator" changed to "(k) Fiduciary Duties of Plan Administrator" |
| 681-682 | Change | "Administrator and Fiduciary Duties. The Plan" changed to "Administrator and Directors of ATLT. The Plan" |
| 683 | Change | "Administrator shall act in" changed to "Administrator and each ofof ATLT shall act in" |
| 684-685 | Change | "Holders of ATLT Certificates" changed to "Holders of ATLT Trust Interests" |
| 686-687 | Change | "(k) Dissolution" changed to "(I) Dissolution" |
| 688 | Change | "ATI and ATCW a certification" changed to "ATI and ATCW, as the case may be, a certification" |
| 689-690 | Change | "on behalf of ATI and ATCW or payments" changed to "on behalf of such Entity or payments" |
| 691 | Change | "equivalent document. After (a)" changed to "equivalent document,ATLT Assets. After (a)" |
| 692 | Change | "Assets (or on the proceeds" changed to "Assets (or the proceeds" |
| 693-694 | Change | "and the ATLT Certificates pursuant to" changed to "and the ATLT Trust Interests pursuant to" |
| 695 | Change | "Bankruptcy Court, and (c) the" changed to "Bankruptcy Court inSection, and (c) the" |
| 696-697 | Change | "(I) Closing of the" changed to "(m) Closing of the" |
| 698-699 | Insertion | (n) Distributions. Theor the ATLT Agreement. |
| 700-701 | Insertion | (o) Federal Income Tax Treatment. |
| 702-705 | Change | "(m) Tax Treatment. If possible," changed to "(i) General. If possible," |
| 706 | Change | ""liquidating trust," as defined" changed to ""liquidating trust" as defined" |
| 707-708 | Change | "established by the IRS in Rev. Proc." changed to "established by theService in Rev. Proc." |
| | Change | "94-45, 1994- 2 C.B. 684," changed to "94-45, |

| | | 1994 2 C.B. 684," |
|---------|------------|--|
| 740 744 | | (ii) ATLT Assets Treatedof the ATLT |
| 710-711 | Insertion | Assets. |
| 712-713 | Insertion | (iii) Tax Reporting. |
| 714-715 | Insertion | A. The Planinterests in ATLT. |
| 716-717 | Insertion | B. As soon asany governmental unit. |
| 718-719 | Insertion | C. Subject towith the foregoing. |
| 720-721 | Insertion | D. The Planof such Disputed Claims. |
| 722-723 | Insertion | E. The Planthe dissolution of ATLT. |
| 724 | Change | "Unexpired Leases. Pursuant to" changed to "Unexpired LeasesSale Order. Pursuant to" |
| 725-728 | Change | "Schedules 2, 3, 4 and 5hereto shall be deemed" changed to "Schedules 2, 3 and 4 shall be deemed" |
| 729-736 | Change | "shall be deemed rejected2.5 of the Purchase" changed to "shall be deemed assumedwith the Purchase" |
| 737-740 | Change | "Purchase Agreementor lease. Contracts" changed to "Purchase Agreement andDate. Contracts" |
| 741 | Deletion | Buyer on the Initial Effective Date. Contracts |
| 742 | Moved from | listed on Schedule 4 shall be deemed |
| 743 | Change | "assumed and (whereon the date that Buyer" changed to "date that Buyer" |
| 744-756 | Change | "Contracts listed on(ii) 180 days after" changed to "Contracts and leasesof (i) 180 days after" |
| 757-763 | Change | "Effective Date or (iii)Date as reflected on" changed to "Effective Date and (ii) the date reflected on" |
| 764-773 | Change | "such Schedule 5hereto may be modified" changed to "such Schedule 4;the Plan may be modified" |
| 774-775 | Change | "Purchase Agreement." changed to "Purchase Agreement,Schedules 1, 2, 3 and 4." |
| 776-777 | Change | "the date of entry of anthe Confirmation" changed to "the date of notice of theof the Confirmation" |
| 778-779 | Change | "Confirmation Order) approving such" changed to "Confirmation Order or other order approving such" |
| 780 | Insertion | provided herein. Solelyan executory |

| | | contract. |
|---------|-----------|---|
| | | "unexpired leases may otherwise" changed to |
| 781 | Change | "unexpired leases and the Buyer may |
| | | otherwise" |
| 700 | Change | "Debtors and Buyer shall" changed to "Debtors |
| 782 | Change | and the Buyer shall" |
| 783 | Change | "Agreement. Buyer shall" changed to |
| 103 | Change | "Agreement. The Buyer shall" |
| 784 | Deletion | 6.2 Utility Services. |
| 785 | Insertion | . Utility Companies |
| 786 | Moved to | shall not be entitled toany additional deposits |
| 787 | Insertion | , payments |
| 788 | Moved to | or other financialthe Chapter 11 Cases |
| 789 | Insertion | . On and after theServices contract |
| 790 | Moved to | . Any Claim against a Debtor by |
| 791 | Insertion | a Utility Company (or aSection 3.4 of the |
| 791 | Insertion | Plan |
| 792 | Moved to | . The Buyer shall haveto Claims arising out |
| 192 | | of |
| 793 | Insertion | Utility Services. |
| 794-796 | Insertion | 6.3 Tariff Services» |
| 797 | Change | "After the" changed to ". After the" |
| | | "Effective Date, allCompanies shall |
| 798-801 | Change | continue" changed to "Effective Date, |
| | | allProviders shall continue" |
| | | "Reorganized STFI, the Reorganized" |
| 802 | Change | changed to "Reorganized STFI, the Buyer or |
| | | the Reorganized" |
| | | "Subsidiaries or Buyer without interruption" |
| 803-804 | Change | changed to "Subsidiaries, as the casebe, |
| | | without interruption" |
| 005 000 | | "interruption all Utility Services provided to" |
| 805-809 | Change | changed to "interruption all Tariffservices, |
| | | provided to" |
| 040.044 | | "Effective Date whetherCompanies shall not |
| 810-814 | Change | be" changed to "Effective Date. Access |
| | | Providers shall not be" |
| 915 946 | Change | "a Debtor by a Utility Company (or a Holder" |
| 815-816 | Change | changed to "a Debtor by an Access Provider (or a Holder" |
| | | "a Claim of a Utility Company) for the |
| 817-818 | Change | provision" changed to "a Claim of an Access |
| | | Provider) for the provision" |
| | | "provision of Utility Services to" changed to |
| 819-820 | Change | "provision of Tariff Services to" |
| L | | |

| 821-822 | Change | "Section 3.4 hereof. The Buyer" changed to "Section 3.4 of the Plan. The Buyer" |
|---------|-----------|--|
| 823-824 | Change | "arising out of Utility" changed to "arising out of Tariff" |
| 825-827 | Change | "6.3 Tariff Services" changed to "6.4 Risks of Litigationand Tariff Services" |
| 828 | Deletion | Tariff Services. |
| 829 | Deletion | After the InitialProviders shall continue |
| 830-847 | Change | "to provide to theout of Tariff Services." changed to ". The principalSubsidiaries." |
| 848-850 | Insertion | 6.5 Tariffs Servicesbe Executory Contracts» |
| 851 | Insertion | . To the extent the such Utility Services |
| 852 | Moved to | shall be treated in accordance with |
| 853 | Insertion | the provisions of sectionby the Court. |
| 854 | Change | "with the ATI Indentures and" changed to "with the ATI Note Indentures and" |
| 855 | Change | "ATLT until such time as" changed to "ATLT until the earlier of such time as" |
| 856-857 | Change | "7.6 Fractional ATLTDe Minimis Distributions" changed to "7.6 De Minimis Distributions" |
| 858 | Change | ". ATLT Certificates No Cash payment" changed to ". No Cash payment" |
| 859-860 | Change | "less than \$250.00 shall be required" changed to "less than \$500.00 shall be required" |
| 861 | Insertion | such Claim. The deemedthe ILEC Cure Amounts. |
| 862 | Change | "Bankruptcy Court or (ii) in the" changed to "Bankruptcy Court oror (ii) in the" |
| 863 | Change | "accordance with this Section 8.3" changed to "accordance with Section 8.3" |
| 864 | Change | "Section 8.3 and applicable" changed to "Section 8.3 of the Plan and applicable" |
| 865 | Change | "contained in this Section 8.3" changed to "contained in Section 8.3" |
| 866 | Change | "Section 8.3 shall constitute" changed to "Section 8.3 of the Plan shall constitute" |
| 867 | Change | "connection with estimation" changed to "connection with anyto or any estimation" |
| 868-869 | Change | "Plan for Class 4;law. Nothing contained" changed to "Plan for Class 4. Nothing contained" |
| 870-871 | Change | "Cash, ATLT Certificates, as applicable," changed to "Cash, ATLT Trust Interests, as |

| | | applicable," |
|-----------|-----------|---|
| 872-873 | Change | "and the ATLT Certificates reserved in" changed to "and the ATLT Trust Interests reserved in" |
| 874-875 | Change | "Cash and ATLT Certificates to be distributed" changed to "Cash and ATLT Trust Interests to be distributed" |
| 876 | Change | "Disputed Claim." changed to "Disputed Claim. The Planof such Disputed Claims." |
| 877-878 | Change | "Cash or ATLT Certificates, as applicable" changed to "Cash or ATLT Trust Interests, as applicable" |
| 879-880 | Change | "Cash or ATLT Certificates that such Holder" changed to "Cash or ATLT Trust Interests that such Holder" |
| 881-882 | Change | "thereon) and ATLT Certificates reserved in" changed to "thereon) and ATLT Trust Interests reserved in" |
| 883 | Insertion | Reorganized STFI and each of |
| 884 | Change | ", each of the" changed to "the IT Reorganized Subsidiaries, each of the" |
| 885 | Change | "approvals, rulings,opinions or documents" changed to "approvals, rulings or documents" |
| 886 | Change | "Non-Transferred Assets." changed to "Non- Transferred AssetsReorganized Subsidiary." |
| 887-888 | Change | "restoration shall not restore liens held by any" changed to "restoration shall notSale Order) held by any" |
| 889-890 | Insertion | (c) The Debtors are invalid Causes of Action. |
| 891-892 | Change | "10.5 Releases by the Debtors" changed to "10.5 RELEASES BY THE DEBTORS" |
| 893-1048 | Change | ". As of the Initialor gross negligence." changed to ". AS OF THE INITIALOR GROSS NEGLIGENCE." |
| 1049-1051 | Change | "10.6 Releases by Holdersand Equity Interests" changed to "10.6 RELEASES BY HOLDERSRECEIVING DISTRIBUTIONS" |
| 1052-1170 | Change | ". On the Initialor gross negligence." changed to ". ON THE INITIALOR GROSS NEGLIGENCE." |
| 1171-1172 | Change | "10.7 Release of Buyer" changed to "10.7 RELEASE OF BUYER" |
| 1173 | Deletion | In consideration for thethe Assumed Liabilities. |

| 1174 | Insertion | ON THE INITIAL EFFECTIVETHE ASSUMED LIABILITIES. |
|-----------|-----------|---|
| 1175-1176 | Change | "10.10 Injunction" changed to "10.10 INJUNCTION" |
| 1177-1360 | Change | ". All injunctions orthe Purchase Agreement." changed to ". ALL INJUNCTIONS OROF APPLICABLE LAW." |
| 1361 | Change | "and (ii) the written consent" changed to "and (ii) the prior written consent" |
| 1362 | Change | "Senior Lenders), upon order" changed to "Senior Lenders, and thethe Buyer), upon order" |
| 1363 | Change | "Liquidating Trust." changed to "Liquidating Trust" |
| 1364-1365 | Change | "March 18, 2004" changed to "April 21, 2004" |

| Statistics: | |
|----------------|-------|
| | Count |
| Insertions | 693 |
| Deletions | 630 |
| Moved from | 21 |
| Moved to | 21 |
| Format changed | 0 |
| Total changes | 1365 |